

University of North Dakota UND Scholarly Commons

Theses and Dissertations

Theses, Dissertations, and Senior Projects

January 2018

"Representative Bureaucracy" In Police Hiring Practices: A Case Study Of A Diverse Police Agency

Sara Franceen Edel

Follow this and additional works at: https://commons.und.edu/theses

Recommended Citation

Edel, Sara Franceen, ""Representative Bureaucracy" In Police Hiring Practices: A Case Study Of A Diverse Police Agency" (2018). *Theses and Dissertations*. 2204.

https://commons.und.edu/theses/2204

This Dissertation is brought to you for free and open access by the Theses, Dissertations, and Senior Projects at UND Scholarly Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of UND Scholarly Commons. For more information, please contact zeineb.yousif@library.und.edu.



"REPRESENTATIVE BUREAUCRACY" IN POLICE HIRING PRACTICES: A CASE STUDY OF A DIVERSE POLICE AGENCY

by

Sara Franceen Edel Bachelor of Arts, Saint Cloud State University, 1983 Master of Arts, University of Saint Thomas, 1994

A Dissertation Submitted to the Graduate Faculty

of the

University of North Dakota

in partial fulfillment of the requirements

for the degree of

Doctor of Philosophy

Grand Forks, North Dakota

August 2018



Copyright 2018 Sara Franceen Edel



This dissertation, submitted by Sara Franceen Edel in partial fulfillment of the requirements for the Degree of Doctor of Philosophy from the University of North Dakota, has been read by the Faculty Advisory Committee under whom the work has been done and is hereby approved.

Dr. Michael Meyer, Committee Chairperson

Dr. Bruce DiCristina, Committee Member

Dr. Wendelin Hume, Committee Member

Dr. Martin Gottschalk, Committee Member

Dr. Daphne Pedersen, Committee Member

This dissertation is being submitted by the appointed advisory committee as having met all of the requirements of the School of Graduate Studies at the University of North Dakota and is hereby approved.

Dr. Grant McGimpsey

Dean of the School of Graduate Studies

May 13, 2018

Date

PERMISSION

Title "Representative Bureaucracy" in Police Hiring Practices: A Case Study of a

Diverse Police Agency

Department Criminal Justice

Degree Doctor of Philosophy

In presenting this dissertation in partial fulfillment of the requirements for a graduate degree from the University of North Dakota, I agree that the library of this University shall make it freely available for inspection. I further agree that permission for extensive copying for scholarly purposes may be granted by the professor who supervised my dissertation work or, in his absence, by the Chairperson of the department or the dean of the School of Graduate Studies. It is understood that any copying or publication or other use of this dissertation or part thereof for financial gain shall not be allowed without my written permission. It is also understood that due recognition shall be given to me and to the University of North Dakota in any scholarly use which may be made of any material in my dissertation.

Sara Franceen Edel July 7, 2018



TABLE OF CONTENTS

LIST OF TA	BLES	XV
ACKNOWL	EDGMENTS	xxi
ABSTRACT	,	xxi
CHAPTER		
I.	INTRODUCTION	1
	Statement of the Problem	4
	Benefits of Representative Police Agencies	5
	Relevant Events	9
	Multicultural Communities and the Challenges for Law Enforcement	12
	Purpose of the Study	13
	Research Question, Seven Sub-questions, and Four Data Sets	14
	Primary Research Question	14
	Sub-question #1	14
	Sub-question #2	14
	Sub-question #3	15
	Sub-question #4	15
	Sub-question #5	15
	Sub-question #6	16

	Sub-question #7	16
	Significance of the Study	16
	Interchangeability of Terms	17
	Diversity	17
	Representativeness/Diversity/Reflectiveness	18
	Summary	18
	Organization of the Chapters	18
II.	REVIEW OF LITERATURE	19
	Important Social Events and Historical Context at the National Level	19
	Social Changes	20
	The Police Recruitment and Hiring Process	20
	A History of the Issue and Problem/President's Commissions and Task Forces, and Federal Assessments and Investigations	23
	The Kerner Commission – Watts Riots	23
	The Christopher Commission – Rodney King	24
	U.S. Commission on Civil Rights Reports	24
	Principles for Promoting Police Integrity	24
	The President's Task Force on 21st Century Policing	25
	The After-action Assessment of the Police Response In Ferguson, Missouri	25
	Four Models of Policing	27
	Community-Oriented and Problem-Oriented Policing	29



Color by Type of Agency and Size of Agency	31
Status of Females in Policing	32
Status of Officers of Color	34
The Effects of Changing Demographics within Police Agencies	35
The Effects of Representation of African American Police Officers	37
Hiring in the Spirit of Service	38
Theoretical and Conceptual Framework	38
Overview of Representative Bureaucracy Theory	39
Passive Representative Bureaucracy	41
Factors and Variables that Influence Passive Representation of Female Police Officers and Police Officers of Color	42
Economic Factors	43
Environmental or Demographic Factors	44
Population of Communities of Color	44
City Population	45
Region in the United States	46
Political Factors	48
Presence of Females or People of Color in Public Office	48
Structure or Type of Government	49
Professional Associations	50
Institutional or Organizational Factors	50



Consent Decrees	
	54
Police Chiefs of Color/Top Management of Color	J T
Agency Size	55
Presence and Type of Union or Collective Bargaining Unit	55
Autonomous Professional Personnel Bureaus	57
African Americans Personnel Directors and Administrators	58
Active Representative Bureaucracy	58
The Relationship between Passive Representative Bureaucracy and Active Representative Bureaucracy	59
Police Performance Outcomes of a Representative Police Force	62
Street-level Bureaucrats and Discretion	64
Administrative Discretion over Policy Outcomes in the Context of Representative	66
Role Acceptance of Administrators	68
Public and Police Attitudes	69
Attitude Congruence	71
Police Attitudes towards the Community	72
The Police Subculture	72
Leadership at the Local Level to Initiate Change	74
How Social Equity Theory Ties into Representative Bureaucracy Theory	75



	Police Officer Retention and Separation	76
	Police Officers who Leave for another Law Enforcement Agency	80
	Females who Leaving Policing	80
	Length of Service and Separation	81
	Summary	81
III.	METHODS	82
	Theoretical and Conceptual Framework	82
	Research Design, Philosophical Orientation, and Methodology	82
	The Case Study as the Research Approach	83
	Research Questions	85
	Primary Research Question	85
	Sub-question #1	85
	Sub-question #2	85
	Sub-question #3	85
	Sub-question #4	86
	Sub-question #5	86
	Sub-question #6	86
	Sub-question #7	86
	Data Sources, Data Collection, and Data Analysis	86
	Description of Semi-structured Interviews	88
	Interview Schedule	88
	Data Analysis for Interviews	88



Data Analysis Techniques and Strategies90
Triangulation of Data Sources, Methods, and Levels of Analysis90
The Case Selected for Case Study92
Description of Case Agency93
Ethical Issues
Protection of Participants from Harm94
Voluntary Participation94
Informed Consent94
Confidentiality and Anonymity95
Securing the Data95
Reliability and Validity95
Internal Validity98
Construct Validity98
External Validity98
Authenticity99
Conclusion
IV. COLLECTION AND ANALYSIS OF QUANTITATIVE DATA100
Research Questions
Primary Research Question
Sub-question #1
Sub-question #2
Sub-question #3101
Full-time Police Officer Composition Data101

Full	-time Police Officer Applicant Data	103
	2009 Posting for Full-time Police Officer External and Internal Applicants	107
	2010 Posting for Full-time Police Officer External and Internal Applicants	111
	2011 Posting for Full-time Police Officer External and Internal Applicants	116
	2012 Posting for Full-time Police Officer External and Internal Applicants	122
	2013 Posting for Full-time Police Officer External and Internal Applicants	127
	2014 Posting for Full-time Police Officer External and Internal Applicants	132
	2016 Posting for Full-time Police Officer External and Internal Applicants	137
	Summary of Applicants Hired from Postings Open for External and Internal Applicants	143
	2011 to 2015 Postings for Full-time Police Officer Open Only for Internal Applicants	145
	Comparison of Applicants Hired from Postings Open for External and Internal Applicants to Postings Open Only for Internal Applicants	147
Full	-time Police Officers Separation Data	148
Con	clusion	157
	CTION AND ANALYSIS OF QUALITATIVE INTERVIEW	159
Desc	cription of Interview Data	159
	Sub-question #4	159
	Sub-question #5	159



	Sub-q	uestion #6	159
	Sub-q	uestion #7	160
		entifying Themes from Analysis of the Interview	160
	Theme	es for Recruiting, Hiring, and Retention	162
		The Importance of the Agency being Reflective of its Communities	162
		The Active Role of the Chief in Diversifying the Agency	165
		The Impact of Changes in the Hiring Process on Increasing Diversity	166
		The Influence of Administrative Discretion on Decisions made Regarding Recruiting and Hiring Practices	170
		The Relationship between the Agency being Responsive to its Communities through a Commitment to Community Policing and Recruiting, Hiring, and Retaining Diversity	172
		The Impact of External Factors on the Applicant Pool	175
		The Influence of Organizational Factors on Recruiting, Hiring, and Retention	177
		The Perception that some Diverse Applicants Hired were not Qualified	179
		The Positive Influence of Professional Organizations and Associations on Increasing Recruiting, Hiring, and Retention of Female Officers and Officers of	100
	C. 1. 1	Color	
VI.	DISCUSSION A	ND CONCLUSIONS	182



Summary of the Findings	32
Contributions of this Study to the Literature18	38
Discussion and Conclusions for the Composition Data19	90
Discussion and Conclusions for the Application Data19)2
Overview of Applicant Data19)3
The Impact of Changes in the Hiring Process on Increasing Diversity)7
The First Change – Moving the Second Interview to Before the Background Investigation) 7
The Second Change – Addition of Desired Qualifications20)()
The Third Change – Elimination of the Written Test20)3
The Fourth Change – Changes in the Background Investigations)6
The Impact of the Changes in the Application Process on the Hiring Goals for the Postings Open to External and Internal Applicants)8
The 2011 to 2015 Postings for Full-time Police Officer Internal Only Applicants	2
The Impact of the Changes in the Application Process on the Hiring Goals for the Postings Open Only to Internal Applicants21	3
Discussion and Conclusions for Separation Data21	4
Turnover Rates21	.4
Police Officers who Left for another Law Enforcement Agency (Lateral Moves)	20
Female Officers and Officers of Color who Left Policing22	21
Length of Service and Separation	23
Important Factors in Recruiting, Hiring, and Retaining Diversity22	23



	Importance of the Agency being Reflective of its Communities22	23
	The Relationship between the Agency being Responsive to its Communities through a Commitment to Community Engagement and Recruiting, Hiring, and Retaining Diversity	24
	The Impact of External Factors on the Applicant Pool22	26
	The Influence of Organizational Factors on Recruiting, Hiring, and Retaining Diversity	27
	The Perception that Some Diverse Applicants Hired were Not Qualified	29
	The Influence of Administrative Discretion on Decisions made Regarding Recruiting and Hiring Practices23	31
	Influence of Affirmative Action Programs	32
	The Influence of Structure of Government/Autonomous Personnel Bureaus/Civil Service	33
	The Influence of Professional Associations	34
	The Influence of its Union	35
	The Active Role of Leadership in Diversifying their Agency	36
	Limitations of this study23	39
	Recommendations for Future Research	10
	Conclusion	11
APPENDIX		12
REFERENCES	24	16



LIST OF TABLES

Table		Page
1.	Percent of African American Police Officers, African Americans in Population, and Female Police Officers for the Four Agencies Policing Ferguson, Missouri	. 26
2.	Mean Turnover Rates for the 2003 LEMAS Survey and the 2008 CSLLEA Data for Geographic Location, Urban Status, Type of Agency, and Size of Agency	. 79
3.	Type of Separation for the 2008 CSLLEA Data by Size of Agencies	. 79
4.	Percent of Full-time Police Officers for each Year from 2009 to 2017 by Gender	. 102
5.	Percent of Full-time Police Officers of Color for each Year from 2009 to 2017	. 102
6.	Percent of Full-time Police Officers for each Year from 2009 to 2017 by Ethnic Group	. 103
7.	Gender of 2009 Full-time Police Officer Applicants	. 108
8.	Ethnic Group of 2009 Full-time Police Officer Applicants	. 108
9.	Gender of 2009 Full-time Police Officer Applicants Passing Step 1	. 109
10.	Ethnic Group of 2009 Full-time Police Officer Applicants Passing Step 1	. 109
11.	Gender of 2009 Full-time Police Officer Applicants Passing Step 2	. 110
12.	Ethnic Group of 2009 Full-time Police Officer Applicants Passing Step 2	. 110
13.	Gender of 2009 Full-time Police Officer Applicants Hired – Step 3	. 111
14.	Ethnic Group of 2009 Full-time Police Officer Applicants Hired – Step 3	. 111
15.	Gender of 2010 Full-time Police Officer Applicants	. 112
16	Ethnic Group of 2010 Full-time Police Officer Applicants	112

17.	Gender of 2010 Full-time Police Officer Applicants Passing Step 1	. 113
18.	Ethnic Group of 2010 Full-time Police Officer Applicants Passing Step 1	. 113
19.	Gender of 2010 Full-time Police Officer Applicants Passing Step 2	. 114
20.	Ethnic Group of 2010 Full-time Police Officer Applicants Passing Step 2	. 114
21.	Gender of 2010 Full-time Police Officer Applicants Passing Step 3	. 115
22.	Ethnic Group of 2010 Full-time Police Officer Applicants Passing Step 3	. 115
23.	Gender of 2010 Full-time Police Officer Applicants Hired – Step 4	. 116
24.	Ethnic Group of 2010 Full-time Police Officer Applicants Hired – Step 4	. 116
25.	Gender of 2011 Full-time Police Officer Applicants	. 117
26.	Ethnic Group of 2011 Full-time Police Officer Applicants	. 117
27.	Gender of 2011 Full-time Police Officer Applicants Passing Step 1	. 118
28.	Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 1	. 118
29.	Gender of 2011 Full-time Police Officer Applicants Passing Step 2	. 119
30.	Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 2	. 119
31.	Gender of 2011 Full-time Police Officer Applicants Passing Step 3	. 120
32.	Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 3	. 120
33.	Gender of 2011 Full-time Police Officer Applicants Passing Step 4	. 120
34.	Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 4	. 121
35.	Gender of 2011 Full-time Police Officer Applicants Hired – Step 5	. 121
36.	Ethnic Group of 2011 Full-time Police Officer Applicants Hired – Step 5	. 122
37.	Gender of 2012 Full-time Police Officer Applicants	. 123
38.	Ethnic Group of 2012 Full-time Police Officer Applicants	. 123
39.	Gender of 2012 Full-time Police Officer Applicants Passing Step 1	. 124



40.	Ethnic Group of 2012 Full-time Police Officer Applicants Passing Step 1	. 124
41.	Gender of 2012 Full-time Police Officer Applicants Passing Step 2	. 124
42.	Ethnic Group of 2012 Full-time Police Officer Applicants Passing Step 2	. 125
43.	Gender of 2012 Full-time Police Officer Applicants Passing Step 3	. 125
44.	Ethnic Group of 2012 Full-time Police Officer Applicants Passing Step 3	. 126
45.	Gender of 2012 Full-time Police Officer Applicants Hired – Step 4	. 127
46.	Ethnic Group of 2012 Full-time Police Officer Applicants Hired – Step 4	. 127
47.	Gender of 2013 Full-time Police Officer Applicants	. 128
48.	Ethnic Group of 2013 Full-time Police Officer Applicants	. 128
49.	Gender of 2013 Full-time Police Officer Applicants Passing Step 1	. 129
50.	Ethnic Group of 2013 Full-time Police Officer Applicants Passing Step 1	. 129
51.	Gender of 2013 Full-time Police Officer Applicants Passing Step 2	. 130
52.	Ethnic Group of 2013 Full-time Police Officer Applicants Passing Step 2	. 130
53.	Gender of 2013 Full-time Police Officer Applicants Passing Step 3	. 131
54.	Ethnic Group of 2013 Full-time Police Officer Applicants Passing Step 3	. 131
55.	Gender of 2013 Full-time Police Officer Applicants Hired – Step 4	. 132
56.	Ethnic Group of 2013 Full-time Police Officer Applicants Hired – Step 4	. 132
57.	Gender of 2014 Full-time Police Officer Applicants	. 133
58.	Ethnic Group of 2014 Full-time Police Officer Applicants	. 133
59.	Gender of 2014 Full-time Police Officer Applicants Passing Step 1	. 134
60.	Ethnic Group of 2014 Full-time Police Officer Applicants Passing Step 1	. 134
61.	Gender of 2014 Full-time Police Officer Applicants Passing Step 2	. 135
62.	Ethnic Group of 2014 Full-time Police Officer Applicants Passing Step 2	. 135

63.	Gender of 2014 Full-time Police Officer Applicants Passing Step 3	136
64.	Ethnic Group of 2014 Full-time Police Officer Applicants Passing Step 3	136
65.	Gender of 2014 Full-time Police Officer Applicants Hired – Step 4	137
66.	Ethnic Group of 2014 Full-time Police Officer Applicants Hired – Step 4	137
67.	Gender of 2016 Full-time Police Officer Applicants	138
68.	Ethnic Group of 2016 Full-time Police Officer Applicants	138
69.	Gender of 2016 Full-time Police Officer Applicants Passing Step 1	139
70.	Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 1	139
71.	Gender of 2016 Full-time Police Officer Applicants Passing Step 2	140
72.	Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 2	140
73.	Gender of 2016 Full-time Police Officer Applicants Passing Step 3	141
74.	Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 3	141
75.	Gender of 2016 Full-time Police Officer Applicants Passing Step 4	141
76.	Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 4	142
77.	Gender of 2016 Full-time Police Officer Applicants Hired – Step 5	143
78.	Ethnic Group of 2016 Full-time Police Officer Applicants Hired – Step 5	143
79.	Number and Percent of Female and Persons of Color Hired for Full-time Police Officer from 2009 to 2016 Postings Open to External and Internal Applicants	
80.	Number and Percent of Full-time Police Officers Hired from 2009 to 2016 Postings Open to External and Internal Applicants by Ethnic Group	144
81.	Gender of 2011 to 2015 Full-time Internal Police Officer Applicants	146
82.	Ethnic Group of 2011 to 2015 Full-time Internal Police Officer Applicants	146
83.	Gender of 2011 to 2015 Full-time Internal Police Officer Applicants Hired	146



84.	Ethnic Group of 2011 to 2015 Full-time Internal Police Officer Applicants Hired	147
85.	Number and Percent of Female and Person of Color Hired for Full-time Police Officer by Type of Posting (Open to all Applicants and Internal Only Applicants)	148
86.	Number of Full-time Police Officer Separations by Turnover Indicator and Total Turnover Rates for each Year from 2010 to 2016	150
87.	Number and Average Turnover Rate for the Total Number of Officers who Separated from 2010 to 2016 by Turnover Indicator	151
88.	Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Reason of Separation and Type of Separation	152
89.	Percent of Reason of Separation from 2010 to mid-2017 using Reaves and Case Agency Categories	152
90.	Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Reason of Separation, Gender, and Ethnic Group	154
91.	Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Five or Less Years of Service and More Than 5 Years of Service	155
92.	Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Years of Service	155
93.	Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Years of Service, Gender, and Ethnic Group	155
94.	Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Years of Service and Type of Separation	156
95.	Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Reason of Separation and Years of Service	157
96.	Percent of Females and Persons of Color Hired for Full-time Police Officer by each step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants	195
97.	Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – First Change	200



98.	Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – Second Change	202
99.	Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – Third Change – First Analysis	204
100.	Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – Third Change – Second Analysis	206
101.	Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – Fourth Change	207
102.	Percent of Underrepresented Groups Applied and Hired for Full-time Police Officer from the 2009 through 2016 Postings Open to External and Internal Applicants	209
103.	Percent of Females and Persons of Color Applied and Hired for Full-time Police Officer from the 2009 through 2016 Postings Open to External and Internal Applicants	210
104.	Percent of Females, Persons of Color, and Underrepresented Groups Applied and Hired for Full-time Police Officer Internal Only Postings from 2011 through 2015	214
105.	Agency Turnover Rates for Officers who Separated from 2010 to 2016 and the 2003 LEMAS Data and 2008 CSLLEA Data (Wareham et al., 2015) by Turnover Indicators	216
106.	Mean Turnover Rates for the Case Agency, 2003 LEMAS Survey, and the 2008 CSLLEA Data for Geographic Location, Urban Status, Type of Agency, and Size of Agency	218
107.	Type of Separation for the 2008 CSLLEA Data (Reaves, 2015) and Case Agency	220
108.	Number and Percent of Agency Turnover Rates and Fyfe and Kane's (2006) Study by Female and Male Resignations and Involuntary Terminations	222



ACKNOWLEDGMENTS

I wish to thank numerous people who have helped me achieve my doctoral degree. A special thanks to my advisor and dissertation chair, Dr. Michael Meyer, who gave me his time, guidance, patience, and feedback to grow as a scholar and to conduct quality research. I also wish to thank my committee members, Dr. Bruce DiCristina, Dr. Martin Gottschalk, Dr. Wendelin Hume, and Dr. Daphne Pedersen, for their time and commitment in reviewing my work and offering advice throughout the process. Also, to Dr. Hume, who taught several of my courses, offered feedback to improve my writing skills, and over the past several years, listened, and offered advice and encouragement when I needed it the most. To Dr. Gottschalk for helping me write more concisely and believe in myself. Thanks to Dr. DiCristina for offering advice and his expertise in conducting qualitative methods. Thanks to Dr. Roni Mayzer for providing her expertise and guidance. I also want to thank the administration, criminal justice faculty, and IT staff from the University of North Dakota, St. Cloud State University, and Hennepin Technical College, who arranged for the remote videoconferencing for some of my courses.

I also want to acknowledge the support from many family, friends, and co-workers who kept me on the path to completion. To friend and colleague Dr. Lori Schroeder, who is the epitome of a "you can do this" attitude. Over the past few years, she has lent a sympathetic ear, mentored and inspired me, and offered encouragement and advice to finish my doctoral degree. To Alec Sonsteby, friend and confidant, for his tireless effort in obtaining several older documents, teaching me to be more comfortable using technology, and keeping me motivated.



To Mike Colestock who advised, encouraged, and supported me, and always knew the right thing to say to keep me motivated and focused. To Dr. Everett Doolittle and Dr. Raj Sethuraju, who have supported me throughout this process and who let me bounce ideas off of them.

I want to personally thank my dear friend, Cheryl Schroeder, who has always been there for me, who encouraged me to pursue my doctorate, who listened, offered advice, and kept me determined throughout this journey. To John Pundsack, Joe Briol, and Roger Jacquemart, who listened, offered support, and cheered me on. To all the staff and managers at Starbucks who provided me with "office space", sustenance, and heartfelt words of encouragement during the hundreds of hours I spent there.

To Gloria Alston, who told me to follow my heart and everything else would fall into place. To my spouse Dee, who stood by me, put her life on hold so I could follow my dream, and who repeatedly asked me when I would be finished with the numerous stacks of paper that took over our house. To my in-laws who understood when I missed family activities to study and write. To aunt Candy Doyle and uncle Richard Prien, who always asked me how my dissertation was coming along and encouraged me to keep going when I was overwhelmed.

Finally, a special thanks to the agency and command staff for providing the data and in the agency's efforts to be reflective of and responsive to its communities. To the head of the agency who models innovative leadership and who supported me. I also wish to thank the interview participants who contributed to this study by providing their valuable perceptions. A special thanks to the HR manager who met with me in the early stages of the study and provided me with the data to ensure this would be a "robust study". I especially wish to thank my primary agency contact, who spent countless hours providing me with data, coordinating the interviews,



and who was always available to answer questions, serve as a sounding board, and offer feedback. Without this individual, this study would not have been possible.



xxiii

To my spouse, Dee Reinking; and in memory of my dad, John Edel; aunt, Frances Edel; mother-in-law, Gloria Alston; cousins, Colleen and Lorraine Kadlec; forever friend, Diane Wallin; and dog, AJ. You all inspired me throughout this journey!



ABSTRACT

Diverse police agencies that are reflective of communities they serve are vital to community-oriented and problem-oriented policing, and in repairing fractured relationships by establishing trust and a partnership with communities. Agencies that engage in community-oriented and problem-oriented policing also attract more female officers and officers of color. The purpose of this study is to describe how one police agency in the Midwest recruits, hires, and retains a representative agency, and to describe what factors are important in this process. It is a descriptive, single case study that uses triangulation of multiple methods, including quantitative and qualitative methods, and multiple data sources. The quantitative methods include composition data from 2009 through 2017, application data from 2009 through 2016, and separation data from 2010 through mid-2017. The one qualitative method, interviews of personnel responsible for the implementation and oversight of recruiting, hiring, and retention, provide detailed descriptions from the perspectives of those involved in these practices.

It is evident from the data that representation of females, officers of color, and officers who are fluent in languages other than English, has increased in numbers and percentages for both applications and officers hired. Factors responsible for these increases include the Human Resources Office and command staff prioritizing being reflecting of and responsive to the communities it serves, targeting recruitment efforts, setting goals for hiring and composition of female officers and officers of color, and implementing changes in their application process.

xxiv



This study has implications for this agency, the communities it serves, and other agencies. This agency may use the results of this study to assess the future impact of its recruiting, hiring, and retention practices. This agency may also evaluate the future impact of increasing its diversity on the job performance of officers. Other police agencies who want to be reflective of their communities and engage in community-oriented and problem-oriented policing, can use this case study as an example. Also discussed are the limitations of the study and recommendations for future research.



CHAPTER I

INTRODUCTION

Recent deadly encounters between the police and predominantly African American males have brought race relations between the police and communities, particularly the African American community, to the forefront. These deadly events have occurred in a number of cities across the nation, including Tulsa, Oklahoma; Charlotte, North Carolina; Minneapolis, Minnesota; Baton Rouge, Louisiana; Ferguson, Missouri; Baltimore, Maryland; North Charleston, South Carolina; New York City, New York; Cleveland, Ohio; Chicago, Illinois; and Milwaukee, Wisconsin. These events sparked intense scrutiny and a push to reform policing. Many cities that have had events in which the police were accused of racial bias, misconduct, and use of excessive force involving White officers and people of color, are communities with populations which are growing in the number of residents who are people of color. However, these communities have city leaders, police officers, and police chiefs who are mostly White. Recruiting, selection, and training have become pivotal issues for police agencies due to major shifts in the role and function of policing and external demands for greater accountability, transparency, professionalism and legitimacy (White & Escobar, 2008, p. 119). Recruiting, hiring, and retaining a representative police force is one way to repair fractured relationships and improve the image of police, police-community relations, trust, and the delivery of police services.



Relationships between the police and communities of color, particularly the African American community, have been problematic for law enforcement agencies throughout the United States for several decades. Shusta, Levine, Wong, Olson, and Harris (2011) have highlighted that "in the past 30 years, these struggles have become even more complex and challenging as large numbers of non-English-speaking immigrants, documented and undocumented, have come to America seeking economic, religious, or political freedom" (p. xi). Demands by society in general, community groups and leaders, and professional groups, have pushed for police to be more accountable and engage in community-oriented and problemoriented policing in order to build trust and cooperation between the community and police agencies. The community-oriented policing philosophy includes police agencies working with the community to prevent and solve crime and improve quality of life issues, an emphasis on tasks other than crime fighting, increased victim assistance, and increased accountability to the public (J. Greene, 2000; Mastrofski, Worden, & Snipes, 1995; Skolnick & Bayley, 1988). Problem-oriented policing is more focused than community-oriented policing and includes working with the community and other agencies to address the symptoms of problems within the community (J. Greene, 2000). Recruiting, hiring, and retaining diversity within law enforcement agencies that are reflective of the communities they serve is vital because citizens are more willing to cooperate and be supportive if they perceive the police as legitimate and credible (Shusta et al., 2011, p. xi). This recognizes that "officers come into contact with persons from different cultural or economic backgrounds, socioeconomic classes, religions, and sexual orientations on a daily basis" (Shusta et al., p. 41). Shusta et al. also state that the different groups of people that police interact with bring different perspectives to these interactions, and police agencies with officers from diverse backgrounds are more responsive in providing police



services (p. 41). Ineffective interactions with citizens lead to poor police-community relations, increase tension, and diminish trust and cooperation from the community.

Although diversity is a broad concept that includes gender, race, ethnicity, national origin, culture, language, religion, sexual orientation, age, disability, homelessness, and people with other diverse backgrounds and experiences, this study only addresses gender and race/ethnicity. This research uses the following terms when discussing race and ethnicity: race, ethnicity, ethnic groups, diversity, Whites, African Americans, Blacks, Hispanics, Latinos, American Indians, Native Americans, Alaskan Natives, Asians, Pacific Islanders, non-Whites, people of color, officers of color, communities of color, minorities, and underrepresented groups or populations. This research also uses the terms representative, representativeness, reflective, and diversity interchangeably.

This research is part of a larger discussion in the U.S. on the relationships between race and policing. In order to build trust in fractured relationships and community partnerships, one of the strategies recommended is for agencies to reflect the demographics of the communities they serve. There were a number of agencies and people who wrote reports which supported the recommendation that police agencies should reflect the communities they serve. The notable reports include: *Task Force Report: The Police* (U.S. President's Commission on Law Enforcement and Administration of Justice, 1967), the *Report of the National Advisory Commission on Civil Disorders* (U.S. National Advisory Commission on Civil Disorders, 1968), the *Report of the Independent Commission on the Los Angeles Police Department* (Independent Commission on the Los Angeles Police Department (Independent Commission on the Los Angeles Police Department (U.S. Commission on Civil Rights in America (U.S. Commission on Civil Rights, 2000), the *After-action Assessment of the Police Response to the August 2014*



Demonstrations in Ferguson, Missouri (Institute for Intergovernmental Research, 2015), and the Final Report of the President's Task Force on 21st Century Policing (President's Task Force on 21st Century Policing, 2015).

Statement of the Problem

Having diverse law enforcement agencies that are representative of the communities they serve is critical as officers with "diverse backgrounds bring a broad range of cultural understanding and language skills to the force" (Matthies, Keller, & Lim, 2012, p. 1). The Kerner Commission established "deep hostility between police and ghetto communities" to be a key cause of "grievance, tensions, and ultimately disorder" during the 1960s (U.S. National Advisory Commission on Civil Disorders, 1968, p. 299). As a solution, it recommended hiring more minority officers and assigning them in minority neighborhoods (U.S. National Advisory Commission on Civil Disorders). Due to the extensive history of negative race relations between the police and minority communities, particularly African American communities, police agencies "have come under increasing pressure from community groups, professional organizations, and their constituents to hire more female and minority police officers" (Raganella & White, 2004, p. 501). Many scholars, reports on disorders and riots, President's Commissions, community leaders and groups, and citizens stress that the police will be viewed as more legitimate, transparent, and more effective in partnering with the community in preventing and controlling crime if police officers reflect the communities they serve (Matthies et al., 2012; President's Task Force on 21st Century Policing, 2015; Raganella & White, 2004; Scrivner, 2006). The President's Commission on Law Enforcement and the Administration of Justice (1967) concluded that disparity between the racial composition of police agencies and residents in neighborhoods was related to brutality, unequal treatment of African Americans, and poor



relations between the African American community and the police, leading to civil unrest and riots. Though agencies have made efforts to increase the representation of women and officers of color, including through Affirmative Action and consent decrees, many law enforcement agencies still have not achieved "labor population benchmarks" and do not reflect the demographics in the communities they serve (Matthies et al., 2012; Sklansky, 2006). "Affirmative Action" is a concept where employers take active steps to improve employment or equal opportunity for women and people of color and to "redress past discrimination" (Dempsey & Forst, 2016, p. 201). The concept of "Equal Opportunity" is different than Affirmation Action and is defined as ensuring "that no discrimination takes place and that everyone has the same opportunity to obtain a job or promotion" (p. 201). Employers can be both an Affirmative Action and an Equal Opportunity employer. A "consent decree" "is an agreement that binds the agency to a particular course of action in regard to hiring and promoting women and minorities in law enforcement" (p. 209).

Benefits of Representative Police Agencies

Having police agencies that are representative of the communities they serve is necessary and timely. Traditionally, law enforcement has been made up of mostly White males from the middle class. Gustafson (2013) stated that the significance of hiring people of color has the potential to destroy stereotypes and increase perceptions of legitimacy by improving interactions between White officers and officers of color, increasing appreciation of other cultures, viewing communities of color more positively, increasing understanding and communication, and closing gaps in social distance leading to more impartiality, fairness, and trustworthiness in the delivery of police services (p. 720).



Female officers and officers of color bring many benefits to a police agency, which includes bringing their culture, language skills, values, and beliefs to those of the diverse communities they serve, leading to a better understanding of those who share similar backgrounds. It also increases understanding and awareness on those from a different background and challenges preconceived notions, prejudice, and racism people hold regarding different groups (Shusta et al., 2011, p. xiii). In Weitzer and Tuch's (2006) national representative survey of U.S. citizens, citizens stated that police departments that are racially diverse "would improve police-community relations by promoting understanding between officers and residents" (p. 141), "curb racially biased practices", improve police behavior, enhance "community confidence in police", change "internal workings of the police department" (p. 142), and have a symbolic effect by promoting equal opportunity and fairness (p. 143).

Agencies with women officers and officers of color are likely to increase awareness of and sensitivity toward diverse cultures and communities among other police officers. This increases effectiveness, trust, and police-community relations. It is also important for women and people of color to see other women and people of color in all positions and ranks within police agencies to serve as role models for those aspiring to be police officers.

The organizational culture is also important to set the proper tone for respecting diversity, including how an agency treats co-workers and the citizens that police officers encounter. To gain the respect and cooperation from the communities they serve, law enforcement agencies must show the communities that they make a sincere effort in recruiting and hiring a diverse agency. It is imperative to treat all officers within the department with respect and dignity. If law enforcement agencies do not have the respect and trust from their female officers and officers of color, agencies cannot expect to gain the respect and trust from the communities they



serve. For example, if an agency discriminates against its officers based on gender and/or race/ethnicity, how can the community entrust that it will be treated fairly and with respect? This sends a strong message. Research shows that having officers from diverse backgrounds increases organizational credibility and has a positive influence on the beliefs, attitudes, stereotypes, biases, and prejudices of other officers. This includes the agency valuing and "understanding the unique characteristics, learning styles, and work attitudes" of the people working within the agency (Shusta et al., 2011, p. 100).

For decades, many people questioned the legitimacy of law enforcement because of its discriminatory practices in recruiting, hiring, retaining, and promoting women and people of color and inability to break down the traditional white male profession (G. Cordner & A. Cordner, 2011; Jordan, Fridell, Faggiani, & Kubu, 2009; Morabito & Shelley, 2015). Law enforcement in America has not fully attained the goal of being representative of the communities they serve. Martin (1994) attributes this to long standing discrimination against women and minorities. Some of the recruiting and hiring practices of police agencies may result in unintentional discrimination, which violates the theory of "adverse impact". Dempsey and Forst (2016) described the concept of adverse impact from the U.S. Equal Employment Opportunity Commission (EEOC) guidelines as "when the selection rate for any gender, race, or ethnic group is less than 80 percent of the selection rate for the group with the highest selection rate" (p. 115). Title VII of the 1964 Civil Rights Act (CRA) that prohibits employment discrimination on the basis of race, color, national origin, sex, and religion diminished this discrimination to some degree (Morabito & Shelley, 2015). Morabito and Shelley add that Executive Order 11246 in 1965 laid the groundwork for Affirmative Action programs to increase diversity in the United States work force.



Over the past several decades, racial and ethnic minorities and women have made progress in being hired in police agencies in the U.S. However, many scholars have noted that this progress has slowed to a halt in many agencies (G. Cordner & A. Cordner, 2011; Morabito & Shelley, 2015; Reaves, 2010; Reaves, 2015). The numbers in the Law Enforcement Management and Administrative Statistics (LEMAS) survey collected by the U.S. Department of Justice demonstrate that women police officers and police officers of color are not representative of the general population or general workforce (Morabito & Shelley, 2015; Reaves, 2010; Reaves, 2015).

In their study on the status of women in policing, the National Center for Women and Policing (2002) affirmed that "the continued under-representation of women in policing is a significant contributing factor to the widespread excessive force and corruption scandals plaguing law enforcement today" (p. 3). Research has shown that women police rely less often on physical force and more on their communication skills, and that they are involved less in excessive force complaints, sustained allegations, and civil liability payouts (p. 3).

Through a grant from the U.S. Department of Justice, Bureau of Justice Assistance, the National Center for Women and Policing (2001) developed a self-assessment guide for law enforcement agencies in recruiting and retaining women. The guide endorses increasing the presence of women in all ranks of police agencies "as a strategy to strengthen community policing, reduce police use of force, enhance police response to domestic violence, and provide balance to the workforce" (p. 1). It also found that research in both the United States and internationally "demonstrates that women police officers use a style of policing that relies less on physical strength, can effectively de-escalate potentially violent confrontations, are less likely to become involved in the use of force, and respond effectively to violence against women" (p. 1).



Recruiting and hiring for a diverse law enforcement agency is not just about social equity or inclusiveness, it is about effectively responding to the changing demographics both within the workforce and within the communities. Having a representative police force brings a wider range of perspectives and experiences that are reflective of the community. Officers of color can assist in making connections with communities of color, educate other officers about their communities, and interpret for their respective communities. The executive director of the Hispanic American Police Command Officer Association, Anthony Chapa, asserted that "one benefit of diversity is to avoid the perception of discrimination" (Sullivan & Gillum, 2014, para. 15). Diversity helps build trust in communities, especially in agencies that have a long history of mistrust.

Many times, police agencies state that they are committed to diversifying their workforce, but, run into obstacles. For example, hiring and promoting females and people of color may be complicated by civil service rules, labor contracts, internal hiring versus outside hiring rules and procedures, and veteran's preference. Also, candidates of color may be discouraged from pursuing a police career due to the negative relations that police have with communities of color.

Relevant Events

Poor police community relations is not a recent issue in U.S. policing. Since the 1960s, there have been many examples of racial unrest, riots, and civil disobedience, which have been instrumental in moving policing from the professional model to the community policing or service model. These examples include the cities in the Kerner Commission's *Report of the National Advisory Commission on Civil Disorders* (1968), Tampa, Cincinnati, Atlanta, Newark, Northern New Jersey, and Detroit; the riots in Los Angeles after Rodney King; and most recently, the riots in Baltimore, and Ferguson. Recruiting women and members of diverse



groups is also not a recent issue for law enforcement in the United States (Shusta et al., 2011, p. 82). The U.S. National Advisory Commission on Civil Disorders (1968) recommended that in order to address institutional racism by police, more African American or minority officers should be hired and promoted. The Kerner Commission, formed because of the Watts riots, found that there was a significant problem with the lack of African American officers and recommended improving minority hiring and promotion policies and procedures (U.S. National Advisory Commission on Civil Disorders, 1968; Shusta et al., 2011, p. 82). The Warren Christopher Commission Report on the Los Angeles Police Department that followed the Rodney King riots found bias and racism with the police department (Independent Commission on the Los Angeles Police Department, 1991; Shusta et al., 2011, p. 82). Its recommendations included improving the "hiring and promotion process that would benefit all groups" (Shusta et al., 2011, p. 82).

Many of the more recent events involving the deaths of Blacks in encounters with police have sparked protests by the community, including groups such as Black Lives Matter. Recent federal investigations by the Justice Department following the Ferguson demonstrations found that the police in Ferguson engaged in racially biased policing and profiling, that race influenced enforcement tactics, and that the Ferguson municipal courts operated as a revenue making scheme (Institute for Intergovernmental Research, 2015). The recommendations that came from the assessment included the need to recruit and hire more diversity, the need for training on probable cause to make stops and arrests, and a call for effective and strong police leadership to influence positive interactions with citizens and to improve the police culture. The recommendations also included the effective implementation of community-oriented and problem-oriented policing to build trust, transparency, and accountability. The culture of an



agency, its leadership, and treatment of women and people of color are vital in establishing and keeping good police-community relations.

The United States Commission on Civil Rights delineated in several reports that in order for law enforcement agencies to protect people's civil rights, two actions must be taken: "increase the recruitment and hiring of women and people of color from the communities being served and train officers not only in police work but also in community relations, cultural sensitivity, and the importance of diversity" (U.S. Commission on Civil Rights, 2000, Chapter 2, p. 1). In 1999, President Clinton and Attorney General Janet Reno invited "police executives, union representatives, academic experts, civil rights and community leaders" to participate in a conference, Strengthening Police-Community Relationships, to "identify police practices that build trust, enhance police accountability, and reduce police misconduct" (U.S. Department of Justice, 2001, p. 1). In their report, *Principles for Promoting Police Integrity*, the participants in the conference and follow-up meetings argued that for the police to reduce crime and be successful in community policing, citizens must trust their police (p. 1). The report also included the importance of recruiting, hiring, and retaining a diverse agency to improve relationships with the community, generate trust, and facilitate citizen support and cooperation (p. 18). The report further stated that "law enforcement agencies should seek to hire and retain a diverse workforce that can bring an array of backgrounds and perspectives to bear on the issues the agencies confront and the choices they make in enforcing the law" (p. 18).

Other recent events are examples of African American male involvement with the police in the U.S. where the public has called for the resignation of the police chief and city officials. For example, the Chicago Police Superintendent was terminated, and the Baltimore Police Chief and the Ferguson Police Chief stepped down because of fractured trust in the police and police



leadership due to racial profiling, aggressive street tactics, and excessive force against people of color. This underscores the importance of leadership qualities and attributes in shaping accountable and responsive law enforcement agencies and gaining the trust from the communities they serve. This also emphasizes the need for leadership at the local level to initiate change and bring qualified women and people of color into law enforcement agencies by examining their application process to ensure that it is related to the tasks officers perform on a daily basis.

Multicultural Communities and the Challenges for Law Enforcement

Demographic changes in the make-up of communities have impacted "the composition of the law enforcement workforce and the people with whom officers make contact" (Shusta et al., 2011, p. xiv). Police agencies must be responsive to the changing make-up within their agencies and in the communities in which they serve. Population changes that result in increased diversity call for officers who can work with different populations of people (Shusta et al., p. 100). The nature of policing has changed, requiring a broader variety of tasks and a more complex set of skills (p. 100). These skills include communicating with the public, co-workers, and people in other agencies; oral and written communication skills; the ability to solve problems; understanding of people from different cultures and backgrounds; the ability to use technology; and the ability to adapt to change. The adoption of a community-oriented policing model calls for police agencies to be more representative and responsive to the communities they serve (p. 100). The relationship between the community-oriented policing approach and the complexities of police work, as well as the relationship between a representative police force and communityoriented policing, are important. It is also important to understand how a representative police force impacts police performance. This is consistent with the philosophy of community-oriented



policing, specifically, how community-oriented policing can increase public trust and cooperation, and can decrease complaints, excessive use of force, aggressive tactics, biased policing, and racial profiling. Strained relationships between the community and police have been long documented. The importance of recruiting and hiring female officers and officers of color that are reflective of the make-up of the communities in which the police serve is more applicable today because of the evolving demographics throughout cities in the United States.

Purpose of the Study

The purpose of this research is to describe how one police agency in the Midwest recruits, hires, and retains diversity within the agency that is representative of the communities it serves. This study is a descriptive, single case study using triangulation of multiple methods, both quantitative and qualitative, and multiple data sources.

The main theoretical and conceptual framework draws from representative bureaucracy theory. The theory of social equity and the concepts of racial and gender equity also apply to hiring for diversity within police agencies and to the quality of service delivered to the public. In addition to the goal of equality and equal opportunity for the position of police officer, it is important to emphasize the importance of recruiting and hiring the most qualified candidates to be police officers. Hiring more female officers and officers of color is not in opposition to enhancing the applicant pool. Hiring the most competent and qualified candidates to meet the complexities of policing and the challenges of policing a multicultural society, is consistent with having a representative police agency to engage in community policing.

Training for police officers in diversity and cultural awareness is not enough. In addition, it is important to examine why people may not consider law enforcement as a career possibility and to reach out to potential candidates starting in middle school. While this study



acknowledges additional efforts to increase diverse representation in law enforcement need to be made, it contributes to these efforts by addressing recruiting, hiring, and retaining diversity within law enforcement agencies. There is a need to diversify police agencies and improve police services to all segments of the community.

Chapter II lays out a logical theoretical and conceptual framework and identifies what variables from the literature are relevant and applicable for describing how the agency selected for the case study recruits, hires, and retains diversity within its agency. This helps to refine theory and concepts, and to contextualize police agency recruitment and hiring for diversity.

Research Question, Seven Sub-questions, and Four Data Sets

This study investigates one primary research question and seven sub-questions to identify and explore factors important in recruiting, hiring, and retaining diversity in the organization.

Primary Research Question

How does one law enforcement agency in the Midwest recruit, hire, and retain full-time police officers that are representative of the communities it serves?

Sub-question #1

What is the gender and racial/ethnic composition of officers?

The first set of data, composition data, is analyzed to address this sub-question and consists of the percentages of full-time police officers for all ranks for each year from 2009 to 2017 by gender and ethnic group.

Sub-question #2

What is the gender and racial/ethnic composition of applicants for each step of the application process?



The second set of data, applicant data, is analyzed to address this sub-question and consists of the frequencies and percentages of all applicants who applied for full-time police officer from 2009 to 2016 by each stage of the hiring process, gender, and ethnic group. There was a total of 13 job postings, seven postings open to all applicants (external and internal applicants) and six postings open to only internal employees.

Sub-question #3

What are the turnover rates, reasons for separation, length of service, gender, and race/ethnicity for officers who separated?

The third set of data, separation data, is analyzed to address this sub-question and consists of the frequencies and percentages of all full-time police officers who separated from the agency from 2010 to mid-2017, by turnover rates, reason for separation, length of service, gender, and ethnic group.

Sub-question #4

What practices are used by the agency for recruiting, hiring, and retaining diversity?

The applicant data and interview data are analyzed to address this sub-question. The interview data consists of semi-structured interviews with persons responsible for implementation and oversight of recruiting, hiring, and retention practices at the case agency.

Sub-question #5

What factors are important for recruiting, hiring, and retaining diversity within the agency?

The applicant data, separation data, and interview data are analyzed to address this subquestion.



Sub-question #6

What characteristics are important in leadership for recruiting, hiring, and retaining diversity within the agency?

The interview data are analyzed to address this sub-question.

Sub-question #7

What modifications in the traditional recruiting, selection, and hiring processes are needed to recruit and hire for a diverse police agency?

The applicant data and interview data are analyzed to address this sub-question.

Significance of the Study

This research is intended to fill gaps identified in the review of literature relating to recruiting, hiring, and retaining diversity in law enforcement agencies. This study examines the various practices, factors, and variables of one police agency that recruits, hires, and retains diversity that is reflective of the communities it serves and makes several contributions to the literature. This study examines full-time police officer composition data over a nine-year period from 2009 to 2017 and application data over an eight-year period from 2009 through 2016. This includes examining data before and after a change in command staff in 2012 to assess the impact of the changes in recruiting and hiring practices. In addition, this study examines each step in the application process to determine if certain steps in the application process result in different passing rates for particular gender and/or ethnic groups.

The separation data from 2010 through mid-2017 provides information over an eight-year period and is also more current than the most recent 2008 data found in prior literature, which includes different and/or additional factors that may influence police officer separation from 2008 to mid-2017. This study also contributes to existing research by addressing factors that



may affect early separation of female officers and officers of color. This study benefits from collection and triangulation of four data sets and by conducting qualitative in-depth interviews to capture the perspectives of those responsible for the implementation and oversight of recruiting, hiring, and retention practices in the context of a law enforcement agency. The interviews provided validation to the three quantitative data sets. The interviews also add detailed descriptions and explanations of the steps in the application process, information regarding the changes implemented in the application process, and factors perceived to influence recruitment, hiring, and retention. The interviews also provide richness in detail and capture the complexity and a combination of factors involved. This study also contributes to the literature by increasing our understanding of representative bureaucracy by extending and offering new insights to the theory, concepts, and prior literature, and contextualizing what factors influence recruiting, hiring, and retaining representative police officers.

Interchangeability of Terms

Diversity

Although diversity broadly includes gender, race, ethnicity, national origin, culture, languages, religion, sexual orientation, age, disability, homelessness, and people with other diverse backgrounds and experiences, this study only addresses gender and race/ethnicity. This research will use the following terms when discussing race and ethnicity: race, ethnicity, ethnic groups, diversity, Whites, African Americans, Blacks, Hispanics, Latinos, American Indians, Native Americans, Alaskan Natives, Asians, Pacific Islanders, non-Whites, people of color, officers of color, communities of color, minorities, and underrepresented groups or populations.



Representativeness/Diversity/Reflectiveness

This study will use the terms representative, diversity, and reflective, interchangeably.

Summary

This chapter has laid out the statement of the problem, the background leading up to the problem, why it is a significant problem, the benefits of having law enforcement agencies that are representative of the people they serve, the purpose of the study, and the research questions and four data sets of the study.

Organization of the Chapters

Chapter II outlines important social events, historical content, and issues that lay the foundation for this research. The chapter discusses the status of female police officers and officers of color in the U.S. and the benefits of a diverse police agency. The chapter also defines and describes the theoretical and conceptual framework that guides the study, representative bureaucracy theory and social equity theory. The chapter also identifies factors and variables from prior research that influence passive representation for female police officers and police officers of color and describes how passive representation is translated into active representation in terms of police performance outcomes, public and co-worker attitudes, and the police subculture. Lastly, the chapter discusses police officer separation.

Chapter III explains and describes the methodology for this study. Chapter IV presents the analysis of the three quantitative data collection methods used for this mixed methods case study. Chapter V then presents the analysis of the qualitative interviews. Finally, Chapter VI presents discussion, conclusions, and implications of the results from the four data sets, contributions of this study to the literature, comparisons of the results of this study to prior research, limitations of the study, and recommendations for future research.



CHAPTER II

REVIEW OF LITERATURE

This chapter discusses important social events, historical content, and issues that lay the foundation for this research. This chapter also discusses community-oriented policing, the status of female police officers and officers of color in the U.S., and the benefits of a diverse police agency. This chapter then defines and describes the theoretical and conceptual framework that guides this study, including representative bureaucracy theory, passive representation and active representation, and social equity theory. Discussed next are the factors and variables from prior research that influence passive representation for female police officers and police officers of color. This chapter then describes the factors and variables from prior research that translate passive representation into active representation in terms of the impact of passive representation on police performance outcomes, public and co-worker attitudes, and the police subculture. Discussed last is police officer separation.

Important Social Events and Historical Context at the National Level

"Many police reformers have recommended increased employment of minority and female officers, not as an end in itself, but as a means to general police reform" (Walker, 1985, p. 563). Literature regarding policing in the 1960s identified the police subculture as a source of many of the problems in policing. Many reformers asserted that minority officers "would be less prejudiced, more sensitive to minority group citizens, would serve to sensitize white officers, and by their very presence undermine the public image of the department as an all-white institution"



(p. 563). Some reformers asserted that women officers would improve police-community relations by not being as aggressive, not fulfilling the "macho" image of the police, and their ability to "defuse volatile situations and provoke less hostility than men" (p. 563).

Social Changes

External demands by "community or special interest groups within the community can pressure law enforcement agencies to diversify their workforce" (Schuck, 2014, p. 59). A current example of a group that affects police agencies is Black Lives Matter, who has protested the actions of law enforcement officers regarding deaths of Black men involving the police, and demanded more discipline against officers, including firing officers. Other examples of external pressure are consent decrees, court orders, settlements, and lawsuits regarding discriminatory hiring and promotion practices in police agencies (p. 59). A study by the National Center for Women and Policing (2003b) found that consent decrees and court-orders lead to greater representation for women (pp. 59-60). Schuck (2014) also found in her study of over 4,000 law enforcement agencies that consent decrees were associated with more female officer representation (p. 66). Consent decrees are settlements reached by parties involved in lawsuits and are one example of external pressure to increase diversity.

The Police Recruitment and Hiring Process

Several authorities, including Shusta et al. (2011) and Dempsey and Forst (2016), have cited major shifts in law enforcement employment trends in the past couple of decades resulting in many agencies having difficulty in recruiting qualified police candidates due to shifting economic factors, as well as social and demographic changes. Koper (2004), in the report *Hiring and Keeping Police Officers*, and Dempsey and Forst (2016) offered the following possible explanations for the lack of qualified candidates: the good economy that attracts candidates into



the private sector, increased education requirements, the higher attrition rate due to the retirement of baby boomers, the deployment of candidates in Iraq and Afghanistan, and the homeland security industry enticing candidates away from law enforcement. Dempsey and Forst (2016) also noted that the recession that began around 2008 helped recruitment efforts in policing but budget cuts also left police jobs vacant.

Police agencies set selection standards and qualifications for police officer positions, which are influenced by the minimum selection standards set by their state's regulatory board, their union contract, the rules and regulations of their Human Resources office, and the rules and regulations set by their Civil Service Commissions. Standards for law enforcement applicants set by federal, state, and local law enforcement agencies screen for age, education, drug use, criminal history, and physical and mental fitness requirements. The selection process includes written aptitude tests, personal interviews, physical agility tests, background checks, credit checks, physical agility tests, psychological evaluations, and in some states polygraph tests (Shusta et al., 2011; Dempsey & Forst, 2016). Civil service regulations govern the selection of police officers in many states, with the size of municipalities dictating whether civil service is required or optional (Dempsey & Forst, 2013; Stevens, 2018). Stevens (2018) described civil service regulations as "the rules and regulations that govern classified employment linked to specific occupations ..., such as police officers. Rules have the force and effect of law. Regulations implement the rules issued by an official commission. U.S. state and local governments often have competitive civil service systems" (p. 226). Civil service regulations were instituted to "ensure that personnel decisions would be based on objective criteria as opposed to favoritism, biases, and political influence" and "limits the power of police command in the hiring, promotion, and firing processes" (p. 226).



Traditional recruitment and hiring practices use models that emphasize the law enforcement aspect and not the service aspect of policing. Many of these standards were challenged due to their lack of relating to the performance of the tasks performed by police officers, resulting in changes to the standards (Shusta et al., 2011, p. 84). Dempsey and Forst (2016) stressed that "a crucial element of the police selection process is that each step be defensible in court and have validity to the job performance of a police officer" (p. 115). Police agencies need to assess whether all of their hiring requirements are predictive of key job outcomes and do not unnecessarily disqualify applicants from protected groups at a disproportionate rate.

The hiring process needs to be modified to meet the changing demands and role of policing, and to incorporate innovations in recruiting and hiring candidates suited to be police officers. Dempsey and Forst (2016) noted that "in recent decades these standards have changed to allow a greater number of females and minorities to become police officers, but they are still more stringent than standards in most other professions" (p. 107). It is important to note that this does not equate to lowering the standards, instead, the standards need to change to reflect the skills and qualifications necessary to perform the tasks police perform on a daily basis. During the past several decades, the tasks that police officers perform have expanded to include a broad variety of tasks and the skills necessary to provide services in a pluralistic society and engage in community-oriented and problem-oriented policing. Recruiting candidates with a variety of diverse backgrounds and skills increases the pool of qualified and competent candidates for police officers beyond the traditional pool of candidates (Shusta et al., 2011, p. 89).



A History of the Issue and Problem / President's Commissions and Task Forces, and Federal Assessments and Investigations

Since the 1960s, there have been many examples of racial unrest, riots, and civil disobedience that have been instrumental in moving policing from the professional model to the community policing or service, model. Recruiting women and members of diverse groups is not a recent issue for law enforcement in the United States (Shusta et al., 2011, p. 82). Several national Commissions have recommended increasing the employment of minority and female police officers (U.S. National Advisory Commission on Civil Disorders, 1968; U.S. National Advisory Commission on Criminal Justice Standards and Goals, 1973; U.S. President's Commission on Law Enforcement and Administration of Justice, 1967). Walker (1985) emphasized that almost every national report on the police since the mid-1940s (U.S. National Advisory Commission on Civil Disorder, 1968; U.S. National Advisory Commission on Criminal Justice Standards and Goals, 1973; U.S. President's Commission on Law Enforcement and Administration of Justice, 1967) and Bloch and Anderson's (1974) research has overtly recommended hiring more women officers.

The Kerner Commission – Watts riots. The President's Crime Commission Report in 1967 recommended that to address institutional racism by police, more African American or officers of color should be hired and promoted (Shusta et al., 2011, p. 8; U.S. President's Commission on Law Enforcement and Administration of Justice, 1968). The report on the Watts riots by the Kerner Commission concluded that the lack of African American officers was a significant problem and recommended improving minority hiring and promotion policies and procedures (Shusta et al., 2011, p. 82; U.S. President's Commission on Law Enforcement and Administration of Justice, 1967). The U.S. President's Commission on Law Enforcement and



Administration of Justice (1967) identified that hiring African American officers was a critical step in improving relationships between the police and minority communities, which would lead to better control of urban crime.

The Christopher Commission – Rodney King. The Warner Christopher Commission Report on the Los Angeles Police Department following the "Rodney King riots" found bias and racism within the Department (Independent Commission on the Los Angeles Police Department, 1991; Shusta et al., 2011, p. 82). Its recommendations included improving the "hiring and promotion process that would benefit all groups" (Shusta et al., 2011, p. 82). The Rodney King civil rights case in Los Angeles urged many people to re-evaluate the causes of urban social unrest and poor police-community relations and reaffirmed that a main cause of the social unrest was the lack of representation of women and minorities in local police agencies (Kim & Mengistu, 1994; Independent Commission on the Los Angeles Police Department, 1991).

U.S. Commission on Civil Rights Reports. The United States Commission on Civil Rights (2000) delineated in several of its reports that in order for law enforcement agencies to protect people's civil rights, two actions must be taken: "increase the recruitment and hiring of women and people of color from the communities being served and train officers not only in police work but also in community relations, cultural sensitivity, and the importance of diversity" (2000, Chapter 2, p. 1). This again illustrated a connection between community-oriented and problem-oriented policing and attracting qualified people to perform community service tasks and safeguard people's rights.

Principles for Promoting Police Integrity. President Clinton and Attorney General Janet Reno invited "police executives, union representatives, academic experts, civil rights and community leaders" to attend a conference, "Strengthening Police-Community Relationships" to



"identify police practices that build trust, enhance police accountability, and reduce police misconduct" (U.S. Department of Justice, 2001, p. 1). The results of the conference and follow-up meetings were a report, *Principles for Promoting Police Integrity*. The report argued that to reduce crime and be successful in community policing, citizens must trust their police (p. 1). The report also asserted the importance of recruiting, hiring, and retaining a diverse agency, improving relationships with the community, generating trust, and facilitating citizen support and cooperation (p. 18). The report further stated that "law enforcement agencies should seek to hire and retain a diverse workforce that can bring an array of backgrounds and perspectives to bear on the issues the agencies confront and the choices they make in enforcing the law" (p. 18).

The President's Task Force on 21st Century Policing. The Final Report of the President's Task Force on 21st Century Policing was created to identify best practices in reducing crime and improving trust between the police and the communities they serve (President's Task Force on 21st Century Policing, 2015). The task force asserted that trust "is essential in a democracy. Trust is key to the stability of our communities, the integrity of our criminal justice system, and the safe and effective delivery of police services" (p. 1). The task force identified six main pillars with building trust and legitimacy as the first pillar. It also emphasized that law enforcement agencies should endeavor to develop a broad range of diversity within their agencies to increase understanding and effectively deal with communities (p. 2).

The After-action Assessment of the Police Response in Ferguson, Missouri. The After-action Assessment of the Police Response to the August 2014 Demonstrations in Ferguson, Missouri (Institute for Intergovernmental Research, 2015) found that the protests in Ferguson were motivated by police brutality in the four main agencies who responded: the Ferguson Police Department, the St. Louis County Police Department, the St. Louis Metropolitan Police

Department, and the Missouri State Highway Patrol. The report also concluded that the percentage of African American officers in the Ferguson Police Department, the St. Louis County Police Department, and the Missouri State Highway Patrol were not reflective of the percentage of African Americans in the population (Institute for Intergovernmental Research). Table 1 shows that the St. Louis Metropolitan Police Department had the highest percentage of African American officers (33.4%), which was about 15 times that of the Missouri State Highway Patrol (2.6%), over four times that of the Ferguson Police Department (7.4%), and over three times that of the St. Louis County Police Department (10.2%) (Institute for Intergovernmental Research). Regarding female officers, Table 1 shows that the Ferguson Police Department had 5.6% female officers and the Missouri State Highway Patrol had 5.1% female officers (Institute for Intergovernmental Research). Both of these agencies were less than half of the national average of 12% percent for female officers in the U.S (Langton, 2010; Reaves, 2015). However, St. Louis County Police Department had 13.1% female officers and St. Louis Metropolitan Police Department had 15.9% female officers, which are both above the national average of 12% female officers (Institute for Intergovernmental Research, 2015, pp. 138-139).

Table 1. Percent of African American Police Officers, African Americans in Population, and Female Police Officers for the Four Agencies Policing Ferguson, Missouri.

Agency	% African American	% African Americans	% Female Officers
	Officers	in Population	
Ferguson Police	7.4%	67.4%	5.6%
Department			
St. Louis County	10.2%	23.7%	13.1%
Police Department			
St. Louis Metropolitan	33.4%	49.2%	15.9%
Police Department			
Missouri State	2.6%	Includes areas in other	5.1%
Highway Patrol		three agencies	



The results of the Department of Justice report concluded that officers had historically used excessive force on African Americans at a much higher and disproportionate rate than on Whites; that officers had displayed a pattern of racial bias and profiling, and unconstitutional policing; that officers did not know the basic probable cause standards for making stops and arrests; that city and police leadership compelled officers to write citations; and that the local courts targeted African Americans to feed Ferguson's budget (Institute for Intergovernmental Research, 2015). The 26 recommendations in the report from the Justice Department included giving raises to officers to attract and retain qualified people, implementing a strong system of community policing, and establishing new hiring practices to recruit minority officers (Institute for Intergovernmental Research). The report also recommended a consent decree calling for major reforms, including improving training, leadership, and the culture to generate community policing and build trust in police-community relationships (Institute for Intergovernmental Research).

Four Models of Policing

J. Greene (2000) has discussed four different models of contemporary policing: traditional policing, community policing, problem-oriented policing, and zero-tolerance policing (pp. 299-320) and noted that the growth of community policing and problem-oriented policing are vital in increasing "the effectiveness of police efforts in communities and as means of reforming police organizations" (p. 299). He denotes that there is some overlap of the elements in these models. The first model, traditional policing, resulted from the goal of early reform to disentangle the police from politics, which focused on reactive law enforcement or crime fighting and crime control, rather than on order maintenance or provision of general services (pp. 310-311). J. Greene also noted that in this model, line officers have a great deal of discretion,



are not closely supervised, and use "aggressive street tactics coupled with broad application of the criminal law" (p. 310). Values that accompany traditional policing include secrecy, skepticism, cynicism, and dislike for the public in general (p. 310). Measures of success include arrest and crime rates, particularly with serious Part I crimes: homicide, rape, robbery, aggravated assault, burglary, larceny-theft, motor vehicle theft, and arson (p. 310).

Skolnick and Bayley's (1988) stated "that community policing is a coherent concept grounded on the notion that, together, police and public are more effective and more humane coproducers of safety and public order than are the police alone" (p. 1). Skolnick and Bayley also described the core elements of community policing as including "community-based crime prevention, reorientation of patrol to stress nonemergency service, increased accountability to the public, decentralization of command, and sometimes, civilianization" (p. 1). J. Greene (2000) also discussed that community policing expanded to include "public safety, crime, fear of crime, and community quality of life" (p. 312). Mastrofski et al. (1995) observed the following on community policing:

"Community building" focuses on crime prevention, victim assistance, and building greater rapport with racial minorities. The police strengthen citizens' capacity and resolve to resist crime and recover from it. This requires positive relationships with those "invested" in the neighborhood. Crime prevention and victim assistance do not involve law enforcement directly. To the extent that a community policing program concentrates on community building, it de-emphasizes law enforcement activities, an object of some criticism. (pp. 540-541)

This emphasizes that the police must work with the community and other public and private agencies regarding "quality-of-life issues" (J. Greene, 2000, p. 313). It also stresses "making



police agencies less bureaucratic, specialized, and hierarchical", and more decentralized (p. 314). Measures of success include calls for service, reduction in fear, use of public places, community contacts, community health, community volunteerism, new businesses starting up, home ownership increases, safer neighborhoods, and surveying citizens regarding their perceptions on safety and the police (pp. 311 & 314).

Problem-oriented policing is more focused than community policing and is problem specific that addresses the symptoms of crime rather than the underlying causes of crime. J. Greene (2000) describes this as the police being more analytic in their interventions to "address persistent community crime, disorder, and fear problems" (p. 315). One model police agencies use in problem-oriented policing is the SARA model: scan, analyze, respond, and assess. Measures of success for problem-oriented policing include the goal of absence, reduction, or the problem moving to another geographic location (pp. 311 & 316).

The last model of policing that J. Greene (2000) described is zero-tolerance policing. This is a variation of problem-oriented policing, and became well-known with Chief William Bratton, New York City Police, when its officers used aggressive street tactics to address disorder, neighborhood decay, and little problems before they became big problems. The model is also associated with Wilson and Kelling's (1982) "Broken Windows" thesis. Measures of success include counting stops of vehicles and pedestrians, arrests, and location-specific reductions in targeted activity (pp. 311 & 320).

Community-Oriented and Problem-Oriented Policing

As the police role, function, and tasks have evolved, a police officer's skill set required to successfully fulfill the role of police officer has expanded. This includes the skills necessary to meet the changing demographics of the communities. Findings from research show that



community-oriented and problem-oriented policing have had some impact on crime. However, they have a larger impact "on the quality of interaction between the police and the public" (J. Greene, 2000, p. 299). J. Greene notes that in agencies that practice community-oriented or problem-oriented policing, police officers perceive their roles and job tasks as changing for the better, and that job satisfaction is improving. J. Greene also noted that the broad label of community policing began in the mid-1980s and "emphasizes greater interaction with the community in resolving persistent neighborhood crime and disorder problems" (p. 308).

The policies and practices of a law enforcement agency in general, including an agency that practices a community-oriented style of policing, can greatly impact the representation of more female officers and officers of color in police departments (Schuck, 2014, p. 58). Wilson and Kelling (1982) defined community policing as a philosophy where police agencies and the community work together in solving problems, and empowering and engaging the community to work with the police. Schuck (2014) states that "one of the central tenets during the community-policing era has been the pursuit of workforce diversity" (p. 58). This is important because officers of color can help their community work with the police, serve as an interpreter in languages other than English to improve communication between diverse communities and police, and help in improving their agency's credibility. This also has a positive impact on the community because it shows that the agency is committed to equality and access within the agency and for residents (p. 58). She also found in her study of over 4,000 agencies that law enforcement agencies who are committed to and practice community policing have higher levels of female officer representation (pp. 54 & 59).

The Department of Justice's "Office of Community Oriented Policing Services (COPS) defines three major components of community policing: (1) collaborative police-community



partnerships; (2) support from agency management, structure, personnel, and information systems; and (3) a problem-solving process that develops and rigorously evaluates effective responses" (Reaves, 2015, p. 8). The Law Enforcement Management and Administrative Statistics (LEMAS) survey started collecting data in 1987 on "community policing personnel, activities, and policies in response to Department of Justice (DOJ) funding programs that were impacting local law enforcement agencies" (p. 8). The 2013 LEMAS survey found that about 70% of local police departments and 90% of departments serving 25,000 or more residents had a mission statement that included community-oriented policing (p. 8). In summary, there is a strong connection between community-oriented and problem-oriented policing and recruiting, hiring, and retaining qualified officers, including women officers and officers of color, who can perform the increased complexity of skills needed for policing and policing reforms.

Percentages of Female Police Officers and Police Officers of Color by Type of Agency and Size of Agency

This section reviews the status of female police officers and police officers of color and their progress toward equal employment opportunity in the United States. The changing external environment and internal culture of police agencies influence the employment of diverse police officers (Walker, 1985, p. 556). Also, there are great differences in the numbers and percentages of female officers and officers of color between agencies. Morabito and Shelley (2015) note that there is great disparity in the representation of female officers and officers of color across police agencies, ranging from agencies having no women officers or officers of color to agencies that are representative of the communities which they serve (p. 331). In their study on factors that contribute to representation of diverse law enforcement officers using the 1987 LEMAS survey data, Kim and Mengistu (1994) found that White males continued to possess disproportionately



more sworn positions in most of the police agencies in their study (p. 161). Steel and Lovrich (1987) have attributed the increases in the number of female officers to the Equal Employment Opportunity Act of 1972, which was an amendment to Title VII of the Civil Rights Act of 1964. Many cities have documented racial and gender disparities in city hiring, the majority of police officers are White males.

Status of Females in Policing

Although the percentage of women in policing has increased since the mid-1970s, the rates of increase have slowed down. Martin (1993) stated that for agencies in large cities, the representation of female officers increased from 4.6% to 12.6% between 1980 and 1990. The National Center for Women and Policing (2002) (NCWP) conducted a study on women in large, small, and rural agencies in the U.S. (p. 2). Overall, their research demonstrated that the number of women sworn officers is small and the increase that was observed in earlier years has "stalled or even reversed" (p. 2). For all sworn law enforcement officers in the U.S. in 2001, women comprised 11.2%, which was significantly less than women in the labor force at 46.5% (p. 2). The NCWP reported in 2001 that women comprised 12.7% of total officers in large agencies (100 or more officers), compared to 9% in 1990 (p. 2). In 2001, for small and rural agencies (less than 100 officers), women comprised 8.1% of total sworn officers (p. 2). In 2001, within large agencies, sworn women comprised 7.3% of top command positions, 9.6% of supervisory positions, and 13.5% of line operation positions (p. 4). It is interesting to note in 2001, that 55.9% of the large agencies surveyed reported no women in top command positions and 97.4% of small and rural agencies had no women in top command positions (p. 4). For state agencies in 2001, 5.9% of the officers were women, for municipal agencies, 14.2% were women, and for county agencies, 13.9% were women (p. 4). The NCWP also noted that one possible explanation



for the stall or decline in women representation was the decrease in the number of consent decrees requiring the hiring and promotion of women and minorities (p. 5). Among the agencies surveyed, eight consent decrees expired between 1999 and 2002, and only six consent decrees were executed in the 1990s, with two consent decrees executed since 1995 (p. 5).

Langton (2010), writing on the status and trends of female sworn officers in federal, state, and local agencies from 1987 to 2008, examined data on Federal Law Enforcement Officers (FLEO) from the Bureau of Justice Statistics' Census and the 2008 LEMAS surveys (p. 1). In 2008, the Office of Inspectors General had the highest percentage, 25%, of female sworn law enforcement officers (p. 1). Among other federal law enforcement agencies, large agencies (more than 500 sworn officers) had 21% women, medium sized agencies (100 to 500 officers) had 9% women, and small agencies (less than 100 officers) had 10% women (p. 1). In 2008, 19% of Federal Bureau of Investigations agents were women (p. 1).

For local police agencies, the percent of female officers grew from 7.6% in 1987 to almost 12% in 2007 (Langton, 2010, p. 3). For sheriff's offices, the percent of female officers decreased from a high of 15.6% in 1997 to 11.2% in 2007 (p. 3). Langton attributed this sharp decrease while other types of agencies increased to the 2007 survey excluding sheriff's offices that did not perform law enforcement functions. It is important to note that women who work in sheriff's offices may be assigned to the jail, therefore, they are not performing typical patrol duties. In large (more than 100 officers) local police agencies in 2007, women made up approximately 15% of sworn officers (p. 2). In local police agencies with 2,000 or more officers, women made up an average of 18% of total sworn officers. In large sheriff's offices, women made up approximately 13% of total sworn officers (p. 3). In small local agencies (1 to 10 officers), women made up 6% of total sworn officers (p. 3). In small sheriff's offices, women



made up 4% of total sworn officers (p. 3), and for state police agencies, the percent of female officers grew from 3.8% in 1987 to 6.5% in 2007 (p. 3).

The LEMAS survey data for 2013 found approximately 12% of officers in local police departments in the U.S. were female compared to 8% in 1987 (Reaves, 2015; Stevens, 2018). In 2013, the percentage of female officers (17%) in jurisdictions with 250,000 or more residents was more than twice that (7%) in jurisdictions with fewer than 25,000 residents (Reaves, 2015, p. 4). In 2013, the LEMAS survey collected data on female supervisors and managers for the first time and reported that 9.5% of first-line supervisors were female (p. 5). For departments serving 250,000 or more residents, 15% of first-line supervisors were females, compared to departments serving less than 50,000 residents, 6% of first-line supervisors were females (p. 5). It is interesting to note that about 3% of local police chiefs were female with 7% in departments serving 250,000 or more residents (p. 5).

Status of Officers of Color

Walker and Katz (2002) described the increase in African American and Hispanic representation in policing in the U.S. African Americans accounted for only 3.6% of all sworn officers in the mid-1960s, 6% in 1973, 7.6% in 1982, and 10% in 1997 (pp. 402-403). Hispanics accounted for 4.8% of all sworn officers in 1988, 6.2% in 1993, and 7% in 1997 (p. 403). Reaves and Goldberg (1999) reported that by 1997, African Americans comprised 10% of local full-time sworn officers and 7% of state police officers (p. xiii). In large cities, Black representation increased from 18.4% in 1990 to 20.1% in 2000 (Reaves & Hickman, 2002, p. 3). H. Greene (2003) stated that there is "an unprecedented number of African Americans serving in leadership positions in local agencies" and "there are also more African American female police officers than White female police officers in some urban agencies" (p. 208). Despite the



progress made in increasing the number of African American police officers, H. Greene also noted that "the ratio of black officer to black citizens continued to be low in large cities" and that Blacks were underrepresented in federal agencies, 11.7% in 2000 (p. 208). African American representation had increased to 11.7% in 2000 and to almost 12% by 2007 (H. Greene).

In local police departments, the LEMAS survey found that officers of color comprised 27% of police officers in 2013 compared to 14.6% in 1987 (Reaves, 2015). For African American officers, the survey found 12% in 2013 compared to 9% in 1987 (p. 5). For Hispanic/Latino officers, the survey found 11.6% in 2013 compared to 4.5% in 1987 (p. 6). For the "Other" category in race/ethnicity, the survey found 3% in 2013 compared to .8% in 1987 (p. 6). The LEMAS survey was consistent with other studies in finding that "departments serving larger jurisdictions were more diverse than departments serving smaller jurisdictions" (p. 6).

The Effects of Changing Demographics within Police Agencies

Establishing that the makeup of police agencies has changed over the past few decades, Sklansky (2006) examined the effects of the changes in police demographics using three categories (p. 1211). The first category comprised "competency effects", which are the different set of abilities that women officers and officers of color possess (Sklansky). Since the 1960s, there has been an emphasis on diversifying police agencies because officers of color have been linked to a different set of skills than those of White officers. A benchmark report by President Johnson's Commission on Law Enforcement and the Administration of Justice (1967), *The Challenge of Crime in a Free Society*, found that all-White police agencies could not "deal successfully with people whose ways of thought and action are unfamiliar" and that "lack of understanding of the problems and behaviors of minority groups is common to most police departments and is a serious deterrent to effective police work in the often turbulent



neighborhoods where these groups are segregated" (p. 107). Sklansky (2006) delineated these special competencies as "greater *understanding* of minority communities, and greater *credibility* in minority communities" (p. 1224).

Regarding female officers, it has been observed that women are equally capable in performing patrol duties, and in some instances, women may have better skills. This includes, less use of excessive force, are better at "defusing and de-escalating potential violent confrontations", are better at gaining trust and cooperation, and are more successful in responding to domestic violence calls (National Center for Women and Policing, 2003a, p. 204; Sklansky, 2006, p. 1227).

The second category identified by Sklansky (2006) is "community effects", which are how diverse representation in police agencies affects the relations they have with the community (pp. 1211-1212). Sklansky denotes that diversity within its ranks can improve credibility of the entire agency (p. 1228). This also had been noted decades previously in President Johnson's Commission on Law Enforcement and the Administration of Justice (1967), which stated that "a major, and most urgent step in the direction of improving police-community relations is recruiting more, many more, policemen from minority groups" (p. 107), and that "every section of the community has a right to expect that its aspirations and problems, its hopes and fears, are fully reflected in the police" (p. 107).

The third category identified by Sklansky (2006) is "organizational effects", which are how a diverse police agency affects the internal dynamics of the agency, including the subculture. Sklansky remarked that "the demographic transformation of American law enforcement has done much to break down the police subculture, by weakening both the occupational solidarity and the social insularity of the police" (p. 1212). He also noted that



police effectiveness had not deteriorated because of diversity within police agencies (p. 1212). Sklansky argued that of the three categories of effects, the most notable effect of increasing the representation of women officers and officers of color is on the "internal dynamics of police departments", which he further divides into three categories: "one-on-one interactions, rival trade groups, and social fragmentation" (p. 1230). Regarding one-on-one interactions, Sklansky stated that officers of color and female officers can change the attitudes and behavior of other officers they interact with on a daily basis (p. 1230). Regarding rival groups, Sklansky asserted that diverse officers can "compete for membership with the longstanding police benevolent associations, which generally serve today not only as social and fraternal organizations but also as collective bargaining agents and lobbying groups" (p. 1230). Regarding social fragmentation, Sklansky argued that this is the most important of the three categories, which he delineates as "the decline of the monolithic police subculture" (p. 1231).

The Effects of Representation of African American Police Officers

African American representation in police agencies has symbolic value and practical outcomes (H. Greene, 2003). African American police chiefs have symbolic value because of the potential in increased political power and African American officers have symbolic value for African American citizens. From a practical standpoint, African American officers have been found to improve relationships between the police and African American community, have less police brutality, and increase recruitment of officers of color (p. 207). H. Greene also points out that it is a difficult task to determine the specific effect of African American police on policing because there are numerous local, state, and federal agencies, different types of agencies, and different jurisdictions (p. 207). The impact of African American officers also varies by the individual officer, neighborhood, community, and region (p. 207).



Hiring in the Spirit of Service

Sam Walker (2015), a renown police scholar, historian, and civil rights expert, has argued that increased diversity in police agencies is one of the key qualities of a successful police agency. However, Jordan et al. (2009) found in their study, using a national survey of law enforcement officers, that only one in five law enforcement agencies implemented targeted recruitment strategies for women and people of color (p. 333). An excellent example of one of the newer useful tools for recruiting that demonstrates a major shift in recruitment philosophy is the federally funded project called Hiring in the Spirit of Service (HSS) (Scrivner, 2006; Shusta et al., 2011, p. 95). HSS stresses that the main change is to recruit applicants who are not drawn to law enforcement because of the crime fighter or tough cop image, or the aggressive, maledominated and predominantly White culture that emphasizes physical strength. Instead, who needs to be recruited are people drawn to the career because they are service minded (Scrivner, 2006; Shusta et al., 2011, pp. 95 & 101). This includes people who can solve problems and possess interpersonal and service-oriented skills, and who accept diversity within the workplace and in the community. Also emphasized by HSS is for agencies to find creative ways to revise hiring practices for police officers to be consistent with the community or service policing model (Scrivner, 2006; Shusta et al., 2011, pp. 96-97).

Theoretical and Conceptual Framework

There are many factors that contribute to why some agencies achieve diversity and other agencies do not. Jordan et al. (2009) assert that "... in addition to the macro-level factors gaining widespread attention, agency- and jurisdictional-level variables play an important role in the ability to recruit and hire qualified female and minority personnel" for sworn positions (p. 333).



Overview of Representative Bureaucracy Theory

The theory of representative bureaucracy has evolved since its inception in the 1940s. Groeneveld and Van de Walle (2010) argue "that representative bureaucracy is a multidimensional and changing concept" (p. 239). This section will define and describe the theory of representative bureaucracy, define the two types of representative bureaucracy that are included in the literature: passive and active representation, and include the history and research on representative bureaucracy up to the present that is relevant to this study.

According to the theory, representative bureaucracy states that public agencies can be more responsive to the public if the people who work in them are reflective of the demographic characteristics of the public they serve (Selden, 1997; Stein, 1986; Sowa & Selden, 2003). Meier, Wrinkle, and Polinard (1999) stated "representative bureaucracy suggests that if a bureaucracy is broadly representative of the public it serves, then it is more likely to make decisions that benefit that public" (p. 1026). Bradbury and Kellough (2008) added that "the theory of representative bureaucracy suggests that diversity within the public workforce, especially in terms of characteristics such as race and ethnicity, will help to ensure that the interests of diverse groups are represented in policy formulation and implementation processes" (p. 697). The theory was also reflected in the Civil Reform Act of 1978 where it reaffirmed the commitment to Affirmative Action and postulated that the civil service should reflect workforce diversity and that reducing veterans' preference would positively influence the recruitment of women and minorities (Ingraham, 1995, pp. 45, 73, & 81).

A history of the theory of representative bureaucracy is provided by Sowa and Selden (2003). They discuss that the theory originally concentrated on benefits of passive representation in a public workforce that is reflective of the characteristics of the people they serve. Sowa and



Selden state that the theory first appeared in academic literature with the work of Kingsley in 1944 that applied to the British Civil Service with the emphasis "on social class as one of the most important demographic variables" (p. 701). Levitan (1946), in applying the theory in the U.S., argued that a public workforce that has similar composition to society would bring more acceptance by the public to actions of a public agency. Sowa and Selden (2003) stated that Long (1952) expanded the theory to emphasize "the symbolic importance of a representative bureaucracy to legitimate policy to the citizenry" (p. 701). Sowa and Selden (2003) further stated that the theory was refined by Krislov (1974) and Krislov and Rosenblom (1981) that "focused on aspects of individual socialization, maintaining that the demographic backgrounds of individuals – including such characteristics of race, ethnicity, and gender – provide an early socialization experience that leads to the creation of certain values and beliefs" (p. 701). Representation would allow "citizens to feel a connection with government, to see their needs and desires reflected in the actions of government, actions that would reflect these similarly held values and beliefs" (p. 701). Long (1952) stated "it is of critical importance that the bureaucracy be both representative and democratic in composition and ethos" (p. 813). Applying Long's (1952) and Kellough's (1990) work to their study, Sowa and Selden (2003) added that representative bureaucracy would provide the public "with a symbol of equal access to the power of government and would fulfill the deficiencies these scholars believe were left by Congress and other political executives" (p. 701).

Representative bureaucracy consists of both passive representation and active representation. Though it is important to have symbolic or passive representation, scholars and researchers started to address how this can be transformed into active representation, which is defined as the benefits from a representative workforce (Sowa & Selden, 2003, p. 701). Meier et



al. (1999) describe how the theory of representative bureaucracy started out as a simple view of bureaucratic motivation and evolved to how individual values influence their decisions in different organizations (p. 1026). Thus, as Meier and Nicholson-Crotty (2006) have asserted, representative bureaucracy is "not just the composition of the bureaucracy but also the representation that representation has for policy making and policy implementation" (p. 850).

The present study uses representative bureaucracy theory to examine factors that influence passive representation of women and people of color in a Midwest police agency. Kennedy (2014), Morabito and Shelley (2015), and Selden (1997) argue that representative bureaucracy is imperative for equity in access to power, expertise in bureaucracies, a representation for different group interests, and the efficient use of resources. Kennedy (2014) asserts that representation is crucial in legitimizing the power of bureaucracies because it assures multiple perspectives go into decision making and policy implementation.

Passive representative bureaucracy. Mosher (1968) described *passive* representativeness as "the source of origin of individuals and the degree to which, collectively, they mirror the total society" (p. 12). This includes statistical measures of locality of origin (rural, urban, or suburban), family social class, race, and religion (p. 12). Selden, Brudney, and Kellough (1998) defined *passive representation* as "the representation of certain groups, particularly racial and ethnic minorities, in the public work force" (p. 717). Passive representation also includes the degree to which a bureaucracy reflects the demographics of the general population (Dolan, 2000). The existence of diversity regarding demographic characteristics, or passive representation, is an "indicator of the organization's commitment to equal opportunity and access to power and can promote the legitimacy of public bureaucracies" (Meier & Nicholson-Crotty, 2006, p. 851). This implies that passive representation has symbolic



value for "citizens to feel empowered and connected to the government" (Schuck, 2014, p. 55). The next section will include the factors and variables that influence passive representation for female police officers and police officers of color.

Factors and Variables that Influence Passive Representation of Female Police Officers and Police Officers of Color

Following is a discussion of the different variables that research starting in the 1970s identified as influencing passive representation of female law enforcement officers and officers of color, including external and internal variables, and an expansion of additional categories. In determining what factors are relevant in mostly municipal governments, Selden (1997) established that scholars examined over 50 variables in four categories: demographic, political, organizational, and economic (p. 51). It is important to note that the explanatory variables found to be important in the literature to influence passive representation on police departments are dynamic and change over time due to the changes in demographics of city populations, and political, social and economic changes. These factors also have independent, interacting, and joint effects on representation of people of color and women in U.S. police agencies. This includes dynamic and reciprocal relationships between factors and "between police agencies and the social context in which such organizations operate" (Schuck, 2014, p 56). Some scholars have placed factors that are correlated with female representation in policing into two broad categories: environmental and institutional (Schuck, 2014; Warner, Steel, & Lovrich, 1989; Zhao, Herbst, & Lovrich, 2001).

The increase of female representation in U.S. police agencies has led to organizational changes within police agencies. Zhao et al. (2001) notes that trends in American policing "emphasize elements of outreach, victim assistance, police partnership in the community, and the



philosophy of community oriented policing" (p. 244). Research by Bloch and Anderson (1974) and Steel and Lovrich (1987) have established that female officers are equally competent as male officers in traditional law enforcement, order maintenance, and service tasks and functions. The NCWP (2002) found female officers are preferred in some situations for duties that call for empathy and sensitivity rather than a threat to use physical force. The NCWP also noted that female police officers are less involved in serious misconduct and perpetual misconduct and received fewer citizen complaints. Morabito and Shelley (2015) found in their research, and highlighted in other studies, that the factors that predict female representation are not the same factors that predict representation of officers of color.

Economic Factors

Organizational growth and fiscal well-being have been cited by scholars to influence hiring and promotional opportunities (Eisinger, 1982; Warner et al., 1989). Zhao and Lovrich (1998) suggest that police officer jobs may now attract more candidates of color because many blue-collar jobs in manufacturing have vanished in many cities in the U.S. since the 1960s (p. 276).

Unemployment rates are one indicator of how the U.S. economy is doing. During a recession, unemployment rates are higher and during a good economy, unemployment rates are lower. Dempsey and Forst (2016) stated that in a good economy, there are fewer candidates for police officer jobs because police agencies "have to compete with the private sector for employees" (p. 113). In a poor economy, there are more police officer candidates because of the limited opportunities in the private sector. The following are the unemployment rates from the Labor Force Statistics for December of each year from 2009 to 2017: 9.9% in 2009, 9.3% in 2010, 8.5% in 2011, 7.9% in 2012, 6.7% in 2013, 5.6% in 2014, 5.0% in 2015, 4.7% in 2016,



and 4.1% in 2017 (U.S. Department of Labor, 2018). There was a recession starting around 2008, which resulted in fewer jobs and higher unemployment rates overall. The unemployment rate decreased to the lowest rate since the early 2000s with the 4.1% in 2017 (U.S. Department of Labor). This means that even though there are more jobs overall, there are fewer police applicants because police agencies are competing with the private sector.

Stein (1986) demonstrated that unemployment rates significantly impair representation of people of color in municipal employment. Kim and Mengistu (1994) found in their study using 1987 LEMAS survey date that unemployment rates had a negative effect on Blacks and Hispanics, however, unemployment rates significantly increased female employment in police agencies.

Environmental or Demographic Factors

Factors cited in the research under environmental factors are the pressures put on police departments stemming from the external environment. These pressures include broader social and political dynamics and changes that push police agencies to hire more women and people of color in traditionally white male dominated professions (Zhao et al., 2001).

Population of communities of color. Selden (1997) affirmed that demographic characteristics of cities affect the characteristics of city employees (p. 52). Selden concludes that "the size of the minority population is expected to be a major determinant of the racial composition of the local government work force" (p. 52). There have been numerous studies that support this hypothesis (Goode & Baldwin, 2005; Hochstedler, Regoli, & Poole, 1984; Lewis, 1989; Selden, 1997; Stein, 1986). In their study on the factors that contribute to representation of women law enforcement officers and law enforcement officers of color using the 1987 LEMAS survey data, Kim and Mengistu (1994) found that the size of the local minority



population is the single most significant explanatory variable for minority and female police officer representation (p. 174). They found that Black and Hispanic representation in policing is highly correlated with their presence in the community (pp. 161 & 174). Zhao and Lovrich's (1998) primary finding was that "the percentage of African American population is the strongest and most important contributor to the prediction of hiring rates of African American police officers" (p. 274). The increase in African American representation in city populations has also been found by other scholars to lead to higher numbers and percentages of African American police officers (Lewis, 1989; Walker, 1985).

Zhao and Lovrich (1998) offer reasons why more African Americans in the population may lead to more African American police officers. One reason is that there are more African American candidates applying for police officer positions (p. 268). Having a larger African American population can lead to more political power to elect African American mayors and/or city council members, influence personnel policies and equitable hiring and promotional practices, and hire more Black police chiefs and Black officers (Zhao & Lovrich, 1998; Walker, 1985). Zhao, He, and Lovrich (2005) conducted research to identify the key explanatory variables in predicting African American and Latino police officer representation in the U.S. using 281 police agencies in 47 states in 1993 and 1996. They found that the population of communities of color was one of the most important predictors in the employment of African American and Latino police officers in the U.S. (pp. 377 & 384). However, there have also been cases where there was a Black police chief and/or mayor where Black political power had not translated into Black police officer employment (Walker, 1985, p. 560).

City population. In studies in municipal employment, scholars have noted that city size is an essential predictor of female and people of color work force demographics (Eisinger, 1982;



Kim & Mengistu, 1994; Mladenka, 1991; Selden, 1997; Stein, 1986). Schuck (2014) identified that "two of the strongest predictors of female representation in law enforcement are the population of the jurisdiction served by the agency and the racial and ethnic diversity of the community" (p. 57). According to Schuck (2014), studies over the past 30 years "have consistently shown that female officers are more likely to be employed in agencies that serve large metropolitan jurisdictions rather than smaller municipalities or rural area" (p. 57).

Warner et al. (1989) emphasized that larger cities are more likely to have more educated and civic minded persons who are supportive of Affirmative Action programs. However, Kim and Mengistu (1994) found that size of the total population is positively correlated with employment of police officers of color, but it is negatively correlated with female officer employment (p. 173).

Larger cities traditionally have more employment opportunities and have diverse groups that make up the city population (Stein, 1986; Warner et al., 1989). In their study of 281 U.S. police agencies in 47 states in 1993 and 1996, Zhao et al. (2005) found the size of the city population, percentage of African American population, and the presence of a female mayor were positively associated with female officer representation.

Region in the United States. Region in the U.S. is also related to economic factors.

Selden (1997) and Stein (1986) contend that region of the U.S. is associated with representation of female police officers and police officers of color. Stein intimates up to the mid-1980s, that economies in the Sun Belt region (the South and West) were flourishing while economies in the Rust Belt region (Northeast or Midwest) were depressed. However, Riccucci (1986) argued that a long history of discrimination in employment in the South may counteract the effects of the thriving economy. Research conducted by Steel and Lovrich (1987) found that police agencies



in the West and South had the highest rate of increase in utilizing women in policing from 1978 to 1981, which they suggested was due to better economic conditions in the "Sun Belt" region.

Riccucci (1986) found that discriminatory private-sector employment practices in the South have impeded the employment progress of women in the South.

Slack (1987) found that African American city managers and police chiefs in the South were more likely than White city managers and police chiefs to hire African American officers. Steel and Lovrich (1987) found that police agencies in the South had higher increases of female officer employment, and that female representation in Northeast and North Central cities was much lower.

In their study of factors that contribute to representation of women law enforcement officers and law enforcement officers of color, using the 1987 LEMAS survey data, Kim and Mengistu (1994) found that Blacks are more represented in Southern police agencies. They also found that Hispanics are more employed in the Southern and Western regions, and women are better represented in Northwest U.S. agencies (p. 161).

Morabito and Shelley (2015), in their study of 1,478 police jurisdictions using data from the 2003 LEMAS surveys, the 2000 U.S. Census, and the 2003 Uniform Crime Reports (UCRs), explored factors that relate to representative bureaucracy theory as its framework. They found that police agencies in the Northeast are not as likely to have higher proportions of women officers (p. 341). For officers of color, they found that police agencies in the South were more likely to have higher proportions of officers of color and the agencies in the West are least likely (p. 344), and that police agencies in the South and West were more likely to have higher proportions of African American officers (p. 345).



Political Factors

The following are various political factors found in the literature that influence representation of female officers and officers of color.

Presence of females or people of color in public office. Many sources have cited the presence of females in elective public offices in city governments to be important, for example, female mayors and city council members (Eisinger, 1982; Warner et al., 1989). Females in elected positions can directly and indirectly impact higher rates of female officers. The positive effects of female representation in public office can directly affect females being hired "through immediate oversight of hiring practices which promote the employment of women" and indirectly "by influencing general attitudes about women working in previously male-dominated municipal occupations" (Warner et al., 1989, p. 566). Gustafson (2013) found in his sample of 181 municipal police agencies in the U.S. that the presence of people of color in political office was one of the most significant factors in affecting diversity among line officers.

Kelly (1998) notes that the literature emphasizes the difference women and people of color in legislatures and public bureaucracies make for their respective groups (p. 204). She also highlights the role of elected and appointed leaders in advocating "multiracial, multiethnic, and gender diversity" (p. 204). According to Kelly, when women and people of color are in positions to make decisions, "more women and minorities are hired, their concerns are more likely to be addressed, their material well-being rises, and their sense of the fairness and justice of the political entity improves" (p. 204). However, Warner et al. (1989) did not find that having a female mayor in the 281 cities in the United States included in their study from 1984 to 1987 increased utilization of women as police officers (p. 574).



Stein (1986) found that people elected to public office are more likely to serve the interests of the people who elected them. Eisinger (1982) suggested that a mayor of color will lead to more people of color hired in municipal jobs. The mayor may also have discretionary power in influencing local police agencies' practices and policies regarding personnel, including recruitment and establishing qualification standards and minimum acceptable test scores, which have been shown to be relevant to hiring people of color (Zhao & Lovrich, 1998, p. 270).

Zhao et al. (2005) found in their study that having a Latino mayor was correlated with increased employment of people of color (pp. 377 & 384). Lewis (1989) found that the presence of a Black mayor was one of the strongest predictors in increasing the percentage of Black officers in his study of large police departments in the United States.

Warner et al. (1989) found in their study that "women on city councils does indeed play a major independent role in estimating the level of utilization of women as police officers" (p. 574). They concluded that "more benefits are available for women in the labor force when other women are in political decision-making positions" (p. 576). This argument can also apply to people of color in the workforce when people of color are in political decision-making positions, specifically a chief law enforcement officer of color. Several studies have found that representation on city councils by people of color is significant for increasing employment for people of color (Mladenka, 1991; Selden, 1997) and that having a mayor of color increases employment opportunities for communities of color (Eisinger, 1982; Lewis, 1989; Mladenka, 1991; Selden, 1997).

Structure or type of government. Stein (1986) found in her research using survey data from 134 large U.S. cities that the nature of a city's government influences employment of people of color, by itself, and combined with autonomy of its personnel bureau, and the presence



of a mayor of color or city council members of color. Stein also asserts that the form of government may influence compliance with Affirmative Action programs for public leaders and administrators who support Affirmative Action.

Professional associations. African American police officers acknowledged that they need to organize to shape change (H. Greene, 2003, p. 212). In 1972, the National Black Police Association (NBPA) was started to pull together African American police associations with the goal of improving police agencies and relationships between communities of color and the police (p. 212). In 1976, the National Organization of Black Law Enforcement Executives (NOBLE) was formed to bring Black police executives together to address crime in urban areas (p. 212). Today, "these organizations address numerous issues confronting officers and black citizens, including brutality, crime, discrimination, police-community relations, and recruitment" and influence local and national policies, practices and programs (p. 212). NOBLE has also initiated efforts in increasing recruitment of people of color and promoting career fairs (p. 214).

Institutional or Organizational Factors

Institutional factors include "policies, procedures, and opportunities that are internal to the agency" (Schuck, 2014, p. 56). Warner et al. (1989) also found that the attitudes of hiring agents have been found to influence equal opportunity employment. This also highlights the role of leadership to the commitment and practices of a diverse police agency. The following are institutional or organizational factors cited in the literature that influence representation of female officers and officers of color.

A formal administrative structure to achieve social equity and representativeness / Affirmative Action and Equal Opportunity programs, and consent decrees. Affirmative Action programs may consist of hiring quotas or targets/goals in hiring. Today, it is more



common for police agencies to have targets and goals in hiring instead of quotas. Hochstedler et al. (1984) "found that a hiring quota was the single most important variable in explaining employment patterns" (p. 561) and stated that "the absence of a hiring quota generally spells affirmative action failure" (p. 561). Walker's (1985) data found that progress in hiring Black officers was significant in agencies that had court-ordered quotas for minorities. Walker stated that progress in Affirmative Action may be influenced by the "quality of leadership in police departments and/or other elected or appointed public officials" if the political power translates into a commitment to Affirmative Action (p. 562). Walker has noted that commitment to affirmative action is not necessarily connected to the race or gender of leadership as White male leaders may be committed to Affirmative Action and female leaders or leaders of color may not be committed to equal employment opportunity (p. 562). Lewis (1989) found in his study of Blacks in city police organizations that the case of Affirmative Action consent decrees in 1985 was one of the most powerful variables for increasing Black percentages among sworn officers (p. 257).

Warner et al. (1989) discuss that even though women have made progress since the passage of the Equal Employment Opportunity Acts of 1972, which extended the prohibition of discriminatory personnel practices by state and local governments in the 1964 Civil Rights Act, "less progress has been seen in municipal police departments" (p. 562). Based on their survey of 281 United States cities from 1984 to 1987, they found that utilization of women as police officers is due to a combination of economic conditions, the percentage of women in public office at the local level, including mayors and city council members, and "explicit administrative mechanisms established for hiring women" (p. 562). Their research suggests "that progress



towards social equity for women in such male-dominated occupations as policing is highly dependent upon a formal administrative structure established to achieve this goal" (p. 562). Warner et al. found "that change occurs only with a formal affirmative action plan or a courtenforced program" (p. 576). Their research supports "much of the research on employment opportunity in the public sector" by Hochstedler et al. (1984) and Stein (1986) (p. 577). Warner et al. (1989) concluded that "formal affirmative action programs are essential for the eventual accomplishment of a representative bureaucracy in the United States" (p. 577).

The Equal Employment Opportunity Act (EEOA) of 1972 required local governmental agencies that receive federal funding to be placed under federal civil rights laws and administrative orders requiring Affirmative Action (Kim & Mengistu, 1994, p. 164). Though women have made progress, law enforcement agencies are still mostly white and maledominated (p. 164). Selden (1997) noted that Affirmative Action programs that resulted from the Civil Service Reform Act of 1978 also supported representative bureaucracy. According to Zhao and Lovrich (1998), several scholars have found that African American representation in police agencies in the U.S. has slowly increased since early efforts to encourage social equity (p. 268). One of the key elements of affirmative action programs is for agencies to specify "goals and timetables keyed to pools of qualified applicants" (p. 269). Zhao et al. (2001), in agreeing with Martin (1991), stated that one of the most significant factors that influence female representation in police agencies, especially in urban agencies, has been Affirmative Action policies and programs. Zhao et al. also found that the presence of a formal Affirmative Action program had a large impact of female representation in their study of 281 police agencies in the U.S.



Zhao et al. (2005) found in their study of officers of color in the 1990s "that the presence of neither a formal Affirmative Action program nor informal Affirmative Action program predicted the rate of employment" for African American or Latino officers (p. 384). They cited a significant decrease in the number of formal Affirmative Action programs between 1996 and 2000 and the number of informal programs remained about the same throughout the 1990s (p. 384). They offered two reasons for the decrease in formal Affirmative Action programs. The first reason may be that after about three decades, many police departments increased their numbers of officers of color (p. 384). The second reason for the decrease in formal Affirmative Action programs is that there have been challenges through union grievances and lawsuits to Affirmative Action Programs across the U.S. (p. 384).

Conservative presidents, such as Ronald Reagan, have significantly reduced federal enforcement of EEOA on the federal level (Warner et al., 1989, p. 565). Individual states have followed suit in striking down Affirmative Action efforts in college admissions and employment. Selden (1997) warns that abolishing Affirmation Action programs could decrease the numbers of women and people of color working in public agencies (p. 141). She further states that if agencies get rid of Affirmative Action programs and policies, recruitment, hiring, and promotion of women and people of color will not be a high priority and impede future progress (p. 142). Selden asserts that increasing women and people of color in management positions will positively affect the decisions regarding interests of women and people of color within their agencies.

Zhao et al. (2005) observed in their study of key variables in employment of African American and Latino police officers in the U.S. throughout the 1990s, "that the hiring of one ethnic group did not exclude the hiring of another group or white female officers" (p. 384). This



an important finding because some people have argued that hiring people of color and women is a "zero-sum game", which means hiring one group comes at the detriment of another group.

Police chiefs of color/Top management of color. Eisenberg, Kent, and Wall (1973) found in a study of 493 county and municipal police agencies that 84 percent of the participants identified the chief of police as "most frequently designated as having the major responsibilities for establishing personnel policies for sworn police personnel" (p. 13). Zhao and Lovrich (1998) found in their study that the second significant variable in predicting representation of African American officers in U.S. municipal police agencies is "the presence of an African American police chief" (p. 274). They also suggest this is because the police chief has "significant power in the hiring procedure" (p. 274). Zhao and Lovrich, however, add that any changes in personnel policies are due to several factors, including political, organizational and legal factors (p. 267).

Zhao et al. (2005) found in their longitudinal panel study in the 1990s that having an African American or Latino police chief was correlated with an increase in employment of police officers of color in the U.S. (pp. 377 & 384). Lewis (1989) found that Black police chiefs were one of the strongest predictors in increasing the percentage of Black officers in his study of large police departments in the United States. Gustafson (2013) found in his sample of 181 municipal police agencies in the U.S. that the presence of people of color in police leadership positions was one of the most significant factors in affecting diversity in line officers.

H. Greene (2003) asserts that African American police chiefs can influence policing more than African Americans at the officer rank because they have the power to set priorities within their agencies, as well as have an impact on policy (p. 212). This includes restricting the use of force, recruiting officers of color, and practicing community-oriented policing (p. 212). Many Black police chiefs, and Blacks in administrative and command positions, espouse community-



oriented policing "because it provides an opportunity to remove the historical barriers between police and the black community" (p. 212).

In addition to the police chief being important, the administration under the chief is important in diversifying police agencies. Van der Lippe, Graumans, and Sevenhuijsen (2004) studied police force gender policies in four European countries and found that "the role of top management is important when it comes to getting gender policies to be effective" (p. 398). They stated that in Sweden, the top managers are instrumental in "breaking down the barriers which women encounter" (p. 398). They also noted "in Sweden and the Netherlands, senior managements are very supportive and have stated publicly that they wish women to join the league of decisionmakers in the respective police force" (p. 398).

Agency size. Warner et al. (1989) also found that hiring women policer officers was not increased by an increase in the number of total officer positions. However, for officers of color in police agencies, growth in the police department is positively associated with employment opportunities for officers of color (p. 574). Morabito and Shelley (2015), in their study of 1,478 police jurisdictions of differing size and composition using data from the 2003 LEMAS surveys, the 2000 U.S. Census, and the 2003 UCRs, explored the factors that relate to using representative theory as its framework. They also found that for females, officers of color, and African Americans, agency size was positively associated with increased representation of all three groups (pp. 341, 344 & 345).

Presence and type of union or collective bargaining unit. Walker (1985) has noted that police unions opposed ways to improve police-community relations and have represented White officers who challenged Affirmative Action programs and plans in courts. Warner et al. (1989) have also cited several large police unions and associations that have filed grievances and



lawsuits against Affirmative Action, including the Fraternal Order of Police and the International Association of Police Chiefs. However, Riccucci (1986) found that of the 74 cities in her study, the presence of a union had a limited effect on female employment within municipal police departments. Walker (1985) emphasizes that police unions are dynamic and change as their social and economic environment, and the demographic composition of an agency changes. Kim and Mengistu (1994) found in their study that collective bargaining did not significantly influence female and Asian police office employment and that police unions did not promote hiring Black police officers, but did Hispanic officers (p. 171).

Walker (1985) has asserted that police unions have represented the interests of White officers. Walker (1985) and Warner et al. (1989) assert that unions have openly opposed Affirmative Action programs. Schuck (2014) discussed the mixed results of empirical research regarding union influence on passive representation. Warner et al. (1989) did not find a correlation between collective bargaining and female representation in their study of 281 metropolitan police departments. Using a more diverse and larger sample of agencies, Sass and Troyer (1999) found less female representation with unionization in the early 1980s, but not for the late 1980s and early 1990s. Schuck (2014) hypothesized in her study of over 400 agencies that "law enforcement organizations with a collective bargaining unit will have lower levels of female representation than those that do not have a collective bargaining unit" (p. 59). She found that "collective bargaining had a direct negative impact on gender diversity" (p. 68). Kellough (1990) examined the employment of people of color and women in federal agencies and found a strong positive relationship, rather than the negative relationship he hypothesized between unions and the employment of women, Blacks, and Hispanics at the federal level (p. 562). As an explanation for this finding, Kellough suggested that public employee unions are not a hindrance



to employment of females and people of color at the federal level as found at the municipal level. He further elaborated that federal unions "have a much more limited scope of collective bargaining and must represent employees on issues other than wages and benefits" and are less oppositional toward Affirmative Action and equal employment opportunity than municipal employee unions (p. 562).

Walker (2008) reiterated that "police unions are widely recognized as having a major influence on American policing, with respect to day-to-day management, disciplinary procedures, and police-community relations, especially with minority communities and the African American community specifically" (p. 95). Walker highlighted that many police experts perceive that police unions had, and continue to have, an enormous negative influence on police-community relations (p. 105). Morabito and Shelley (2015) found that the presence of collective bargaining was significantly and positively associated with increased female representation, but not for officers of color (pp. 341, 345 & 345).

Autonomous professional personnel bureaus. Stein (1986) stated "the growth of autonomous, professionalized urban bureaus was made possible by the weakening or elimination of a politically based local executive during the Progressive reform period" (p. 696). She describes the civil service with its aim of substituting political hiring with selection by merit (p. 696). However, she also noted that the restrictive qualifications that define the merit principle "prevent many minority applicants from obtaining positions with local government" (p. 697). Stein found in her study that professional autonomous bureaus were resistant to direction from the outside and the presence of a civil service commission had a significant negative relationship on employment of people of color in large city work forces (p. 702).



African American personnel directors and administrators. Meier and Bohte (2001) argued that representative bureaucracy theory asserts that African American personnel directors and administrators may exercise their discretion in making decisions based on their values, attitudes, and beliefs that are formed from their racial socialization. Goode and Baldwin (2005) found in their study of new predictors of African American representation in city government, that the presence of African American personnel directors and administrators was correlated with African American representation in municipal government (pp. 29 & 40). They concluded that those involved in the internal environment of personnel are "generally more predictive of African American representation in city government than variables in the external environments of municipalities" (p. 29).

Active representative bureaucracy. Mosher's (1968) defined the term *active* representation as "wherein an individual (or administrator) is expected to press for the interests and desires of those whom he is presumed to represent, whether they be the whole people or some segment of the people" (pp. 11-12). Selden et al. (1998) defined *active representation* as "how minority employment or 'passive' representation is translated into the 'active' representation of minority interests in the decisions of public agencies" (p. 717).

It was assumed that public agencies can transform passive representation into active representation to achieve representative outcomes (Sowa & Selden, 2003, p. 701). Selden (1997) and Kennedy (2014) stated that a bureaucracy that is diverse will promote public policy that is more responsive and that will represent all interests in the formulation and implementation of policy.

Kennedy (2014) identified three components necessary to have active representative bureaucracy. The first is that bureaucrats must have discretion (p. 406). Second, policy



decisions must have direct relevance to passively represented groups (p. 406). Third, bureaucrats must be linked to a specific policy output (p. 406). The system in place with Human Resources, for example, needs to be compatible with recruiting qualified candidates of diverse backgrounds.

The Relationship between Passive Representative Bureaucracy and Active Representative Bureaucracy

Research has shown a relationship between passive and active representation, but there is a debate as to the nature of the relationship (Meier & Smith, 1994). The debate includes what kinds of decisions to include and "how does the bureaucratic work environment facilitate or constrain the link between passive representation and active representation?" (Hindera & Young, 1998, p. 656). Hindera and Young extended this theory to test whether "the proportion of a particular group in a public bureaucracy (passive representation) is related to enhanced agency outputs of group members in the general population (active representation)" (p. 656). The dependent variable was measured as the proportion of charges filed on behalf of Black complainants (p. 664). The independent variable was the percentages of investigative Black and White staff making the decisions. They found that Black passive representation is related to active representation in that Black staff filed higher numbers of EEOC charges on behalf of Black complainants (p. 665). Hindera and Young found that for active representation to take place, there must be "some minimal level of political support from within the organization" (p. 662).

Passive representation leads to increases in perceptions of the legitimacy of police actions. The need for research to identify the process on how passive representation may lead to active representation, including how the employment of women and people of color in government, political offices, and police leadership positions, increases passive and active



representation, is well documented in the literature. The findings of Hindera and Young (1998) also suggest that the relationship between passive and active representation "is fundamentally different for each type of bureaucratic environment" (p. 655).

Meier and Nicholson-Crotty (2006) argue that active representation "focuses almost exclusively on how [active] representation affects policy making and implementation" and "assumes that shared values arising from passive representation will produce actions that are more beneficial to the clients in question" (p. 851). An example of how passive representation can be translated into active representation is that females and communities of color who see police officers that have similar demographic characteristics to them, may be more likely to contact police, report a crime, be a witness, and cooperate with police. Another way in which passive representation can be turned into active representation is that female officers and officers of color can influence the attitudes and behaviors of other police officers (p. 853). This will benefit the workplace and interactions between the public and the police. This demonstrates that indirect contact can have benefits (p. 853).

Recent research (Grissom, J. Nicholson-Crotty, & S. Nicholson-Crotty, 2009) on representative bureaucracy theory has paid attention to the circumstances and situations that moderate the relationship between the race or gender of public bureaucrats and the active representation of group interests. Meier and Bohte (2001) have demonstrated that structural characteristics of an organization, the political and social context outside of the organization, and the views and opinions of individual bureaucrats, can augment or inhibit bureaucratic representation. Grissom et al. (2009) found in their study using a representative sample of more than 3,000 public schools, that in the South, "region of residence is an important moderator of active representation because it helps to determine the salience of race as an issue and the degree



of identification with racial group interest" (p. 911). Their results implied that Black teachers in the South generate larger benefits for Black students compared with other regions in the United States (p. 911).

Mosher (1968) describes how passive representation can have symbolic affects:

While passive representation is no guarantor of democratic decision-making, it carries some independent and symbolic values that are significant for a democratic society. A broadly representative public service, especially at the level of leadership, suggest an *open service* in which access is available to most people, whatever their station in life, and in which there is *equality of opportunity*. ... The importance of passive representativeness often resides less in the behaviors of public employees than in the fact that the employees who are there are there at all. Negatively, its significance rests on the absence, or conspicuous underrepresentation, of certain categories of people, suggesting or reflecting barriers to their entry or advancement. (pp. 13-14)

Lim (2006) further expounded on how bureaucrats of color can create substantive benefits through direct sources by their own behavior, or through indirect sources, which are the changes they encourage in the behavior of others (p. 195). Saltzstein (1979) and Selden (1997) have examined the direct construction of substantive benefits for the social group of bureaucrats of color. Saltzstein (1979) argues that "the theory of representative bureaucracy requires that bureaucratic policy making be responsive to the interests of represented groups which necessarily implies a link between passive representation and values, values and behavior, and behavior and policy" (p. 470). Lim (2006) interprets this as meaning bureaucrats who share a demographic background, experience similar socialization, attitudes, values, and beliefs with other people who share their social background, will act in ways that enhances substantive



benefits for their social group (p. 195). Walker (1985) stated that there are three main areas of research for the potential impact (active representation) of people of color and females (passive representation) in law enforcement agencies: police performance, public attitudes, and the police subculture (p. 563).

Police Performance Outcomes of a Representative Police Force

The first area of research on the active representation of women police officers and police officers of color identified by Walker (1985) is their influence on police performance. Meier et al.'s (1999) research found that teachers of color benefit students of color and do not disadvantage White students. They concluded "that representative bureaucracies are more effective than their nonrepresentative counterparts" (p. 1025). They also stated "that discriminatory personnel policies result in less able employees, which, in turn, detrimentally affects agency performance" (p. 1037). Although this research is outside policing, these results are applicable to law enforcement because a representative agency will improve the delivery of police services to all segments of the community.

In one of the first studies following the integration of women into patrol duties, Bloch and Anderson (1974) found that there were not any significant differences between the behavior of female and male officers. Walker (1985) raises valid concerns regarding studies of police performance for officers of color in stating that many studies did not include officer race as an independent variable and many behaviors had not been adequately studied, except for use of deadly force. This is still applicable today, what constitutes good police performance is of much debate, particularly regarding what are good performance indicators to evaluate officers who practice community-oriented policing.



Meier and Nicholson-Crotty (2006) found in their study of 60 urban areas over an eight year period that the percentage of women police officers in a department was "positively associated" with the number of sexual assault reports and the number of arrests for sexual assaults (pp. 850, 856 & 857). They also have highlighted the importance from their study "that passive representation affects the actions of other bureaucrats and clients" (p. 858).

The first studies to apply representative bureaucracy theory to traffic stops were at the aggregate level (Mason, 2013; Wilkins & Williams, 2008, 2009). Wilkins and Williams (2008), in examining the relationship between passive and active representation, found in their study using 2000 data from the San Diego Police Department, that as the percentage of African American police officers increased, so did racial profiling in traffic stops (p. 654). They concluded that organizational socialization can impede the relationship between passive and active representation (p. 654). Using the same data as above, Wilkins and Williams (2009) also found that the presence of Latino officers increased racial disparity in traffic stops (p. 775). Gilliard-Matthews, Kowalski, and Lundman (2008) using data from citizens in the United States to examine police-citizen contacts, found that minority drivers were significantly more likely to be given a ticket by white officers than by African American officers.

Sun and Payne (2004) in an analysis of 3,130 separate encounters between the police and suspects in Indianapolis and St. Petersburg, found that in predominantly African American neighborhoods, African American officers were more apt to show supportive activities than White officers. However, they also found that African American officers were more coercive in their reaction to interpersonal conflicts. Sun and Payne detected that "decisions to perform traditional law enforcement functions (e.g., using force, arresting, and maintaining order) seem to be congruent with the "blue cop" identity whereas decisions to perform supportive activities



are congruent with identities that show concern about their community" (pp. 535-536). They use the metaphor of "blue cop" for Black police who were influenced by the police traditional socialization process.

Legewie and Fagan (2016), using a dataset from 2013 to 2015, hypothesized that a police force that is representative of the population it serves will reduce group threat and conflict, thereby decreasing the number of officer-involved killings of African Americans. Their findings identified an "interaction between group threat and proportional representation of minority groups in police departments" (p. 1). They explained that "group threat is not simply driven by population shares and in-migration of out-group members as suggested by previous studies" (p. 31). Instead, Legewie and Fagan found that "the number and rate of officer involved killings of African-Americans is substantially higher in cities with a high level of ethno-racial polarization where two equal-sized groups face each [other]" (p. 31). They explained that diversity within a police agency might not by itself decrease the number of officer-involved shootings, however, minority representation "mitigates different dimensions of threat" (p. 32). Their study makes a significant contribution to research because it is an example of how passive representation, a diverse police agency, can impact active representation, a benefit to the African American community. In addition, this study also identifies mitigating factors relating to group threat, thus, decreasing the strained relationship between the police and African Americans (pp. 32-33). This is an example on how representation affects police performance. These findings may be extended to include other communities of color.

Street-level bureaucrats and discretion. Many scholars, including Dempsey and Forst (2013), have emphasized that "the use of discretion is one of the major challenges facing U.S. police today" (p. 73). They also noted that the most critical decisions in the criminal justice



system "take place on the streets", thus "the police officer, generally the first decision maker in the U.S. criminal justice system, is often the most important" (p. 73). The amount of discretion police officers use in the performance of their daily tasks is much greater at the bottom of the police organizational hierarchy (p. 39).

Lipsky (1980) described street-level bureaucrats, which include the police, as workers in agencies who "interact with and have wide discretion over the dispensation of benefits or the allocation of public sanctions" (p. xi) and as "public service workers who interact directly with citizens in the course of their jobs, and who have substantial discretion in the execution of their work" (p. 3). Meier and Nicholson-Crotty (2006) also described police officers as an example of "street-level bureaucrats" because police officers interact with the public and exercise much discretion in the performance of their duties. In applying Lipsky's (1980) work to their study on the relationship between women police officers and sexual assault reports and arrests, Meier & Nicholson-Crotty (2006) stated "because street-level bureaucrats exercise discretion, their attitudes, values, and predispositions are important in understanding policy implementation" (p. 851). This emphasizes the relationship between a bureaucrat's personal characteristics and policy issues that affect individuals who share those characteristics, which is a form of representative bureaucracy and assumes that "shared values from passive representation will produce actions that are more beneficial to the clients in question" (p. 851). This highlights the importance of representation of women and people of color in law enforcement because law enforcement has been traditionally a White male dominated profession. Representative bureaucracy theory suggests that the representation of female and officers of color (passive representation) impacts police agencies and the communities they serve (active representation).



Administrative discretion over policy outcomes in the context of representative bureaucracy. One of the key concepts in studies on the theory of representative bureaucracy is administrative discretion (Sowa & Selden, 2003; Dolan, 2000). It is important to understand the role and impact of administrator discretion on policy making and implementation in recruitment, hiring, and retaining female and officers of color. Also important is the impact of discretion on officers at the street-level who make decisions and exercise discretion on a daily basis that affect females and communities of color. Selden et al. (1998) emphasized that for administrators to influence policy outcomes, public administrators must have "a significant amount of discretion in the decision-making process" that impacts representative groups (p. 728). Among decisions that impact the interests of representative groups are decisions regarding the hiring of a representative police force and decisions that affect the delivery of police services to women and people of color within the community.

Sowa and Selden (2003) found in their study of discretion in making loan-eligibility decisions for the Farmer's Home Administration that "administrators who perceive themselves as having more administrative discretion produce outcomes that are more responsive to the interests of minorities than those of their colleagues" and had a significant impact on the rural housing loans granted to minorities (pp. 701 & 706). They sought to expand the theory of representative bureaucracy to "an unexplored aspect of the theory: the impact of administrative discretion on active representation" (p. 701).

One of the important ways of how passive representation can be transformed into active representation are the benefits from a representative workforce. Active representation acknowledges discretion on the part of administrators to produce outcomes that are more representative and equitable to the people being represented (Sowa & Selden, 2003, p. 701).



Also important are particular attributes of administrators, such as race, ethnicity, and gender, that affect early socialization experiences, that then influence the values and attitudes of administrators (p. 701). For passive representation to result in active representation, it is important that values and attitudes influence actions and that administrators use "their discretion to foster improved equity for those who have been underrepresented in the implementation of public programs" (p. 701).

Dolan (2000) also asserted that public administrators affect public policy through their administrative discretion (p. 513). She studied the influence of female members in the Senior Executive Service (SES), who contributed to making decisions on national policy while working with the president's political appointees. Dolan hypothesized, using representative bureaucracy theory, that because of different life experiences for women, women in the SES would make dissimilar decisions compared with their male equals (p. 513). She suggested that a woman's life experiences shape her attitudes and values, which then influences her decisions. She found that women in the federal SES act as active representatives. She also concluded that the relationship between passive and active representation may vary by organizational setting and is more likely to happen in agencies that advocate for women's issues.

Meier and Bohte (2001) using a sample of six hundred school districts in Texas to analyze the role of discretion in the relationship between passive and active representation asserted that active representation is increased in organizations that provide more discretion to their employees. Using span of control as their measure of organizational structure, they found that the performance of students of color increased in organizational structures that generated more discretion by teachers of color. They hypothesized that the more people a person supervises, the more discretion they allow to the subordinates they supervise because they do not



have enough time to closely supervise everyone (p. 457). This applies to the amount of discretion given to police officers, especially at the street level.

Role acceptance of administrators. Sowa and Selden (2003) found in their study on discretion in making loan-eligibility decisions in the Farmer's Home Administration that "administrators who perceive themselves as possessing significant discretion and who assume the role of minority representative in their agencies are more likely to enact policy outcomes that favor minority interests" (p. 700). Sowa and Selden defined work roles as a "particular set of behaviors expected" by a particular job or position (p. 704) and described traditional role acceptance as including the concepts of merit and neutral competence to achieve efficiency and economy, and the focus on complying with standard operating procedures rather than outcomes favoring minority interests (pp. 704-705). Selden (1997) introduced the concept of minority role representative in representative bureaucracy literature as bureaucrats willing to view themselves as advocates for, or representative of, minority interests and Sowa and Selden (2003) described minority representative role acceptance as administrators who perceive they have a responsibility to serve minority communities (p. 704). Sowa and Selden found that supervisors who perceived their role as a minority representative granted more loans to minorities, thus they are more likely to make decisions that advance minority interests (p. 706). It is suggested that those who assume a minority representative role will emphasize improving access to groups that have been traditionally underrepresented in the policy process (p. 705). They concluded that discretion is one mechanism that strengthens the relationship between passive and active representation. It is important to note that Whites can also advocate for the interests of people of color.



Selden et al. (1998) conducted research regarding whether the Farmer's Home

Administrators granted loans to minority applicants differently than to White applicants. They were interested in how passive representation translates into active representation by proposing a "crucial intermediary concept in this process" ... "of a minority representative role perception by public administrators" (p. 717). The most significant finding in their study was "that minority status, as hypothesized, exerts a strong influence on adherence to the minority representative role", which is "crucial to the theory of representative bureaucracy", because "race and ethnicity appear to influence the attitudes and role perceptions of government employees regarding minority representation" (p. 732). Kennedy (2014) emphasizes that few scholars have analyzed how bureaucrats view their representative role (p. 214). Kennedy also suggested that qualitative analysis may offer a more insightful and contextual understanding of representative bureaucracy (p. 414).

Public and Police Attitudes

The second area of research identified by Walker (1985) is the impact of women officers and officers of color on public attitudes towards the police. Walker stressed from surveys of citizens "that black citizens are less satisfied with urban public services than are white citizens" (p. 564). There has not been much research on whether progress in people of color or female employment in police agencies improves the image of a police department.

Theobald and Haider-Markel (2008) found in a study using individual level data from a national police-citizen contact survey that symbolic representation occurred and that the perceptions of citizens regarding the legitimacy of police actions were shaped by the interaction of citizen race and officer race. They also found that Black citizens were more likely to perceive the actions of Black officers as legitimate and that Whites were more likely to perceive the



actions of White officers as legitimate (p. 409). They defined the concept of symbolic representation as being different than active representation. While active representation is where the representative acts on behalf of represented groups, symbolic representation "works cognitively on the audience of those who belong to a group that is being represented" (p. 410). In symbolic representation, "attitudes and outcomes can change without any purposeful actions taken by the representatives other than holding a government office or position" (p. 410).

Research has found that race shapes citizens perceptions of fairness and whether police actions are legitimate (Theobald & Haider-Markel, 2008). Sigelman, Welch, Bledsoe, and Combs (1997) found that 90% of African Americans perceived that the police are not fair to African Americans and that 60% of Whites perceive that the police are not fair to African Americans. Hurwitz and Peffley (2005) found that when police actions involve African Americans, African Americans were more likely to perceive police actions as unfair and were more suspicious of those actions involving African American citizens. They also found that Whites were more likely to support the police when there were concerns of discrimination regarding interactions between the police and Black citizens. Theobald and Haider-Markel (2008) emphasize that if increased Black representation affects Black's attitudes about police actions, then this has crucial implications for policy making and the community (p. 414).

Mansbridge (1999) added that if descriptive or passive representation increases trust, increased perceptions of inclusion may produce citizens perceptions that the actions of the state and their institutions were more legitimate.

Cochran and Warren (2012) found in their research analyzing data from the 2005 Bureau of Justice Statistic's Police Public Contact Survey (PPCS), which is a supplement to the National Crime Victimization Survey (NCVS), that officer race was relevant in influencing citizen's



assessments of police encounters in traffic stops, especially for Black citizens. They found that Black males and females were more apt to negatively evaluate police behavior when a White officer initiated the traffic stop (p. 219). However, when citizens were stopped by officers of color, they found no citizen-race or ethnicity effects and the citizens were more likely to perceive the stop as legitimate (p. 219). Cochran and Warren concluded that this important finding supports increasing the number of officers of color to improve relationships between citizens and police officers and decrease the perceptions of racial profiling. Other research regarding citizen attitudes towards the police show that people living in impoverished and severely disadvantaged communities and neighborhoods are not satisfied nor have a positive perception of the police, regardless of the racial demographics of the police agency (Brunson & Gau, 2015; Weitzer, Tuch, & Skogan, 2008).

Attitude congruence. Hindera (1993) describes attitude congruence as group attitudes leading to predictable behavior (p. 100). Bradbury and Kellough (2008) surveyed public administrators who have authority to make decisions and citizens in Athens, Georgia to examine their attitudes regarding the responsibilities of local government administrators to promote the concerns and issues of the African American community (p. 697). They predicted that public administrators of color "will have similar attitudes to minority citizens on issues of critical import and relevance to those citizens, and those attitudes, in turn, will influence policy decisions" (p. 697). They found that African American administrators are more likely than White administrators to adopt an African American representative role. Their most significant finding was that African American administrators who had similar attitudes to African American citizens was "a significant predictor of the adoption of an African-American representative administrator role, overwhelming the influence of other variables including race" (p. 697). They



also found that African American citizens and administrators are more likely than White citizens and administrators to support actions that benefit the interests of the African American community. Selden (1997) and Bradbury and Kellough (2008) point out that even though bureaucrats of color were a significant predictor of the minority representative role, White bureaucrats can also take the role of a minority representative.

Police attitudes towards the community. In addition to attitudes citizens have of the police, there has been research on attitudes police have towards the community. Lasley, Larson, Kelso, and Brown (2011) examined the longitudinal changes in police officer attitudes towards the community using the framework of representative bureaucracy theory. Their sample was made up of 405 male Caucasian, African American, and Latino/Hispanic patrol officers who started working for the Los Angeles Police Department between 1985 and 1991 under Affirmative Action hiring (Lasley et al.). They surveyed the same officers at two points in time, in 1992 and in 2007 (Lasley et al.). The results over the 15 years of the study found that "African-American and Latino officers significantly increased their desire to engage in active representation or 'partnerships' with the community" (p. 474). They also found a similar effect on White officers in 2007, who in 1992, indicated they had lower desires for community interaction than officers of color in 1992 (Lasley et al.). They suggested that White officers "may have changed their attitudes towards police-community interaction as a result of prolonged exposure to the influences of ethnic diversity with the LAPD organization" (p. 482).

The Police Subculture

The third area of research identified by Walker (1985) is the impact women police officers and police officers of color have on the police subculture. The concept of a police subculture began in the 1960s with renowned studies that described the police subculture as



causing many of the problems between the police and communities of color (Crank, 2004; Niederhofer, 1969; Skolnick, 1994; Westley, 1970). Walker (1985) describes "a police subculture in which minorities and women are seriously underrepresented and in which the attitudes and behavior of police officers reflect a dominant white male group" (p. 556). Traditionally, police officers were White males from the working class and White lower middle class, who were conservative, authoritarian and racially prejudiced (p. 563).

The subculture underscored solidarity of the group, antagonism toward the public, a "we versus them" attitude, secrecy, and the "code of silence" towards misconduct and brutality (Walker, 1985, p. 103) and was "characterized by clannishness, secrecy, and isolation from those not in the group" (Dempsey & Forst, 2013, p. 82). Dempsey and Forst (2013) stated that studies have shown that police officers protect one another from outsiders. They describe the "blue wall of silence" as "a figurative protective barrier erected by the police in which officers protect one another from outsiders, often even refusing to aid police superiors or other law enforcement officials in investigating wrongdoing of other officers" (p. 83). Many scholars, including Walker, have noted that the concept of police subculture needs to be reevaluated because it has morphed from a monolithic concept into a complex, multi-dimensional concept and is more refined than the original concept (Walker, 2008, p. 104). This includes breaking down the secrecy and blue wall of silence because of the current climate of increased scrutiny of police conduct. This calls for officers to not only come forward, but to also intervene when other officers use excessive force or engage in misconduct or brutality.

Diversifying police agencies can change some of the aspects of the police culture. One of the categories of the effects of the changing demographics within police agencies is organizational effects, which is how diversity in a police agency affects the internal dynamics of



that agency, including the subculture (Sklansky, 2006, p. 1212). Sklansky asserts "that the demographic transformation of American law enforcement has done much to break down the police subculture, by weakening both the occupational solidarity and the social insularity of the police" (p. 1212). Lasley et al. (2011) identified one example of how racial and ethnic diversity can impact police culture. They concluded in their study on the longitudinal changes in police officer attitudes towards the community that the "attitude change among all LAPD officers in 2007 – and particularly among Caucasian officers – may have been due, in part, to LAPD's traditional policing culture having been transformed through ethnic diversification into a representative bureaucracy" (p. 490).

Leadership at the Local Level to Initiate Change

Shusta et al. (2011) identified the chief law enforcement officer as a key component in whether their agency is effective and successful in meeting the challenge of policing in a pluralistic society and building good relationships with the community (pp. 72 & 73). Shusta et al. also emphasized the following components of effective leadership in a pluralistic workforce and society:

- Demonstrating commitment to policing a diverse society. This includes developing
 policies from all levels of the agency and community that show a commitment to
 diversity in the workplace and in providing police services.
- Developing strategic, implementation, and transition management plans.
- Managing organizational change.
- Developing police-community partnerships.
- Providing new leadership models. This includes being innovative by transforming workplaces from the status quo to appropriate environments, renewing organization



and becoming role models by transmitting intellectual excitement and vision about their work, and helping personnel to manage change by restructuring their mindsets and values (pp. 73-76).

How Social Equity Theory Ties into Representative Bureaucracy Theory

As stated by Selden et al. (1998), representative bureaucracy "operates to promote equity" where "efforts are made to ensure that minority interests are not ignored in circumstances where those interests need to be heard and might otherwise be disregarded" (p. 739). A work force that is reflective of the community conveys economic benefits and symbolic significance to communities of color. Warner et al. (1989) stated the desired impact of the Title VII Civil Rights Act of 1972 that extends the CRA of 1964, which prohibits discrimination in personnel practices by state and local governments, is moving towards social equity and the start of representative bureaucracy (p. 562). They further state that social equity is one of many competing values in hiring decisions in the public sector (pp. 562-563). Other values to consider are "merit, timely appointment, or cost-effectiveness" (p. 563). Which value is emphasized heavily depends on the current social, political and economic climates that influence hiring decisions (p. 563). For example, Warner et al. note that during the Reagan presidency Affirmative Action programs were severely reduced.

Passive representative bureaucracy can send a message of "commitment to equal opportunity and access to power and can promote the legitimacy of public bureaucracies" (Meier & Nicholson-Crotty, 2006, p. 851). Active representation on the part of women police officers and police officers of color can improve equal opportunity and access to police services and protection to women and people of color.



Police Officer Retention and Separation

Reaves (2012) summarized statistics on the retention of local law enforcement officers in the 2008 Bureau of Justice Statistics (BJS) report of the CSLLEA (Census of State and Local Law Enforcement Agencies), which includes types and reasons of separations. Overall, about 7.4% of all state and local officers separated from agencies in 2008 (Reaves, 2012; Stevens, 2018). "On average, large agencies have a lower turnover rate (5 percent) than do small agencies (7 percent)" (p. 234). Stevens (2018) added that "this attrition rate appears to have remained stable since the 1960s" (p. 234).

Wareham, Smith, and Lambert (2015) using both the 2003 LEMAS survey and the 2008 CSLLEA data for their study, found that "the total turnover rate was 10.8% for both 2003 and 2008" (p. 345). Wareham et al. also found that attrition rates for police officers were higher in smaller agencies, southern regions, and rural areas (p. 345). Their results showed higher separation rates, 10.8%, than Reave's BJS report of 7.4%. Wareham et al. explained that this was due to Reave's report not presenting "separation rates across geographic regions, states, urbanity, or types of agencies (i.e., municipal vs. sheriff)" (p. 348). Wareham et al. used weighted samples "to adjust the sample to the population of law enforcement agencies with fewer than 100 officers based on the nature of an agency and the population size of its jurisdiction" (p. 349).

Wareham et al. (2015) calculated their turnover rates by dividing the total number of full-time sworn officers for each turnover indicator by the number of full-time sworn officers and multiplied that number by 100 to get percentages of turnover (p. 350). The rates are aggregate rates of turnover and do not include reasons for officers who resigned, including officers who left for another law enforcement position (p. 365).



Wareham et al. (2015) used Price's (1977) definition of turnover in their study: a "permanent ending of employment" which "does not include absenteeism, long-term leave, layoffs, transfers, promotions, or demotions" (p. 350). Wareham et al. used the three broad categories of "why employees leave an organization": "voluntary, involuntary, and retirements". Wareham et al. defined voluntary turnover as when someone quits, but not including retirement. Involuntary turnover was defined as when "the turnover is not a willing decision by the employee" and most of the time, it "is initiated by the employing organization" with examples including termination, death, and disability (p. 350). Retirement was defined as when the employee voluntarily retires from the agency (p. 350).

Wareham et al.'s study (2015) asked participating police agencies to report the number of police officers lost within the following six sub-categories: "resignations, dismissals, medical/disability retirements, non-medical retirements, probationary rejections, and other separations (e.g., death)" (p. 350). They used the following four measures of police turnover for their study: resignations (voluntary turnover), retirements (non-medical retirements), voluntary separations (resignations and non-medical retirements), and total turnover (all categories of separation, including dismissal, probationary rejections, medical/disability retirements, and deaths) (pp. 350-351). Their results showed that the average turnover rates for the 2003 LEMAS sample were 7.79% for resignations, .92% for retirements, which combined is 8.71% for voluntary separations, and 10.82% for total turnover (p. 352). This means that 2.11% of the officers were terminated by involuntary separation, which included dismissals, probation rejections, medical disabilities, medical retirements, and death. For the 2008 CSLLEA sample, the results showed 7.31% for resignations, .90% for retirements, which combined is 8.21% for voluntary separations, and 10.76% for total turnover (p. 352). This means that 2.55% of the



officers were terminated by involuntary separation, which included dismissals, probation rejections, medical disabilities, medical retirements, and death. For the Midwest region, they found "resignations and voluntary separations were statistically lower in 2008 than 2003, but retirements were statistically higher" (p. 353). The findings from Wareham et al.'s study found that geographic location, type of agency, and size of agency influenced different kinds of turnover (p. 365). Urban agencies reported lower levels of resignations, voluntary separations, and total turnover than rural and suburban agencies" (p. 365). Smaller agencies had "higher rates of resignations, voluntary separation, and total turnover, as well as lower rates of retirements" (p. 365).

Table 2 summarizes the data from Wareham et al.'s (2015) study on the mean turnover rate (the number of officers who separated, divided by the total number of officers, and multiplied by 100) for the Midwest region and U.S. Division where the case agency is located, urban status (includes rural, suburban, and metro agencies), type of agency (includes municipal, county, and state agencies), and size of agency (includes medium agencies (50-99 officers) and large agencies (100-200 officers)) for the 2003 LEMAS survey data and the 2008 CSLLEA data. Wareham et al.'s definition of medium and large agencies is included. As the data in the table presents, there is a wide range of mean turnover rates which are dependent on geographic location, urban status, type of agency, and size of agency.



Table 2. Mean Turnover Rates for the 2003 LEMAS Survey and the 2008 CSLLEA Data for Geographic Location, Urban Status, Type of Agency, and Size of Agency.

Type of Turnover	Midwest	Rural,	Municipal,	Medium (50-99
	Region and	Suburban, and	County, and	Officers) and
	U.S. Division	Metro	State	Large (100-299
	of Case	Agencies	Agencies	Officers)
	Agency			Agencies
2003 LEMAS				
Resignations	6.1 to 9.4	4.8 to 10.6	2.2 to 8.5	2.9 to 3.1
Retirements	.3 to .9	.7 to 1.3	.9 to 2.8	1.3 to 1.8
Involuntary Separation	1.1 to 1.4	1.4 to 2.7	.9 to 2.2	1.3 to 1.4
Total Turnover	7.5 to 11.5	7.6 to 14.1	5.9 to 11.6	5.5 to 6.3
2008 CSLLEA				
Resignations	3.0 to 7.8	4.0 to 9.8	1.8 to 7.5	3.2 to 3.5
Retirements	.5 to 1.0	.8 to 1.2	.8 to 2.1	1.2 to 1.6
Involuntary Separation	.5 to 2.1	1.8 to 3.6	.9 to 2.7	1.4 to 1.5
Total Turnover	4.1 to 10.6	6.9 to 14.2	4.9 to 11.1	6.2 to 6.4

This data was compiled from data in Wareham et al.'s study (2015).

Using the same 2008 CSLLEA data as Wareham et al. (2015), Reaves (2012), in the BJS report of the 2008 CSLLEA data, calculated the percentage of those who separated by type of separation. The percentages for each category represent the percentage of the total number of separations, rather than the percentage of the number of officers used in Wareham et al.'s study. Reaves calculations of officer separations in 2008 included 54% for resignations, 23% for non-medical retirements, 10% for dismissals, 5% for probationary rejections, and 5% for medical or disability retirements, and 4% for other, which includes disability and death (Reaves, 2012; Stevens, 2018, p. 6). Table 3 shows officer separation from the 2008 CSLLEA, by type of separation and two of the categories for size of agencies.

Table 3. Type of Separation for the 2008 CSLLEA Data by Size of Agencies.

Type of Separation	Agencies of All Sizes	100-499 Officers	25-99 Officers
Resignation	54%	51%	58%
Non-medical retirement	23%	26%	18%
Dismissal	10%	9%	11%



Table 3. cont.

Type of Separation	Agencies of All Sizes	100-499 Officers	25-99 Officers
Probationary rejection	5%	6%	5%
Medical/disability retirement	5%	5%	5%
Other – Disability and death	4%	4%	3%

This data was compiled from data in Reaves (2012) report (p. 7). Total percentages for first two columns total 101% due to rounding.

Police Officers who Leave for another Law Enforcement Agency

Stevens (2018) examined the separation data from Reaves' (2012) report that used the 2008 CSLLEA data. Stevens found that for officers who left the agency for another law enforcement position, about 45% left from small agencies and about 24% left from large agencies (p. 234).

Females who Leave Policing

Regarding females leaving policing, Fyfe and Kane (2006) examined attrition rates for female officers in the U.S. and found that female officers leave policing at higher rates than do male officers. They found that female officers were more likely to resign voluntarily than male officers (4.3 percent versus 3.0 percent, respectively) and that female officers were terminated involuntarily more often than their male counterparts (1.2 percent versus 0.6 percent, respectively).

Martin (1994) concluded that the higher rate of women leaving law enforcement was due to women experiencing a "more hostile work environment" than do males (p. 398). Stevens (2018) added that "female officers who are single parents are also confronted with different problems than males in the workplace and are treated differently than male officers by their commanders as well" (p. 235). Stevens also reports that "minorities resign at higher rates than do white male officers" (p. 235).



Length of Service and Separation

Regarding the relationship between length of service and attrition, Koper, Maguire, and Moore (2002) found that two-thirds of the officers who leave both large and small agencies have 5 or less years of service (p. 47). Stevens (2018) also added that "officers with intermediate lengths of service (i.e., 6 to 14 years of service) are the least likely to leave both large and small agencies" (pp. 235-236).

Summary

This chapter discussed the historical background of recruiting, hiring, and retaining diversity within law enforcement and relevant issues relating to this study. Also discussed were the status of female police officers and officers of color in the U.S. and the benefits of a diverse agency. This chapter also discussed the theoretical and conceptual framework that applies to this study and the factors and variables that are examined. Discussed last was police officer separation.

The next chapter discusses the methodology for this study. This includes the purpose of the study, the research questions, and theoretical and conceptual framework. The next chapter discusses the research design, philosophical orientation, and methodology, why the case study is an appropriate approach, and triangulation of data sources and methods. The chapter then discusses the agency (case) selected for the case study; ethical issues; data sources, collection, and analysis; and reliability, validity, and authenticity.



CHAPTER III

METHODS

The purpose of this study is to describe how one Midwest law enforcement agency recruits, hires, and retains full-time officers that are representative of the communities it serves. This descriptive, single case study uses triangulation of multiple methods, including quantitative and qualitative methods, and multiple data sources through a logical theoretical and conceptual framework. This study also identifies applicable variables in describing how the agency recruits, hires, and retains diversity within the agency.

Theoretical and Conceptual Framework

The main theory for this study is representative bureaucracy as described in the Review of Literature chapter. Also, social equity theory is applicable because one of the elements that representative bureaucracy promotes is social equity, thus this theory is a part of representative bureaucracy theory. Theory is used beforehand to guide data collection of relevant concepts and variables, to construct questions in the interview guide, and interpret the data. Theory is also used to explain what accounts for the findings from the data and the validity of the relationship between theory and the data (Bachman & Schutt, 2014, p. 37).

Research Design, Philosophical Orientation, and Methodology

This research is a single descriptive and interpretive case study approach using both quantitative and qualitative methods. This study draws on data from three quantitative data sets, agency composition data, application data, and separation data; as well as semi-structured



interviews of personnel responsible for the implementation and oversight of recruiting, hiring, and retention practices for the agency.

The Case Study as the Research Approach

This research uses elements of Yin's (2009) case study methods for the philosophy, design, data collection, and data analysis. "Each method has peculiar advantages and disadvantages, depending upon three conditions: the type of research question, the control the investigator has over actual behavioral events, and the focus on contemporary as opposed to historical data" (p. 2). Yin further states that case studies are an appropriate method "when (a) how" or "why" questions is being posed, (b) the investigator has little control over events, and (c) the focus is on a contemporary phenomenon with a real life context" (p. 2). Yin also underscores that case studies address the need to "understand complex social phenomena" (p. 4). Yin (2009) includes two parts to his "technical definition of case studies":

- 1. A case study is an empirical inquiry that
 - investigates a contemporary phenomenon in depth and within its real-life context, especially when
 - the boundaries between phenomenon and context are not clearly evident.
- 2. The case study inquiry
 - copes with the technically distinctive situation in which there will be many more variables of interest than data points, and as one result
 - relies on multiple sources of evidence, with data needing to converge in a triangulating fashion, and as another result
 - benefits from the prior development of theoretical propositions to guide data collection and analysis. (p. 18)



The phenomenon under study in this research meets the criteria of understanding complex social phenomena and where the boundaries between the phenomenon and the context can be blurred. This research describes what is observed and interpreted from the data. Three different quantitative data sets provide information on the composition of the agency, applicants, the application process, and officers who separated from the agency. One qualitative method, semi-structured interviews, provides information on recruiting, hiring, and separation from the perspectives of key personnel involved in the implementation and oversight of recruiting, hiring, and retention practices for the agency selected as the case. An advantage of the one-on-one interview method is a more detailed and nuanced picture of the complexity and interaction of several variables involved in recruiting, hiring, and retention of diverse officers. Explanations and descriptions developed inductively from qualitative research are richer but based on fewer cases, and for this research, only one case. This researcher cannot assume that the people interviewed in this setting and context are similar to other people, settings and agencies. The semi-structured interviews allow the persons interviewed to tell their stories and view of reality from their perspective (Baxter & Jack, 2008, p. 545). This helps the researcher better understand their actions and to describe and interpret the data sets. In addition, the semi-structured interviews provide data that extend and contribute to existing knowledge regarding recruiting, hiring, and retaining diversity within the law enforcement ranks.

According to Berg and Lune (2012), descriptive case studies are where a descriptive theory provides the overall framework to be used before research questions and the case are formulated (p. 338). The literature review identifies relevant factors to this study, so they can be compared to the findings and interpretations of the data. Case studies may also be defined by their purpose. Della Porta and Keating (2008) identified four main types of case studies, each



representing a different purpose: descriptive, interpretive, hypothesis-generating and refining, and theory evaluating (pp. 227-228). This research uses their definition of an interpretive case study in addition to the descriptive nature of the case study. Della Porta and Keating defined an interpretive case study as employing "theoretical frameworks to provide an explanation of particular cases, which can lead as well to an evaluation and refinement of theories" (p. 227). This research uses theoretical and conceptual frameworks to guide data collection, what concepts and variables to look for, how the research questions should be framed, and how the case should be interpreted.

Research Questions

Presented below are the primary research question and seven sub-questions for this study previously presented in Chapter I.

Primary Research Question

How does one law enforcement agency in the Midwest recruit, hire, and retain full-time police officers that are representative of the communities it serves?

Sub-question #1

What is the gender and racial/ethnic composition of officers?

Sub-question #2

What is the gender and racial/ethnic composition of applicants for each step of the application process?

Sub-question #3

What are the turnover rates, reasons for separation, length of service, gender, and race/ethnicity of officers who separated?



Sub-question #4

What practices are used by the agency for recruiting, hiring, and retaining diversity?

Sub-question #5

What factors are important for recruiting, hiring, and retaining diversity within the agency?

Sub-question #6

What characteristics are important in leadership for recruiting, hiring, and retaining diversity within the agency?

Sub-question #7

What modifications in the traditional recruiting, selection, and hiring processes are needed to recruit and hire for a diverse police agency?

Data Sources, Data Collection, and Data Analysis

There is a total of four data sets, three quantitative data sets and one qualitative data set.

The first data set, composition data, examines the percentages (passive representation) of officers of all ranks for each year from 2009 through 2017 by gender and the five categories of race/ethnicity used by the case agency:

- White
- Black / African American
- Hispanic / Latino
- Asian / Pacific Islander
- American Indian / Alaskan Native.

The second data set, applicant data, examines frequencies and percentages of the passing rates for each group of applicants in each step of the application process from 2009 through 2016



by gender and race/ethnicity. There was a total of 13 job postings, seven job postings open to all applicants (external and internal applicants) and six job postings open to only internal employees. The steps in the hiring process include a written application with supplemental questions, a minimum and desired qualifications review, a written test, an interview, a second interview, a background investigation, a physical examination, a drug screen, and a psychological examination. There are differences in the steps in the application process for the different job postings as described in Chapter IV, Data Analysis. For example, one posting had the minimum and desired qualifications as separate steps, and the remaining postings had them combined as one step; some postings included a written test, and some postings did not; and some postings had the background investigation, physical examination, drug screen, psychological examinations, and second interview as separate steps, and some postings included these steps with the hiring step.

The third data set, separation data, examines turnover rates, reason for separation, and the length of service for gender and race/ethnicity for full-time police officers of all ranks who separated from the agency from 2010 through mid-2017. This data is compared to the separation rates in Wareham et al.'s (2015) study on police officer turnover using the 2003 LEMAS survey data and the 2008 CSLLEA data. This data is also compared to Reaves' (2012) separation rates using the 2008 CSLLEA data.

The fourth data set consists of interviews of personnel responsible for the implementation and oversight of recruiting, hiring, and retention practices for the case agency. Analysis of this data examines factors that influence the case agency's recruiting, hiring, and separation policies and practices from the perceptions of those involved in these practices. The semi-structured



interviews seek to provide context for the composition, application, and separation data identified above.

Description of Semi-structured Interviews

The semi-structured interview consists of a number of predetermined questions covering the topics, concepts, and variables delineated in sub-questions #4-7, as well as additional questions, depending on responses from the individuals interviewed. This allows the interview to capture the subject's perspective, described in their own words, and how participants attach meaning to the phenomenon of recruiting, hiring, and retaining diversity in law enforcement. The order, number, and content of the questions vary for each interview, as the interviewer adapts to the progression and flow of the interview.

Interview schedule. The names of interview respondents are not identified in the recordings, transcripts, or dissertation, instead, each respondent is assigned a number. The researcher has questions to begin and guide the discussion. It is important to capture the subject's own words to describe the processes of recruiting, hiring and retention. Additional questions are asked to clarify responses, check for understanding, encourage subjects to elaborate on their responses, to follow up on responses, or to explore an unexpected topic. The interview guide is Appendix A.

Data analysis for interviews. For the interviews, this researcher uses the process of content analysis to analyze the interview data suggested by Yin (2009) and Berg and Lune (2012). Berg and Lune stated that the process is to "carefully and minutely read the document line by line and word by word to determine the concepts and categories that fit the data" (p. 369). Concepts are developed, which lead to further thinking about the data and answers to the



questions, which lead to more questions about "conditions, strategies, interactions, and consequences of the data" (p. 368).

The following summarizes the various stages involved in qualitative analysis that are used in this research:

- Identify the research questions(s).
- Determine analytic categories (develop sociological constructs by sorting themes or categories of various chunks of data).
- Read through the data and establish grounded categories (open and axial coding).
- Determine systematic (objective) criteria of selection for sorting data chunks into analytic and grounded categories.
- Begin sorting the data into the various categories (revise categories or selection criteria).
- Count the number of entries in each category for descriptive statistics and to allow for the demonstration of magnitude.
- Review textual materials as sorted into various categories seeking patterns.
- Consider the patterns in light of relevant literature and/or theory.
- Offer an explanation and analysis for your findings.
- Relate analysis to the extant literature of the subject. (Berg & Lune, 2012, p. 373)

"The first formal analytical step is documentation" (Bachman & Schutt, 2014, p. 263).

Documentation helps keep track of all the data collected and helps in developing an outline for the process of analysis and ongoing conceptualization on the data (p. 263). "Identifying and refining important concepts is a key part of the iterative process of qualitative research" (p. 263).



It is important to read the notes and transcriptions from the interviews to note important statements and ways to code the data, list concepts reflected, diagram relationships among concepts, and then refine concepts throughout data collection and analysis (Bachman & Schutt, 2014, p. 262). Once patterns are found in the data, these are compared to previous research and existing theory to look for similarities and differences (Berg & Lune, 2012; Yin, 2009).

Themes are presented that emerge from the interviews, and descriptions and explanations are developed inductively and are authentic because people say it in their own words (Bachman & Schutt, 2014). The researcher identifies emerging patterns for the codes and code combinations (Yin, 2009). It is important to code and interpret the content in the data that addresses the research questions for this study. Codes and categories are derived from identified relevant theory, identified concepts and variables, and the data itself.

Data Analysis Techniques and Strategies

Baxter and Jack (2008) and Yin (2009) emphasized that to increase confidence in the findings, it is important during analysis in any case study to ensure that the research questions are addressed and are within the scope of the study, and to explore alternate explanations and rival propositions for a phenomenon. Yin (2009) also describes pattern matching as an appropriate strategy for descriptive case studies if the "predicted pattern of specific variables is defined prior to data collection" (p. 137). If the patterns are similar, the internal validity in the case study is increased (Yin).

Triangulation of Data Sources, Methods, and Levels of Analysis

There are many factors that influence recruiting, hiring, and retaining diverse law enforcement officers. Because of the "richness of the phenomenon and the extensiveness of the real-life context", "multiple sources of evidence" should be used so data will "converge in a



triangulating fashion" (Yin, 2009, p. 2). Bachman and Schutt (2014) define triangulation as using multiple methods to examine one research question and/or using two or more different measures for the same variable. Triangulation can be used to strengthen reliability and measurement validity, resulting in more confidence in the validity of each method if similar results are obtained with the other methods (p. 85). Triangulation is defined by Denzin (1970) as the use of a mix of multiple methodologies, data sources, investigators, and theories in the examination of the same phenomena and that "each method has implied a triangulated perspective" (p. 297). Denzin also recommends that researchers should analyze their "problem from as many different methodological perspectives as possible" (p. 297). He defines research methods "as lines of action" where "each research method reveals peculiar elements of symbolic reality" (p. 298). Berg and Lune (2012) further state that "each method, thus, reveals slightly different facets of the same symbolic reality. Every method is a different line of sight directed toward the same point, observing particular aspects of the social and symbolic reality" (p. 6). Berg and Lune further state that "by combining several lines of sight, researchers can obtain a better, more substantive picture of reality; a richer, more complete array of symbols and theoretical concepts; and a means of verifying many of these elements" (p. 6). They refer to the multiple lines of sight as triangulation (p. 6). This research uses triangulation of multiple data sources and multiple methods to address the research questions and variables to increase reliability and validity. This is consistent with Berg and Lune's statement that the most important aspect of triangulation is not just in combining different types of data, "but the attempt to relate them so as to counteract the threats to validity identified in each" (p. 6).

Many researchers define the case selected as the unit of analysis (Baxter & Jack, 2008; Yin, 2009). The unit of analysis for this single case study is an agency that is recognized for



recruiting and hiring females police officers and police officers of color. Four data sources, previously discussed, are drawn upon. The first data source is at the organizational level, analyzing percentages of full-time officers by gender and race/ethnicity. The second data source is at the group level, analyzing groups of applicants for the position of police officer. This is a descriptive analysis of data, including all steps in the application process and the numbers (frequencies) and percentages of officers who successfully pass each step of the application process by gender and race/ethnicity. The third data source is also at the group level, analyzing full-time police officers who have separated from the agency, to include turnover rates, reason of separation, length of service, gender and race/ethnicity. The fourth data source is at the individual level, analyzing semi-structured interviews of personnel responsible for the implementation and oversight of recruiting, hiring, and retention practices for the agency.

The Case Selected for Case Study

The methods employed for this research benefit from a purposive selection of the case. Two other police agencies were approached but did not allow access to their data. To examine the variables and factors that influence recruiting, hiring, and retaining diversity within law enforcement agencies, the case selected for this research is an agency recognized for intentionally recruiting and hiring female police officers and police officers of color. Random selection is rarely used to select subjects for the interviews. Rather, careful and purposive selection should be used (Bachman & Schutt, 2014, p. 255). To identify and describe what factors, practices, and characteristics lead to recruiting, hiring, and retaining diversity, the individuals selected to be interviewed are personnel within the case agency who are responsible for the implementation and oversight of recruiting, hiring, and retention practices.



Description of Case Agency

The agency requested non-disclosure of any identifying details, so, to protect the anonymity of this agency and the individuals interviewed, details that can identify this agency are omitted. Also, the agency did not permit this researcher to survey current officers about their experiences and perceptions on recruiting, hiring, and retention.

The case agency is a large agency in a metropolitan area that serves a population of about 175,000 people. Data from the U.S. Census Bureau were used to calculate the percentages for the demographic make-up of the population that the case agency serves for the five categories of ethnic groups used in the application, composition, and separation data. There were 79% White, 5% Hispanic / Latino, 8% Black / African American, 6% Asian / Pacific Islander, and 0.6% American Indian / Native Alaskan (U.S. Census Bureau, 2013). The remaining percentages were represented by the "Other" category or by "two or more races".

This agency is committed to fostering a diverse department by increasing the representation of female officers, officers of color, and officers who are fluent in languages other than English. They are also committed to engaging in community-oriented and problem-oriented policing by establishing relationships and partnerships between officers and citizens, which emphasizes protecting their communities and improving the quality of life in their neighborhoods. This study examines how community-oriented and problem-oriented policing impact this agency's recruitment, hiring, and retention of diverse officers.

The hiring process is influenced by the state regulatory board's minimum selection standards and the Human Resources Office rules and regulations. Recruitment efforts by the agency and the Human Resources Office included providing job postings to several professional and diverse agencies, associations, and organizations; post-secondary institutions; and veterans



organizations. The Human Resources Office is both an Affirmative Action and Equal Opportunity Employer that aims for their work force to be reflective of their communities and ensures that their job postings are inclusive.

Ethical Issues

This researcher obtained approval from the University of North Dakota's (UND) Institutional Review Board (IRB) for this study, following procedures on the protection of human subjects.

Protection of Participants from Harm

For the data on police officer composition, application, and separation, there are not any foreseeable harms or risks from analyzing the data because there were no individual identifiers included in the data provided. Their employment status was and will not be affected by the data collected for the study. For the individuals interviewed, there are no known risks to those who were interviewed. The benefits from this study include providing data that helps other agencies to recruit, hire, and retain diversity.

Voluntary Participation

Permission was obtained from the agency and its Human Resources Office for the data on police officer composition data, applicant's progression through the selection process, and the separation data. No personal identifiable information was given to the researcher. Permission from the agency was also obtained for the interviews.

Informed Consent

For the interviews, each person interviewed was given an informed consent form stating the purpose of the research, risks and benefits, and that their participation is voluntary. Each



person interviewed signed the consent form agreeing to participate in the interview and whether they agreed to digitally have their interview recorded.

Confidentiality and Anonymity

The data on the composition of the agency, applicants for the position of police officer, and separation were given to this researcher without any information to identify individual applicants, assuring anonymity. To protect the identity of participants and the agency, interview participants and the agency are not identified by name. All information that identifies the agency and individuals is omitted.

Securing the Data

The data regarding the steps in the hiring process for police officers is stored on a password protected laptop. All hard copies of research data and documents, and interview transcripts, are locked in a file drawer. Consent forms from the people interviewed are locked in a separate file drawer. The interviews were recorded on a digital recorder and transcribed by a professional transcriptionist who met the approval of the UND's IRB. The company hired signed a confidentiality/disclosure agreement that was approved by the UND IRB. The digitally recorded interviews were deleted after transcripts were produced. The transcriptions of the interviews are stored in password protected files on the computer and hard copies of the interview transcripts are secured in a locked file drawer. All data and consent forms will be destroyed and/or deleted after the required length of retention has expired.

Reliability and Validity

The triangulation of several data sources and methods increases the reliability and validity of the data. Reliability is defined as when measures give consistent scores on different occasions or under similar research conditions (Bachman & Schutt, 2014, p. 88; Lanier &



Briggs, 2014, p. 83). Yin (2009) defines reliability as demonstrating that the study's methods and operations can be repeated with the same results (p. 40). Reliability is also concerned with replicating the data, which includes whether another researcher using similar or the same methods, can make similar observation (Shaffir & Stebbins, 2003, p. 8). Yin (2009) suggests to document all of the procedures by using a case study protocol and a case study database to increase reliability, this research uses both to reduce bias and errors in the study.

Validity is concerned with obtaining an accurate or true picture of the phenomenon under study (Bachmann & Schutt, 2014; Hagan, 2012; Shaffir & Stebbins, 2003). Using multiple data sources, methods, and levels of analysis has the advantage of utilizing the strengths of one method or measurement to address the weaknesses and threats to reliability or validity of another method or measurement (Hagan, 2012). This is consistent with Denzin's (1970), Yin's (2009), and Berg and Lune's (2012) converging lines of inquiry to address the same research question or area of interest. This increases the confidence in the validity of the measures and the findings.

Berg and Lune (2012) state that a serious problem in asking interview questions is that the intentions of the questions may not be communicated accurately to the subject (p. 123). This could affect the validity of the responses if the concepts and variables are not measured as intended. To decrease this potential threat to validity, the interview schedule was pilot, or pretested, by key informants who are knowledgeable about the subject matter. They reviewed the interview schedule for the wording of questions, terminology used, understanding or misunderstanding of the content, and intent of the questions. The pre-test also allowed several practice interviews to determine whether the interview is effective in obtaining the information, concepts and variables intended (p. 127).



The main threats to validity for this research are reactivity of the researcher's presence in the interviews and researcher bias. Bachman and Schutt (2014) define a reactive effect as "the changes in individual or group behavior that are due to being observed or otherwise studied" (p. 242). For the three data sets: composition data, application data, and separation data, reactivity was not a problem because all three data sets are unobtrusive methods of existing data collected and are used for other purposes not influenced by this researcher. Hagan (2012) defined unobtrusive measures as "nonreactive methods of gathering data" where subjects are not aware that they are subjects of research (p. 239). If subjects are not aware that they are being researched, then reactivity cannot be a threat to validity. Therefore, if the findings from the data and the interviews are similar, then any reactivity from the researcher interviewing the subjects cannot be responsible for influencing the findings. This is what Hagan refers to as methods being used as a variable or rival causal factor (p. 286). Hagan also stated that an obstacle with case studies is possible bias on the part of the researcher (p. 233). This researcher is aware of her role as the interviewer and how this may bias the data or affect reactivity. This researcher is also aware of how perceiving and interpreting the data may be influenced by her personal and professional experiences and viewpoints, so as not to bias the data and findings.

One way to increase validity and ensure that the data collected from the interviews is an accurate reflection of those interviewed, is to ask clarifying questions (extra and probing questions) that get at the same issue but are asked differently. This researcher also had the interview respondents review a transcript of their interviews to ensure that their responses are an accurate reflection of their interviews, and to check for any errors. This also serves as a form of "member validation" (Shaffir & Stebbins, 2003, p. 10).



Internal Validity

As stated by Yin (2009), internal validity applies to explanatory and causal studies, and does not apply to a descriptive or exploratory case study. Thus, the threats to internal validity for experiments, quasi-experiments, and non-experiments do not apply to this research. However, Yin states that one of the most desirable analytic techniques, pattern matching, is still applicable for descriptive case studies, "as long as the predicted pattern of specific variables is defined prior to data collection" (p. 137). If the patterns are similar, the internal validity in the case study is increased.

Construct Validity

Measurement validity can be established through construct validity. This means that the measures used in the study are actually measuring the concepts and variables of interest and that a measure is related to other measures from a theory (Bachman & Schutt, 2014, p. 87; Lanier & Briggs, 2014, p. 104). Yin admits that construct validity "is especially challenging in case study research" (p. 41). This underscores the necessity to carefully operationalize the concepts and measures to be used in this study. Yin (2009) suggest three tactics to increase construct validity when conducting case studies:

- 1. Use multiple sources of evidence
- 2. Establish chain of evidence
- 3. Have key informants review draft case study report. (pp. 41-42)

This research uses these three tactics to increase construct validity.

External Validity

Because this research uses a single case study, the findings cannot be generalized beyond the selected case agency. However, future research can be conducted on other types of law



enforcement agencies, in different geographic locations, and in different sized agencies to examine the influence and interaction of different variables that predict success in recruiting, hiring, and retaining diversity. Future research can increase external validity by comparing their findings to the findings of this study. It is the hope that this study contributes to and refines the theories of representative bureaucracy and social equity. The findings of this case study may be used by other police agencies as an example on how to recruit, hire and retain diversity.

Authenticity

Bachman and Schutt (2014) discuss that authenticity should be a proposed standard used in qualitative research, where the focus is on the various subjective interpretations and perspectives of how the participants view their social world (p. 43).

Conclusion

This chapter presented the purpose of the study, theoretical and conceptual framework, and the research design, philosophical orientation, and methodology. This chapter also discussed why the case study is an appropriate approach for this study, the research questions, and data sources, collection, and analysis. This chapter also discussed triangulation of data sources, methods, and levels of analysis, the agency selected for the case study, and described the case agency. Discussed last were ethical issues, and reliability, validity, and authenticity.

The next chapter, chapter IV, presents the results of the three quantitative data collection methods and Chapter V presents the results of the qualitative interview data.



CHAPTER IV

COLLECTION AND ANALYSIS OF QUANTITATIVE DATA

This chapter presents the results of three quantitative data sets used for this mixed methods case study. The first set of data analyzed is composition data, consisting of the percentages of full-time police officers for all ranks by gender and race/ethnicity for each year from 2009 through 2017. The second set of data analyzed is applicant data, consisting of the frequencies and percentages of applicants for full-time police officers from 2009 through 2016, for each step in the application process by gender and race/ethnicity. The third set of data analyzed is separation data, consisting of agency turnover rates, reason for separation, length of service, gender, and race/ethnicity for all full-time police officers for all ranks who separated from 2010 through mid-2017. The results of the analysis for the fourth and last set of data, qualitative semi-structured interviews, are presented in Chapter V, Collection and Analysis of Interviews. The interviews consist of the perceptions from personnel responsible for the implementation and oversight of recruiting, hiring, and retention practices of full-time police officers at the case agency.

Research Questions

This study has one primary research question and seven sub-questions to address the main research question. The data presented in this chapter address sub-questions #1 through #3.



Primary Research Question

How does one law enforcement agency in the Midwest recruit, hire, and retain full-time police officers that are representative of the communities it serves?

Sub-question #1

What is the gender and racial/ethnic composition of officers?

Sub-question #2

What is the gender and racial/ethnic composition of applicants for each step of the application process?

Sub-question #3

What are the turnover rates, reasons for separation, length of service, gender, and race/ethnicity for officers who separated?

Full-time Police Officer Composition Data

The composition data includes all full-time police officers of all ranks for each year from 2009 through 2017 by the percentages for gender and the five categories of race/ethnicity used by the case agency. All of the tables in this chapter use the label "Ethnic Group" for race/ethnicity because that is the label used by the case agency. The following are the five ethnic groups:

- White
- Hispanic / Latino
- Black / African American
- Asian / Pacific Islander
- American Indian / Alaskan Native



As observed in table 4, in 2009, 3.7% of the officers were female, and the percentage increased over four and one-half times to 17.0% in 2017.

Table 4. Percent of Full-time Police Officers for each Year from 2009 to 2017 by Gender.

Year	2009	2010	2011	2012	2013	2014	2015	2016	2017
Male	96.3%	92.5%	90.8%	90.6%	89.4%	88.2%	86.7%	83.5%	83.0%
Female	3.7%	7.5%	9.2%	9.4%	10.6%	11.8%	13.3%	16.5%	17.0%
Total	100%	100%	100%	100%	100%	100%	100%	100%	100%

Table 5 presents the percentages of officers of color for each year from 2009 to 2017. In 2009, there were 13.0% of officers of color, and the percentage increased over two times to 29.5% in 2017. In 2009, there were 13.0% officers of color, in 2010 the percentage increased to 13.5%, in 2011 and 2012, the percentages decreased, and from 2013 to 2017, the percentages increased each year to 29.5% in 2017.

Table 5. Percent of Full-time Police Officers of Color for each Year from 2009 to 2017.

Year	2009	2010	2011	2012	2013	2014	2015	1016	2017
Officers of Color	13.0%	13.5%	10.8%	9.4%	10.6%	20.0%	25.6%	29.3%	29.5%

Table 6 presents the percent of full-time officers for all ranks by ethnic group for each year from 2009 to 2017. The highest percentage of a non-White group was Black / African American with 14.7% in 2016. This group also had the highest percentages in 2013 (tied with Hispanic / Latino), 2014, 2015, 2016, and 2017, with 4.55%, 11.8%, 14.5%, 14.7% and 14.3% respectively. All non-White groups except American Indian / Alaskan Native were represented in each year. Notable are the increases in the representation of Black / African American officers and Asian / Pacific Islander officers. Comparing the percentages between 2009 and 2017 for Black / African American officers, there was an increase of 286% from 3.7% to 14.3%. Comparing the percentages between 2009 to 2017 for Asian / Pacific Islander officers, there was an increase of 381% from 1.85% to 8.9%



Table 6. Percent of Full-time Police Officers for each Year from 2009 to 2017 by Ethnic Group.

Year	2009	2010	2011	2012	2013	2014	2015	2016	2017
White	87.0%	86.5%	89.2%	90.6%	89.4%	80.0%	74.4%	70.7%	70.5%
Hispanic /	5.6%	6.0%	4.6%	4.7%	4.55%	5.9%	6.7%	7.3%	5.4%
Latino									
Black /	3.7%	4.5%	3.1%	3.1%	4.55%	11.8%	14.5%	14.7%	14.3%
African									
American									
Asian /	1.85%	1.5%	1.55%	1.6%	1.5%	2.3%	3.3%	6.4%	8.9%
Pacific									
Islander									
American	1.85%	1.5%	1.55%	0%	0%	0%	1.1%	.9%	.9%
Indian /									
Alaskan									
Native									
Total									
Full-time	100%	100%	100%	100%	100%	100%	100%	100%	100%
Officers									

Full-time Police Officer Applicant Data

The case agency has a policy on recruitment and selection, which includes using recruitment and selection strategies to target qualified and diverse candidates; working with community groups, local colleges and universities, and the military. The policy also includes not discriminating in their recruiting and screening processes and reaching out to people who have integrity and who are ethical and service-minded. The Human Resources Office has a formal Affirmative Active plan and is an Equal Opportunity Employer that aims for its work force to be reflective of their communities and ensures that their job postings are inclusive.

The minimum qualifications for all of the full-time police officer postings begins with meeting minimum selection standards for a police officer as required by the state regulatory board and possessing a valid driver's license with a good or an acceptable driving record. The required minimum selection standards for police officers by the state regulatory board include being a citizen of the U.S., completing a detailed written application, and passing an oral



interview to assess communication skills necessary to perform the tasks of a police officer. The minimum selection standards also include passing a thorough background investigation, not having any felony convictions, and not having any convictions of specified offenses, such as assault, domestic assault, crimes against a vulnerable adult, narcotics or controlled substance, and theft. Finally, the minimum selection standards include passing a psychological evaluation with an oral interview by a licensed psychologist, and passing a job related medical examination to ensure that applicants are free from any physical or mental condition that will affect the performance of police officer duties. First aid certification is a requirement by the state regulatory board for police officers and it is listed as a minimum qualification in some postings and as a desired qualification in other postings. The minimum qualifications are specified in each posting described in this chapter.

Desired qualifications varied for the seven job postings open to external and internal applicants. These include variations of the following: having a college degree, prior police experience, prior experience in non-paid non-sworn police related positions, prior security experience, relevant volunteer experience, fluency in languages other than English, and experience working with diverse populations. The desired qualifications are specified in each posting described in this chapter.

As part of the detailed written application completed by applicants, supplemental questions were included to address the minimum and desired qualifications. The purpose of the supplemental questions was to evaluate applicant's qualifications, skills, and experiences for eligibility. The supplemental questions were reviewed to determine who passed the qualifications review. Applicants were instructed that if they responded "yes" to a question, they were to answer the narrative response question that followed. Supplemental questions



varied by the different postings. All seven postings open to both internal and external applicants asked applicants whether they were eligible to be licensed under the state regulatory board, had a state driver's license and an acceptable or a good driving record, had first aid certification, and had prior police experience. Five of the postings, in 2011, 2012, 2013, 2014, and 2016, in addition to the questions asked in previous applications, asked whether applicants had relevant volunteer experience and a degree from an accredited university or college. Three postings, 2013, 2014, and 2016, in addition to the questions asked in previous applications, asked applicants whether they had fluency in languages other than English. Two postings, in 2014 and 2016, in addition to the questions asked in previous applications, asked applicants whether they had relevant work experience, non-paid non-sworn experience, and experience working with diverse populations. The last posting, in 2016, in addition to the questions asked in previous applications, asked applicants why they wanted to work for this agency.

Steps in the application process for the seven job postings open to both internal and external applicants included variations of the following: a written application, including supplemental questions to address minimum and desired qualifications; a minimum and desired qualifications review; a written test; an interview; a second interview; a background investigation; a medical examination; a drug screen; and a psychological evaluation. The following are examples of variations in the steps in the application process: one posting separated the minimum qualifications review and desired qualifications review, and the remaining postings combined them into the qualifications review; some job postings included a written test, and other postings did not include a written test; some postings had the second interview after the background investigation, and other postings had the second interview before the background investigation; and some postings had the second interview, background

investigation, and the medical exam, drug screen and psychological evaluation as one step, and other postings had them as separate steps. The steps in the selection process are specified for each posting described in this chapter.

Supplemental questions were reviewed to determine whether applicants met the minimum and desired qualifications. In the application data that follows, applicants who did not pass the minimum qualifications were applicants who did not meet the minimum qualifications for the posting. Applicants who met the minimum requirements step but did not advance onto the next step, were eliminated after reviewing the supplemental questions for desired qualifications.

The application process included a written test for the job postings in 2011, 2012, and 2013. The written test given to applicants was timed and law enforcement specific. The application process did not include a written test for the job postings in 2009, 2010, 2014, and 2016.

For all postings, the first interview consisted of a panel of at least three people within the agency drawn from all ranks. In addition, some interview panels included a person from outside the case agency. Questions in the interview focused on life experiences and were not law enforcement specific questions. For the job postings in 2009, 2010, and 2011, there was a second interview conducted after the background investigation. For the job postings in 2012, 2013, 2014, and 2016, the second interview was moved from after the background investigation to before the background investigation and was conducted by a member of senior command within the case agency.

After the applicant passed the previous steps and was considered for hiring, a conditional job offer was given contingent upon the applicant passing the medical examination, drug screen,



and psychological evaluation, which were all contracted outside the agency. The medical examination was given by a licensed physician to check that the applicant could perform the duties of a police officer and the applicant was given a drug screen. For the psychological evaluation, the applicant completed a written psychological test and was interviewed by a licensed psychologist. If the applicant passed these last steps, a final job offer was given.

As part of the application process, applicants were asked to self-identify their gender, male or female, and one of the following groups of race/ethnicity utilized by the case agency:

- White
- Hispanic / Latino
- Black / African American
- Asian / Pacific Islander
- American Indian / Alaskan Native

Providing gender and racial/ethnic identification was voluntary, therefore, applicants could leave either or both gender and ethnic group blank, which are marked as *missing* in the tables throughout the analysis of the applicant data. The applicant data for all steps in the application process includes the gender and race/ethnicity that applicants identified in their application form. As previously stated, although the five categories include both race and ethnicity, the tables will use the label, "Ethnic Group", because that is the label used by the case agency.

2009 Posting for Full-time Police Officer External and Internal Applicants

Minimum requirements in the 2009 job announcement were meeting the state regulatory board's eligibility requirements for a police officer, a valid driver's license with a good driving record, and first aid certification. The only desired qualification was prior police experience.



The steps in the selection process for the 2009 posting included the following: step 1, minimum and desired qualifications review; step 2, an interview; and step 3, the hiring step, which included a background investigation, a second interview, a medical examination, a drug screen, and a psychological evaluation.

There were 401 applications received for the 2009 posting. Table 7 presents the frequency (n) and percent for gender. Of the 401 applicants, 345 (86.0%) were male and 52 (13.0%) were female. Four applicants omitted gender. Table 8 presents the frequency (n) and percent for each of the five ethnic groups. Of the 401 applicants, 314 (78.3%) were White and 82 (20.5%) were applicants of color. Five applicants omitted ethnic group.

Table 7. Gender of 2009 Full-time Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 345	86.0%
Female	n = 52	13.0%
Missing	n = 4	1.0%
Total	n = 401	100.0%

Table 8. Ethnic Group of 2009 Full-time Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 314	78.3%
Hispanic / Latino	n = 16	4.0%
Black / African American	n = 34	8.5%
Asian / Pacific Islander	n = 29	7.2%
American Indian / Alaskan Native	n = 3	0.8%
Missing	n = 5	1.2%
Total	n = 401	100.0%

Of the 401 applicants entering step 1 of the application process, minimum and desired qualifications review, 57 applicants did not meet the minimum qualifications, 272 applicants did not advance onto step 2 after review of the supplemental questions for desired qualifications, and 72 applicants passed the qualifications review and advanced onto step 2, an interview. Table 9



presents the frequency (n) and percent for gender. Of the 72 applicants who passed step 1, 64 (88.9%) were male and 7 (9.7%) were female. One applicant omitted gender. In this step, female applicants decreased 25% compared to 13% who applied. Table 10 presents the frequency (n) and percent for each of the ethnic groups. Of the 72 applicants who passed step 1, 60 (83.3%) were White and 10 (13.9%) were applicants of color. Two applicants omitted ethnic group. In this step, applicants of color decreased 32% compared to 20.5% who applied.

Table 9. Gender of 2009 Full-time Police Officer Applicants Passing Step 1.

Gender	Frequency	Percent
Male	n = 64	88.9%
Female	n = 7	9.7%
Missing	n = 1	1.4%
Total	n = 72	100.0%

Table 10. Ethnic Group of 2009 Full-time Police Officer Applicants Passing Step 1.

Ethnic Group	Frequency	Percent
White	n = 60	83.3%
Hispanic / Latino	n = 3	4.2%
Black / African American	n = 4	5.5%
Asian / Pacific Islander	n = 2	2.8%
American Indian / Alaskan Native	n = 1	1.4%
Missing	n = 2	2.8%
Total	n = 72	100.0%

Of the 72 applicants entering step 2 of the application process, an interview; 7 applicants did not advance onto step 3 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; 1 applicant withdrew from the application process; and 64 applicants passed the interview and advanced onto step 3, the hiring step. Table 11 presents the frequency (n) and percent for gender. Of the 64 applicants who passed step 2, 56 (87.5%) were male and 7 (10.9%) were female. One applicant omitted gender. In this step, female applicants increased 12% compared to 9.7% of females in the previous step.



Table 12 presents the frequency (n) and percent for each of the ethnic groups. Of the 64 applicants who passed step 2, 52 (81.3%) were White and 10 (15.6%) were applicants of color. Two applicants omitted ethnic group. In this step, applicants of color increased 12% compared to 13.9% of applicants of color in the previous step.

Table 11. Gender of 2009 Full-time Police Officer Applicants Passing Step 2.

Gender	Frequency	Percent
Male	n = 56	87.5%
Female	n = 7	10.9%
Missing	n = 1	1.6%
Total	n = 64	100.0%

Table 12. Ethnic Group of 2009 Full-time Police Officer Applicants Passing Step 2.

Ethnic Group	Frequency	Percent
White	n = 52	81.3%
Hispanic / Latino	n = 3	4.7%
Black / African American	n = 4	6.3%
Asian / Pacific Islander	n = 2	3.1%
American Indian / Alaskan Native	n = 1	1.5%
Missing	n = 2	3.1%
Total	n = 64	100.0%

Step 3 of the application process, the hiring step, included a background investigation, a second interview, a medical examination, a drug screen, and a psychological evaluation. Of the 64 applicants entering the hiring step, 49 applicants did not get hired and remained inactive when the list expired. For the applicants who passed the background investigation and a second interview, and were considered for a full-time police officer position, a conditional job offer was given, contingent upon passing the medical examination, drug screen, and psychological evaluation. A final job offer was given to the applicants who passed the medical examination, drug screen, and psychological evaluation. Fifteen applicants were hired. Table 13 presents the frequency (n) and percent for gender. Of the 15 applicants hired, 9 (60.0%) were male and 5



(33.3%) were female. One applicant omitted gender. In this step, female applicants increased over three times compared to 10.9% of females in the previous step. In addition, the 33.3% of female applicants hired was about two and one-half times higher than the 13% who applied. Table 14 presents the frequency (n) and percent for each of the five ethnic groups. Of the 15 applicants hired, 11 (73.3%) of the applicants were White and 2 (13.4%) were applicants of color. Two applicants omitted ethnic group. In this step, applicants of color decreased 14% compared to 15.6% applicants of color in the previous step. In addition, the 13.4% applicants of color hired was 35% lower than the 20.5% who applied.

Table 13. Gender of 2009 Full-time Police Officer Applicants Hired – Step 3.

Gender	Frequency	Percent
Male	n = 9	60.0%
Female	n = 5	33.3%
Missing	n = 1	6.7%
Total	n = 15	100.0%

Table 14. Ethnic Group of 2009 Full-time Police Officer Applicants Hired – Step 3.

Ethnic Group	Frequency	Percent
White	n = 11	73.3%
Hispanic / Latino	n = 1	6.7%
Black / African American	n = 1	6.7%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 2	13.3%
Total	n = 15	100.0%

2010 Posting for Full-time Police Officer External and Internal Applicants

As was the case in the 2009 posting, the minimum requirements in the 2010 job announcement were meeting the state regulatory board's eligibility requirements for a police officer, a valid driver's license with a good driving record, and first aid certification. As was the case in the 2009 posting, the only desired qualification was prior police experience.



The steps in the selection process for the 2010 posting included the following: step 1, minimum qualifications review; step 2, desired qualifications review; step 3, an interview; and step 4, the hiring step, which included a background investigation, a second interview, a medical examination, a drug screen, and a psychological evaluation.

There were 489 applications received for the 2010 posting. Table 15 presents the frequency (n) and percent for gender. Of the 489 applicants, 426 (87.1%) were male and 56 (11.5%) were female. Seven applicants omitted gender. Table 16 presents the frequency (n) and percent for each of the five ethnic groups. Of the 489 applicants, 402 (82.2%) were White and 82 (16.8%) were applicants of color. Five applicants omitted ethnic group.

Table 15. Gender of 2010 Full-time Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 426	87.1%
Female	n = 56	11.5%
Missing	n = 7	1.4%
Total	n = 489	100.0%

Table 16. Ethnic Group of 2010 Full-time Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 402	82.2%
Hispanic / Latino	n = 20	4.1%
Black / African American	n = 30	6.2%
Asian / Pacific Islander	n = 28	5.7%
American Indian / Alaskan Native	n = 4	0.8%
Missing	n = 5	1.0%
Total	n = 489	100.0%

Of the 489 applicants entering step 1 of the application process, minimum qualifications review, 194 applicants did not meet the minimum qualifications, and 295 applicants passed the minimum qualifications review and advanced onto step 2, the desired qualifications review.

Table 17 presents the frequency (n) and percent for gender. Of the 295 applicants who passed



step 1, 262 (88.8%) were male and 30 (10.2%) were female. Three applicants omitted gender. In this step, female applicants decreased 11% compared to 11.5% who applied. Table 18 presents the frequency (n) and percent for each of the five ethnic groups. Of the 295 applicants who passed step 1, 247 (83.7%) were White and 46 (15.6%) were applicants of color. Two applicants omitted ethnic group. In this step, applicants of color decreased 7% compared to 16.8% who applied.

Table 17. Gender of 2010 Full-time Police Officer Applicants Passing Step 1.

Gender	Frequency	Percent
Male	n = 262	88.8%
Female	n = 30	10.2%
Missing	n = 3	1.0%
Total	n = 295	100.0%

Table 18. Ethnic Group of 2010 Full-time Police Officer Applicants Passing Step 1.

Ethnic Group	Frequency	Percent
White	n = 247	83.7%
Hispanic / Latino	n = 12	4.0%
Black / African American	n = 17	5.8%
Asian / Pacific Islander	n = 15	5.1%
American Indian / Alaskan Native	n = 2	0.7%
Missing	n = 2	0.7%
Total	n = 295	100.0%

Of the 295 applicants entering step 2 of the application process, the desired qualifications review, 257 applicants did not advance onto step 3 after review of the supplemental questions for desired qualifications, and 38 applicants passed the desired qualifications and advanced onto step 3, an interview. Table 19 presents the frequency (n) and percent for gender. Of the 38 applicants who passed step 2, 35 (92.1%) were male and 3 (7.9%) were female. In this step, female applicants decreased 23% compared to 10.2% of females in the previous step. Table 20 presents the frequency (n) and percent for each of the five ethnic groups. Of the 38 applicants



who passed step 2, 36 (94.7%) were White and 2 (5.3%) were applicants of color. In this step, applicants of color decreased 66% compared to 15.6% of applicants of color in the previous step. Table 19. Gender of 2010 Full-time Police Officer Applicants Passing Step 2.

Gender	Frequency	Percent
Male	n = 35	92.1%
Female	n = 3	7.9%
Missing	n = 0	0.0%
Total	n = 38	100.0%

Table 20. Ethnic Group of 2010 Full-time Police Officer Applicants Passing Step 2.

Ethnic Group	Frequency	Percent
White	n = 36	94.7%
Hispanic / Latino	n = 2	5.3%
Black / African American	n = 0	0.0%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 38	100.0%

Of the 38 applicants entering step 3 of the application process, an interview, 7 applicants did not advance onto step 4 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; 9 applicants withdrew from the application process; and 22 applicants passed the interview and advanced onto step 4, the hiring step. Table 21 presents the frequency (n) and percent for gender. Of the 22 applicants who passed step 3, 20 (90.9%) were male and 2 (9.1%) were female. In this step, female applicants increased 15% compared to 7.9% of females in the previous step. Table 22 presents the frequency (n) and percent for each of the five ethnic groups. Of the 22 applicants who passed step 3, 20 (90.9%) were White and 2 (9.1%) were applicants of color. In this step, applicants of color increased 72% compared to 5.3% of applicants of color in the previous step.



Table 21. Gender of 2010 Full-time Police Officer Applicants Passing Step 3.

Gender	Frequency	Percent
Male	n = 20	90.9%
Female	n = 2	9.1%
Missing	n = 0	0.0%
Total	n = 22	100.0%

Table 22. Ethnic Group of 2010 Full-time Police Officer Applicants Passing Step 3.

Ethnic Group	Frequency	Percent
White	n = 20	90.9%
Hispanic / Latino	n = 2	9.1%
Black / African American	n = 0	0.0%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 22	100.0%

Step 4 of the application process, the hiring step, included a background investigation, a second interview, a medical examination, a drug screen, and a psychological evaluation. Of the 22 applicants entering the hiring step, 17 applicants did not get hired and remained inactive when the list expired. Of the applicants who passed the background investigation and second interview and were considered for a full-time police officer position, a conditional job offer was given, contingent upon passing the medical examination, drug screen, and psychological evaluation. A final job offer was given to the applicants who passed the medical examination, drug screen, and psychological evaluation. Five applicants were hired. Table 23 presents the frequency (n) and percent for gender. Of the 5 applicants hired, all 5 (100%) were male. In this step, all 9.1% of females from the previous step were eliminated. In addition, there were 11.5% of females who applied, however, none were hired. Table 24 presents the frequency (n) and percent for each of the five ethnic groups. Of the 5 applicants hired, 4 (80.0%) of the applicants were White and 1 (20.0%) was an applicant of color. In this step, applicants of color increased



120%, compared to 9.1% of applicants of color in the previous step. In addition, the 20% of applicants of color hired was 19% higher than the 16.8% who applied.

Table 23. Gender of 2010 Full-time Police Officer Applicants Hired – Step 4.

Gender	Frequency	Percent
Male	n = 5	100.0%
Female	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 5	100.0%

Table 24. Ethnic Group of 2010 Full-time Police Officer Applicants Hired – Step 4.

Ethnic Group	Frequency	Percent
White	n = 4	80.0%
Hispanic / Latino	n = 1	20.0%
Black / African American	n = 0	0.0%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 5	100.0%

2011 Posting for Full-time Police Officer External and Internal Applicants

As was the case in the 2009 and 2010 postings, minimum requirements in the 2011 job announcement were meeting the state regulatory board's eligibility requirements for a police officer, a valid driver's license with a good driving record, and first aid certification. In addition to having prior police experience as a desired qualification, the 2011 posting added having a college degree and relevant volunteer experience as desired qualifications.

The steps in the selection process for the 2011 posting included the following: step 1, minimum and desired qualifications review; step 2, a written test; step 3, an interview; step 4, the background step; and step 5, the hiring step, which included a second interview, a medical examination, a drug screen, and a psychological evaluation.



There were 428 applications received for the 2011 posting. Table 25 presents the frequency (n) and percent for gender. Of the 428 applicants, 379 (88.6%) were male and 46 (10.7%) were female. Three applicants omitted gender. Table 26 presents the frequency (n) and percent for each of the five ethnic groups. Of the 428 applicants, 353 (82.5%) were White and 71 (16.5%) were applicants of color. Four applicants omitted ethnic group.

Table 25. Gender of 2011 Full-time Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 379	88.6%
Female	n = 46	10.7%
Missing	n = 3	0.7%
Total	n = 428	100.0%

Table 26. Ethnic Group of 2011 Full-time Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 353	82.5%
Hispanic / Latino	n = 19	4.4%
Black / African American	n = 25	5.8%
Asian / Pacific Islander	n = 20	4.7%
American Indian / Alaskan Native	n = 7	1.6%
Missing	n = 4	1.0%
Total	n = 428	100.0%

Of the 428 applicants entering step 1 of the application process, minimum and desired qualifications review, 53 applicants did not advance onto step 2 after review of the supplemental questions for minimum and desired qualifications, and 375 applicants passed the qualifications review and advanced onto step 2, a written test. Table 27 presents the frequency (n) and percent for gender. Of the 375 applicants who passed step 1, 330 (88.0%) were male and 43 (11.5%) were female. Two applicants omitted gender. In this step, female applicants increased 7% compared to 10.7% who applied. Table 28 presents the frequency (n) and percent for each of the five ethnic groups. Of the 375 applicants who passed step 1, 315 (84.0%) were White and 57



(15.2%) were applicants of color. Three applicants omitted ethnic group. In this step, applicants of color decreased 8% compared to 16.5% who applied.

Table 27. Gender of 2011 Full-time Police Officer Applicants Passing Step 1.

Gender	Frequency	Percent
Male	n = 330	88.0%
Female	n = 43	11.5%
Missing	n = 2	0.5%
Total	n = 375	100.0%

Table 28. Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 1.

Ethnic Group	Frequency	Percent
White	n = 315	84.0%
Hispanic / Latino	n = 15	4.0%
Black / African American	n = 18	4.8%
Asian / Pacific Islander	n = 17	4.5%
American Indian / Alaskan Native	n = 7	1.9%
Missing	n = 3	0.8%
Total	n = 375	100.0%

Of the 375 applicants entering step 2 of the application process, the written test, 165 applicants did not participate in this step, 156 applicants did not pass the written test, and 54 passed the written test and advanced onto step 3, an interview. Table 29 presents the frequency (n) and percent for gender. Of the 54 applicants who passed step 2, 44 (81.5%) were male and 10 (18.5%) were female. In this step, female applicants increased 61% compared to 11.5% of females in the previous step. Table 30 presents the frequency (n) and percent for each of the five ethnic groups. Of the 54 applicants who passed step 2, 50 (92.6%) were White and 4 (7.4%) were applicants of color. In this step, applicants of color decreased 51% compared to 15.2% of applicants of color in the previous step.



Table 29. Gender of 2011 Full-time Police Officer Applicants Passing Step 2.

Gender	Frequency	Percent
Male	n = 44	81.5%
Female	n = 10	18.5%
Missing	n = 0	0.0%
Total	n = 54	100.0%

Table 30. Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 2.

Ethnic Group	Frequency	Percent
White	n = 50	92.6%
Hispanic / Latino	n = 1	1.8%
Black / African American	n = 0	0.0%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 3	5.6%
Missing	n = 0	0.0%
Total	n = 54	100.0%

Of the 54 applicants entering step 3 of the application process, an interview, 36 applicants did not advance onto step 4 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; and 18 applicants passed the interview and advanced onto step 4, the background check. Table 31 presents the frequency (n) and percent for gender. Of the 18 applicants who passed step 3, 16 (88.9%) were male and 2 (11.1%) were female. In this step, female applicants decreased 40% compared to 18.5% of females in the previous step. Table 32 presents the frequency (n) and percent for each of the five ethnic groups. Of the 18 applicants who passed step 3, 17 (94.4%) were White and 1 (5.6%) was an applicant of color. In this step, applicants of color decreased 24% compared to 7.4% of applicants of color in the previous step.



Table 31. Gender of 2011 Full-time Police Officer Applicants Passing Step 3.

Gender	Frequency	Percent
Male	n = 16	88.9%
Female	n = 2	11.1%
Missing	n = 0	0.0%
Total	n = 18	100.0%

Table 32. Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 3.

Ethnic Group	Frequency	Percent
White	n = 17	94.4%
Hispanic / Latino	n = 0	0.0%
Black / African American	n = 0	0.0%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 1	5.6%
Missing	n = 0	0.0%
Total	n = 18	100.0%

Of the 18 applicants entering step 4 of the application process, the background check, 9 applicants did not participate in this step, 3 applicants withdrew from the application process, and 6 applicants passed the background check and advanced onto step 5, the hiring step. Table 33 presents the frequency (n) and percent for gender. Of the 6 applicants who passed step 4, 5 (83.3%) were male and 1 (16.7%) was female. In this step, female applicants increased 50% compared to 11.1% of females in the previous step. Table 34 presents the frequency (n) and percent for each of the five ethnic groups. All 6 (100.0%) of the applicants who passed step 4 were White. In this step, all 5.6% of applicants of color in the last step were eliminated.

Table 33. Gender of 2011 Full-time Police Officer Applicants Passing Step 4.

Gender	Frequency	Percent
Male	n = 5	83.3%
Female	n = 1	16.7%
Missing	n = 0	0.0%
Total	n = 6	100.0%



Table 34. Ethnic Group of 2011 Full-time Police Officer Applicants Passing Step 4.

Ethnic Group	Frequency	Percent
White	n = 6	100.0%
Hispanic / Latino	n = 0	0.0%
Black / African American	n = 0	0.0%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 6	100.0%

Step 5 of the application process, the hiring step, included a second interview, a medical examination, a drug screen, and psychological evaluation. Of the 6 applicants entering the hiring step, 1 applicant did not get hired and remained inactive when the list expired. Of the 6 applicants who passed the second interview and were considered for a full-time police officer position, a conditional job offer was given, contingent upon passing the medical examination, drug screen, and psychological evaluation. A final job offer was given to the applicants who passed the medical examination, drug screen, and psychological evaluation. Five applicants were hired. Table 35 presents the frequency (n) and percent for gender. Of the 5 applicants hired, 4 (80.0%) were male and 1 (20.0%) was female. In this step, female applicants increased 20% compared to 16.7% in the previous step. In addition, the 20% of females hired was higher than the 10.7% of females who applied. Table 36 presents the frequency (n) and percent for each of the five ethnic groups. All 5 (100.0%) of the applicants hired were White. In addition, there were 16.5% applicants of color who applied, however, none were hired.

Table 35. Gender of 2011 Full-time Police Officer Applicants Hired – Step 5.

Gender	Frequency	Percent
Male	n = 4	80.0%
Female	n = 1	20.0%
Missing	n = 0	0.0%
Total	n = 5	100.0%



Table 36. Ethnic Group of 2011 Full-time Police Officer Applicants Hired – Step 5.

Ethnic Group	Frequency	Percent
White	n = 5	100.0%
Hispanic / Latino	n = 0	0.0%
Black / African American	n = 0	0.0%
Asian / Pacific Islander	n = 0	0.0%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 5	100.0%

2012 Posting for Full-time Police Officer External and Internal Applicants

As was the case in the 2009, 2010, and 2011 postings, minimum requirements in the 2012 job announcement were meeting the state regulatory board's eligibility requirements for a police officer, a valid driver's license with a good driving record, and first aid certification. Desired qualifications were the same as the 2011 posting, having a degree, prior police experience, and relevant volunteer experience.

The steps in the selection process for the 2012 posting included the following: step 1, minimum and desired qualifications review; step 2, a written test; step 3, an interview; and step 4, the hiring step, which included a second interview, a background investigation, a medical examination, a drug screen, and a psychological evaluation.

There were 511 applications received for the 2012 posting. Table 37 presents the frequency (n) for gender. Of the 511 applicants, 458 (89.6%) were male and 51 (10.0%) were female. Two applicants omitted gender. Table 38 presents the frequency (n) and percent for each of the five ethnic groups. Of the 511 applicants, 404 (79.0%) were White and 101 (19.8%) were applicants of color. Six applicants omitted ethnic group.



Table 37. Gender of 2012 Full-time Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 458	89.6%
Female	n = 51	10.0%
Missing	n = 2	0.4%
Total	n = 511	100.0%

Table 38. Ethnic Group of 2012 Full-time Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 404	79.0%
Hispanic / Latino	n = 20	3.9%
Black / African American	n = 51	10.0%
Asian / Pacific Islander	n = 23	4.5%
American Indian / Alaskan Native	n = 7	1.4%
Missing	n = 6	1.2%
Total	n = 511	100.0%

Of the 511 applicants entering step 1 of the application process, the minimum and desired qualifications review, 72 applicants did not advance onto step 2 after review of the supplemental questions for minimum and desired qualifications, and 439 applicants passed the qualifications review and advanced onto step 2, the written test. Table 39 presents the frequency (n) and percent for gender. Of the 439 applicants who passed step 1, 391 (89.1%) were male and 47 (10.7%) were female. One applicant omitted gender. In this step, female applicants increased 7% compared to 10% who applied. Table 40 presents the frequency (n) and percent for each of the five ethnic groups. Of the 439 applicants who passed step 1, 355 (80.9%) were White and 79 (18.0%) were applicants of color. Five applicants omitted ethnic group. In this step, applicants of color decreased 9% compared to 19.8% who applied.



Table 39. Gender of 2012 Full-time Police Officer Applicants Passing Step 1.

Gender	Frequency	Percent
Male	n = 391	89.1%
Female	n = 47	10.7%
Missing	n = 1	0.2%
Total	n = 439	100.0%

Table 40. Ethnic Group of 2012 Full-time Police Officer Applicants Passing Step 1.

Ethnic Group	Frequency	Percent
White	n = 355	80.9%
Hispanic / Latino	n = 16	3.6%
Black / African American	n = 35	8.0%
Asian / Pacific Islander	n = 21	4.8%
American Indian / Alaskan Native	n = 7	1.6%
Missing	n = 5	1.1%
Total	n = 439	100.0%

Of the 439 applicants entering step 2 of the application process, the written test, 193 did not pass this step, and 246 passed the written test and advanced onto step 3, an interview. Table 41 presents the frequency (n) and percent for gender. Of the 246 applicants who passed step 2, 216 (87.8%) were male and 29 (11.8%) were female. One applicant omitted gender. In this step, female applicants increased 10% compared to 10.7% of females in the previous step. Table 42 presents the frequency (n) and percent for each of the five ethnic groups. Of the 246 applicants who passed step 2, 198 (80.5%) were White and 46 (18.7%) were applicants of color. Two applicants omitted ethnic group. In this step, applicants of color decreased 6% compared to 19.8% of applicants of color in the previous step.

Table 41. Gender of 2012 Full-time Police Officer Applicants Passing Step 2.

Gender	Frequency	Percent
Male	n = 216	87.8%
Female	n = 29	11.8%
Missing	n = 1	0.4%
Total	n = 246	100.0%

Table 42. Ethnic Group of 2012 Full-time Police Officer Applicants Passing Step 2.

Ethnic Group	Frequency	Percent
White	n = 198	80.5%
Hispanic / Latino	n = 9	3.6%
Black / African American	n = 24	9.8%
Asian / Pacific Islander	n = 10	4.1%
American Indian / Alaskan Native	n = 3	1.2%
Missing	n = 2	0.8%
Total	n = 246	100.0%

Of the 246 applicants entering step 3 of the application process, an interview, 125 applicants did not advance onto step 3 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; and 121 passed the interview and advanced onto step 4, the hiring step. Table 43 presents the frequency (n) and percent for gender. Of the 121 applicants who passed step 3, 110 (90.9%) were male and 11 (9.1%) was a female. In this step, female applicants decreased 23% compared to 11.8% of females in the previous step. Table 44 presents the frequency (n) and percent for each of the five ethnic groups. Of the 121 applicants who passed step 3, 96 (79.3%) were White and 25 (20.7%) were applicants of color. In this step, applicants of color increased 11% compared to 18.7% of applicants of color in the previous step.

Table 43. Gender of 2012 Full-time Police Officer Applicants Passing Step 3.

Gender	Frequency	Percent
Male	n = 110	90.9%
Female	n = 11	9.1%
Missing	n = 0	0.0%
Total	n = 121	100.0%



Table 44. Ethnic Group of 2012 Full-time Police Officer Applicants Passing Step 3.

Ethnic Group	Frequency	Percent
White	n = 96	79.3%
Hispanic / Latino	n = 5	4.1%
Black / African American	n = 17	14.1%
Asian / Pacific Islander	n = 2	1.7%
American Indian / Alaskan Native	n = 1	0.8%
Missing	n = 0	0.0%
Total	n = 121	100.0%

Step 4 of the application process, the hiring step, included a second interview, a background investigation, a medical examination, a drug screen, and psychological evaluation. Of the 121 applicants entering the hiring step, 87 applicants were given the second interview but were not hired and remained inactive when the list expired; 4 applicants did not pass the background investigation, did not pass the medical examination, did not pass the drug screening, or did not pass the psychological evaluation; and 3 applicants withdrew from the application process. Of the applicants who passed the second interview and the background investigation and were considered for a full-time police officer position, a conditional job offer was given, contingent upon passing the medical examination, drug screen, and psychological evaluation. A final job offer was given to the applicants who passed the medical examination, drug screen, and psychological evaluation. Twenty-seven applicants were hired. Table 45 presents the frequency (n) and percent for gender. Of the 27 applicants hired, 23 (85.2%) were male and 4 (14.8%) were female. In this step, female applicants increased 63% compared to 9.1% of females in the previous step. In addition, the 14.8% of females hired was 48% higher than the 10% who applied. Table 46 presents the frequency (n) and percent for each of the five ethnic groups. Of the 27 applicants hired, 16 (59.3%) were White and 11 (40.7%) were applicants of color. In this step, applicants of color increased 97% compared to 20.7% of applicants in the previous step. In



addition, the 40.7% of applicants of color hired was a little more than twice the 19.8% who applied.

Table 45. Gender of 2012 Full-time Police Officer Applicants Hired – Step 4.

Gender	Frequency	Percent
Male	n = 23	85.2%
Female	n = 4	14.8%
Missing	n = 0	0.0%
Total	n = 27	100.0%

Table 46. Ethnic Group of 2012 Full-time Police Officer Applicants Hired – Step 4.

Ethnic Group	Frequency	Percent
White	n = 16	59.3%
Hispanic / Latino	n = 3	11.1%
Black / African American	n = 7	25.9%
Asian / Pacific Islander	n = 1	3.7%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 27	100.0%

2013 Posting for Full-time Police Officer External and Internal Applicants

As was the case in the 2009, 2010, 2011, and 2012 postings, minimum requirements in the 2013 job announcement were meeting the state regulatory board's eligibility requirements for a police officer and a valid driver's license with a good driving record. However, first aid certification was moved from a minimum qualification to a desired qualification. In addition to the same desired qualifications in the 2009, 2010, 2011 and 2012 postings, having a degree, prior police experience, and relevant volunteer experience; fluency in languages other than English was added as a desired qualification.

The steps in the selection process for the 2013 posting included the following: step 1, minimum and desired qualifications review; step 2, a written test; step 3, an interview; and step



4, the hiring step, which included a second interview, background investigation, a medical examination, a drug screen, and a psychological evaluation.

There were 640 applications received for the 2013 posting. Table 47 presents the frequency (n) and percent for gender. Of the 640 applicants, 565 (88.3%) were male and 74 (11.6%) were female. One applicant omitted gender. Table 48 presents the frequency (n) and percent for each of the five ethnic groups. Of the 640 applicants, 471 (73.6%) were White and 165 (25.8%) were applicants of color. Four applicants omitted ethnic group.

Table 47. Gender of 2013 Full-time Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 565	88.3%
Female	n = 74	11.6%
Missing	n = 1	.1%
Total	n = 640	100.0%

Table 48. Ethnic Group of 2013 Full-time Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 471	73.6%
Hispanic / Latino	n = 33	5.2%
Black / African American	n = 67	10.5%
Asian / Pacific Islander	n = 54	8.4%
American Indian / Alaskan Native	n = 11	1.7%
Missing	n = 4	0.6%
Total	n = 640	100.0%

Of the 640 applicants entering step 1 of the application process, minimum and desired qualifications review, 253 applicants did not advance onto step 2 after review of the supplemental questions for minimum and desired qualifications, and 387 applicants passed this step and advanced onto step 2, the written test. Table 49 presents the frequency (n) and percent for gender. Of the 387 applicants who passed step 1, 348 (89.9%) were male and 39 (10.1%) were female. In this step, female applicants decreased 13% compared to 11.6% who applied.



Table 50 presents the frequency (n) and percent for each of the five ethnic groups. Of the 387 applicants who passed step 1, 264 (68.2%) were White and 121 (31.3%) were applicants of color. Two applicants omitted ethnic group. In this step, applicants of color increased 21% compared to 25.8% who applied.

Table 49. Gender of 2013 Full-time Police Officer Applicants Passing Step 1.

Gender	Frequency	Percent
Male	n = 348	89.9%
Female	n = 39	10.1%
Missing	n = 0	0.0%
Total	n = 387	100.0%

Table 50. Ethnic Group of 2013 Full-time Police Officer Applicants Passing Step 1.

Ethnic Group	Frequency	Percent
White	n = 264	68.2%
Hispanic / Latino	n = 23	6.0%
Black / African American	n = 46	11.9%
Asian / Pacific Islander	n = 45	11.6%
American Indian / Alaskan Native	n = 7	1.8%
Missing	n = 2	0.5%
Total	n = 387	100.0%

Of the 387 applicants entering step 2 of the application process, the written test, 15 applicants did not pass this step, 190 applicants withdrew from the application process, and 182 passed the written test and advanced onto step 3, an interview. Table 51 presents the frequency (n) and percent for gender. Of the 182 applicants who passed step 2, 158 (86.8%) were male and 24 (13.2%) were female. In this step, female applicants increased 31% compared to 10.1% of females in the previous step. Table 52 presents the frequency (n) and percent for each of the five ethnic groups. Of the 182 applicants who passed step 2, 120 (66.0%) were White and 61 (33.4%) were applicants of color. One applicant omitted ethnic group. In this step, applicants of color increased 7% compared to 31.3% of applicants of color in the previous step.



Table 51. Gender of 2013 Full-time Police Officer Applicants Passing Step 2.

Gender	Frequency	Percent
Male	n = 158	86.8%
Female	n = 24	13.2%
Missing	n = 0	0.0%
Total	n = 182	100.0%

Table 52. Ethnic Group of 2013 Full-time Police Officer Applicants Passing Step 2.

Ethnic Group	Frequency	Percent
White	n = 120	66.0%
Hispanic / Latino	n = 12	6.6%
Black / African American	n = 17	9.3%
Asian / Pacific Islander	n = 27	14.8%
American Indian / Alaskan Native	n = 5	2.7%
Missing	n = 1	0.6%
Total	n = 182	100.0%

Of the 182 applicants going entering step 3 of the application process, an interview, 55 applicants did not advance onto step 4 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; 10 applicants withdrew from the application process; and 117 applicants passed the interview and advanced onto step 4, the hiring step. Table 53 presents the frequency (n) and percent for gender. Of the 117 applicants who passed step 3, 99 (84.6%) were male and 18 (15.4%) were female. In this step, female applicants increased 17% compared to 13.2% of females in the previous step. Table 54 presents the frequency (n) and percent for each of the five ethnic groups. Of the 117 applicants who passed step 3, 74 (63.2%) were White and 43 (36.8%) were applicants of color. In this step, applicants of color increased 10% compared to 33.4% of applicants of color in the previous step.



Table 53. Gender of 2013 Full-time Police Officer Applicants Passing Step 3.

Gender	Frequency	Percent
Male	n = 99	84.6%
Female	n = 18	15.4%
Missing	n = 0	0.0%
Total	n = 117	100.0%

Table 54. Ethnic Group of 2013 Full-time Police Officer Applicants Passing Step 3.

Ethnic Group	Frequency	Percent
White	n = 74	63.2%
Hispanic / Latino	n = 10	8.5%
Black / African American	n = 12	10.3%
Asian / Pacific Islander	n = 18	15.4%
American Indian / Alaskan Native	n = 3	2.6%
Missing	n = 0	0.0%
Total	n = 117	100.0%

Step 4 of the application process, the hiring step, included a second interview, a background investigation, a medical examination, a drug screen, and a psychological exam. Of the 117 applicants entering the hiring step, 100 applicants were given the second interview but were not hired and remained inactive when the list expired, 2 applicants failed the background check, 1 applicant withdrew from the application process, and 1 applicant declined the job offer. For the applicants who passed the second interview and the background investigation and were considered for a full-time police officer position, a conditional job offer was given, contingent upon passing the medical examination, drug screen, and psychological evaluation. A final job offer was given to the applicants who passed the medical examination, drug screen, and psychological evaluation. Thirteen applicants were hired. Table 55 presents the frequency (n) and percent for gender. Of the 13 applicants hired, 10 (76.9%) were male and 3 (23.1%) were female. In this step, female applicants increased 50% compared to 15.4% of females in the previous step. In addition, the 23.1% of females hired was more than twice the 11.6% who



applied. Table 56 presents the frequency (n) and percent for each of the five ethnic groups. Of the 13 applicants hired, 7 (53.8%) were White and 6 (46.2%) were applicants of color. Applicants of color increased 26% compared to 36.8% of applicants of color in the previous step. In addition, the 46.2% applicants of color hired was almost twice that of the 25.8% who applied.

Table 55. Gender of 2013 Full-time Police Officer Applicants Hired – Step 4.

Gender	Frequency	Percent
Male	n = 10	76.9%
Female	n = 3	23.1%
Missing	n = 0	0.0%
Total	n = 13	100.0%

Table 56. Ethnic Group of 2013 Full-time Police Officer Applicants Hired – Step 4.

Ethnic Group	Frequency	Percent
White	n = 7	53.8%
Hispanic / Latino	n = 1	7.7%
Black / African American	n = 3	23.1%
Asian / Pacific Islander	n = 1	7.7%
American Indian / Alaskan Native	n = 1	7.7%
Missing	n = 0	0.0%
Total	n = 13	100.0%

2014 Posting for Full-time Police Officer External and Internal Applicants

As was the case in the 2013 posting, minimum requirements in the 2014 job announcement were meeting the state regulatory board's eligibility requirements for a police officer and a valid driver's license with an acceptable driving record. In addition to the same desired qualifications from the 2013 posting, having first aid certification, a degree, prior police experience, relevant volunteer experience, and fluency in languages other than English; prior experience in non-paid non-sworn police related positions, prior security experience, and experience working with diverse populations were added as desired qualifications.



The steps in the selection process for the 2014 posting included the following: step 1, minimum and desired qualifications review; step 2, an interview; step 3, a second interview; and step 4, the hiring step, which included a background investigation, a medical examination, a drug screen, and a psychological evaluation.

There were 516 applications received for the 2014 posting. Table 57 presents the frequency (n) and percent for gender. Of the 516 applicants, 432 (83.7%) were male and 81 (15.7%) were female. Three applicants omitted gender. Table 58 presents the frequency (n) and percent for each of the five ethnic groups and missing cases. Of the 516 applicants, 375 (72.7%) were White and 133 (25.8%) were applicants of color. Eight applicants omitted ethnic group.

Table 57. Gender of 2014 Full-time Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 432	83.7%
Female	n = 81	15.7%
Missing	n = 3	0.6%
Total	n = 516	100.0%

Table 58. Ethnic Group of 2014 Full-time Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 375	72.7%
Hispanic / Latino	n = 39	7.6%
Black / African American	n = 50	9.7%
Asian / Pacific Islander	n = 40	7.8%
American Indian / Alaskan Native	n = 4	0.7%
Missing	n = 8	1.5%
Total	n = 516	100.0%

Of the 516 applicants entering step 1 of the application process, the minimum and desired qualifications review, 27 applicants did not meet the minimum qualifications, 297 applicants met the minimum qualifications but did not advance onto step 2 after review of the supplemental questions for desired qualifications, and 192 applicants passed the qualifications review and



advanced onto step 2, an interview. Table 59 presents the frequency (n) and percent for gender. Of the 192 applicants who passed step 1, 164 (85.4%) were male and 27 (14.1%) were female. One applicant omitted gender. In this step, female applicants decreased 10% compared to 15.7% who applied. Table 60 presents the frequency (n) and percent for each of the five ethnic groups. Of the 192 applicants who passed step 1, 109 (56.8%) were White and 79 (41.2%) were applicants of color. Four applicants omitted ethnic group. In this step, applicants of color increased 60% compared to 25.8% who applied.

Table 59. Gender of 2014 Full-time Police Officer Applicants Passing Step 1.

Gender	Frequency	Percent
Male	n = 164	85.4%
Female	n = 27	14.1%
Missing	n = 1	0.5%
Total	n = 192	100.0%

Table 60. Ethnic Group of 2014 Full-time Police Officer Applicants Passing Step 1.

Ethnic Group	Frequency	Percent
White	n = 109	56.8%
Hispanic / Latino	n = 23	12.0%
Black / African American	n = 23	12.0%
Asian / Pacific Islander	n = 31	16.2%
American Indian / Alaskan Native	n = 2	1.0%
Missing	n = 4	2.0%
Total	n = 192	100.0%

Of the 192 applicants entering step 2 of the application process, an interview, 107 applicants did not advance onto step 3 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; 2 applicants withdrew from the application process, and 83 applicants passed the interview and advanced onto step 3, the second interview. Table 61 presents the frequency (n) and percent for gender. Of the 83 applicants who passed step 2, 70 (84.3%) were male and 13 (15.7%) were female. In this step,



female applicants increased 11% compared to 14.1% of females in the previous step. Table 62 presents the frequency (n) and percent for each of the five ethnic groups. Of the 83 applicants who passed step 2, 48 (57.8%) were White and 34 (41.0%) were applicants of color. One applicant omitted ethnic group. In this step, applicants of color were about the same percentage compared to 41.2% of applicants in the previous step.

Table 61. Gender of 2014 Full-time Police Officer Applicants Passing Step 2.

Gender	Frequency	Percent
Male	n = 70	84.3%
Female	n = 13	15.7%
Missing	n = 0	0.0%
Total	n = 83	100.0%

Table 62. Ethnic Group of 2014 Full-time Police Officer Applicants Passing Step 2.

Ethnic Group	Frequency	Percent
White	n = 48	57.8%
Hispanic / Latino	n = 5	6.0%
Black / African American	n = 11	13.3%
Asian / Pacific Islander	n = 16	19.3%
American Indian / Alaskan Native	n = 2	2.4%
Missing	n = 1	1.2%
Total	n = 83	100.0%

Of the 83 applicants entering step 3 of the application process, a second interview, 33 applicants did not advance onto step 4 because they did not respond to the agency contacting them, did not show up for the second interview, or did not pass the second interview; 1 applicant withdrew from the application process, and 49 passed the second interview and advanced onto step 4, the hiring step. Table 63 presents the frequency (n) and percent for gender. Of the 49 applicants who passed step 3, 40 (81.6%) were male and 9 (18.4%) were female. In this step, female applicants increased 17% compared to 15.7% of females in the previous step. Table 64 presents the frequency (n) and percent for each of the five ethnic groups. Of the 49 applicants



who passed step 3, 23 (46.9%) were White and 25 (51.0%) were applicants of color. One applicant omitted ethnic group. In this step, applicants of color increased 24% compared to 41.0% of applicants of color in the previous step.

Table 63. Gender of 2014 Full-time Police Officer Applicants Passing Step 3.

Gender	Frequency	Percent
Male	n = 40	81.6%
Female	n = 9	18.4%
Missing	n = 0	0.0%
Total	n = 49	100.0%

Table 64. Ethnic Group of 2014 Full-time Police Officer Applicants Passing Step 3.

Ethnic Group	Frequency	Percent
White	n = 23	46.9%
Hispanic / Latino	n = 3	6.1%
Black / African American	n = 7	14.3%
Asian / Pacific Islander	n = 13	26.5%
American Indian / Alaskan Native	n = 2	4.1%
Missing	n = 1	2.1%
Total	n = 49	100.0%

Step 4 of the application process, the hiring step, included a background investigation, a medical examination, a drug screen, and psychological evaluation. Of the 49 applicants entering the hiring step, 24 applicants were not hired and remained inactive when the list expired, 4 applicants withdrew from the application process, and 1 applicant declined the job offer. Of the applicants who passed the background investigation and were considered for a full-time police officer position, a conditional job offer was given, contingent upon passing the medical examination, drug screen, and psychological evaluation. A final job offer was given to the applicants who passed the medical examination, drug screen, and psychological evaluation, and 20 applicants were hired. Table 65 presents the frequency (n) and percent for gender. Of the 20 applicants hired, 16 (80.0%) were male and 4 (20.0%) were female. In this step, female



applicants increased 9% compared to 18.4% of females in the previous step. In addition, the 20% of females hired was higher than the 15.7% who applied. Table 66 presents the frequency (n) and percent for each of the five ethnic groups. Of the 20 applicants hired, 9 (45.0%) were White and 11 (55.0%) were applicants of color. In this step, applicants of color increased 8% compared to 51% of applicants of color in the previous step. In addition, the 55% of applicants of color hired was more than twice than the 25.8% who applied.

Table 65. Gender of 2014 Full-time Police Officer Applicants Hired – Step 4.

Gender	Frequency	Percent
Male	n = 16	80.0%
Female	n = 4	20.0%
Missing	n = 0	0.0%
Total	n = 20	100.0%

Table 66. Ethnic Group of 2014 Full-time Police Officer Applicants Hired – Step 4.

Ethnic Group	Frequency	Percent
White	n = 9	45.0%
Hispanic / Latino	n = 2	10.0%
Black / African American	n = 4	20.0%
Asian / Pacific Islander	n = 3	15.0%
American Indian / Alaskan Native	n = 2	10.0%
Missing	n = 0	0.0%
Total	n = 20	100.0%

2016 Posting for Full-time Police Officer External and Internal Applicants

As was the case in the 2014 and 2015 posting, minimum qualifications in the 2016 job announcement were meeting the state regulatory board's eligibility requirements for a police officer and a valid driver's license with an acceptable driving record. The desired qualifications were the same as in the 2014 posting, having first aid certification, a degree, prior police experience, prior experience in non-paid non-sworn police related positions, fluency in languages other than English, and experience working with diverse populations.



The steps in the selection process for the 2016 posting included the following: step 1, minimum and desired qualifications review; step 2, an interview; step 3, a second interview; step 4, a background investigation; and step 5, the hiring step, which included a medical examination, a drug screen, and a psychological evaluation.

There were 317 applications received for the 2016 posting. Table 67 presents the frequency (n) and percent for gender. Of the 317 applicants, 268 (84.6%) were male and 46 (14.5%) were female. Three applicants omitted gender. Table 68 presents the frequency (n) and percent for each of the five ethnic groups. Of the 317 applicants, 200 (63.1%) were White and 108 (34.1%) were applicants of color. Nine applicants omitted ethnic group.

Table 67. Gender of 2016 Full-time Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 268	84.6%
Female	n = 46	14.5%
Missing	n = 3	0.9%
Total	n = 317	100.0%

Table 68. Ethnic Group of 2016 Full-time Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 200	63.1%
Hispanic / Latino	n = 18	5.7%
Black / African American	n = 45	14.2%
Asian / Pacific Islander	n = 40	12.6%
American Indian / Alaskan Native	n = 5	1.6%
Missing	n = 9	2.8%
Total	n = 317	100.0%

Of the 317 applicants entering step 1 of the application process, the minimum and desired qualifications review, 27 applicants did not meet the minimum qualifications, 184 did not advance onto step 2 after review of the supplemental questions for desired qualifications, and 106 applicants passed this step and advanced onto step 2, an interview. Table 69 presents the



frequency (n) and percent for gender. Of the 106 applicants who passed step 1, 95 (89.6%) were male and 11 (10.4%) were female. In this step, female applicants decreased 28% compared to 14.5% who applied. Table 70 presents the frequency (n) and percent for each of the five ethnic groups. Of the 106 applicants who passed step 1, 49 (46.2%) were White and 55 (51.9%) were applicants of color. Two applicants omitted ethnic group. In this step, applicants of color increased 52% compared to 34.1% who applied.

Table 69. Gender of 2016 Full-time Police Officer Applicants Passing Step 1.

Gender	Frequency	Percent
Male	n = 95	89.6%
Female	n = 11	10.4%
Missing	n = 0	0.0%
Total	n = 106	100.0%

Table 70. Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 1.

Ethnic Group	Frequency	Percent
White	n = 49	46.2%
Hispanic / Latino	n = 7	6.6%
Black / African American	n = 19	17.9%
Asian / Pacific Islander	n = 28	26.4%
American Indian / Alaskan Native	n = 1	1.0%
Missing	n = 2	1.9%
Total	n = 106	100.0%

Of the 106 applicants entering step 2 of the application process, an interview, 61 applicants did not advance onto step 3 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; and 45 applicants passed the interview and advanced onto step 3, the second interview. Table 71 presents the frequency (n) and percent for gender. Of the 45 applicants who passed step 2, 40 (88.9%) were male and 5 (11.1%) were female. In this step, female applicants increased 7% compared to 10.4% of females in the previous step. Table 72 presents the frequency (n) and percent for each of the five



ethnic groups. Of the 45 applicants who passed step 2, 20 (44.4%) were White and 24 (53.4%) were applicants of color. One applicant omitted ethnic group. Applicants of color increased 3% compared to 51.9% of applicants of color in the previous step.

Table 71. Gender of 2016 Full-time Police Officer Applicants Passing Step 2.

Gender	Frequency	Percent
Male	n = 40	88.9%
Female	n = 5	11.1%
Missing	n = 0	0.0%
Total	n = 45	100.0%

Table 72. Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 2.

Ethnic Group	Frequency	Percent
White	n = 20	44.4%
Hispanic / Latino	n = 2	4.5%
Black / African American	n = 10	22.2%
Asian / Pacific Islander	n = 11	24.5%
American Indian / Alaskan Native	n = 1	2.2%
Missing	n = 1	2.2%
Total	n = 45	100.0%

Of the 45 applicants entering step 3 of the application process, the second interview, 17 did not advance onto step 4 because they did not respond to the agency contacting them, did not show up for the interview, or did not pass the interview; 2 applicants withdrew from the application process; and 26 passed the second interview and advanced onto step 4, the hiring step. Table 73 presents the frequency (n) and percent for gender. Of the 26 applicants who passed step 3, 24 (92.3%) were male and 2 (7.7%) were female. In this step, female applicants decreased 31% compared to 11.1% of females in the previous step. Table 74 presents the frequency (n) and percent for each of the five ethnic groups. Of the 26 applicants who passed step 3, 10 (38.6%) were White and 16 (61.4%) were applicants of color. In this step, applicants of color increased 15% compared to 53.4% of applicants of color in the previous step.



Table 73. Gender of 2016 Full-time Police Officer Applicants Passing Step 3.

Gender	Frequency	Percent
Male	n = 24	92.3%
Female	n = 2	7.7%
Missing	n = 0	0.0%
Total	n = 26	100.0%

Table 74. Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 3.

Ethnic Group	Frequency	Percent
White	n = 10	38.6%
Hispanic / Latino	n = 1	3.8%
Black / African American	n = 7	26.9%
Asian / Pacific Islander	n = 7	26.9%
American Indian / Alaskan Native	n = 1	3.8%
Missing	n = 0	0.0%
Total	n = 26	100.0%

Of the 26 applicants entering step 4 of the application process, the background investigation, 4 applicants did not pass the background investigation, 2 applicants withdrew from the application process, and 20 passed the background investigation and advanced onto step 5, the hiring step. Table 75 presents the frequency (n) and percent for gender. Of the 20 applicants who passed step 4, 18 (90.0%) were male and 2 (10.0%) were female. In this step, female applicants increased 30% compared to 7.7% of females in the previous step. Table 76 presents the frequency (n) and percent for each of the five ethnic groups. Of the 20 applicants who passed step 4, 7 (35.0%) were White and 13 (65.0%) were applicants of color. In this step, applicants of color increased 6% compared to 61.4% of applicants of color in the previous step. Table 75. Gender of 2016 Full-time Police Officer Applicants Passing Step 4.

 Gender
 Frequency
 Percent

 Male
 n = 18 90.0%

 Female
 n = 2 10.0%

 Missing
 n = 0 0.0%

 Total
 n = 20 100.0%



Table 76. Ethnic Group of 2016 Full-time Police Officer Applicants Passing Step 4.

Ethnic Group	Frequency	Percent
White	n = 7	35.0%
Hispanic / Latino	n = 1	5.0%
Black / African American	n = 5	25.0%
Asian / Pacific Islander	n = 7	35.0%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 20	100.0%

Step 5 of the application process, the hiring step, included a medical examination, a drug screen, and psychological evaluation. Of the 20 applicants entering the hiring step, 2 applicants were not hired and remained inactive when the list expired; 2 applicants did not pass the medical examination, did not pass the drug screen, or did not pass the psychological evaluation; and 1 applicant withdrew from the application process. For the applicants who passed the previous steps and were considered for a full-time police officer position, a conditional job offer was given, contingent upon passing the medical examination, drug screen, and psychological evaluation. A final job offer was given to the applicants who passed the medical examination, drug screen, and psychological evaluation. Fifteen applicants were hired. Table 77 presents the frequency (n) and percent for gender. Of the 15 applicants hired, 13 (86.7%) were male and 2 (13.3%) were female. In this step, female applicants increased 33% compared to 10% of females in the previous step. In addition, the 13.3% of females hired was less than the 14.5% who applied. Table 78 presents the frequency (n) and percent for each of the five ethnic groups. Of the 15 applicants hired, 6 (40.0%) were White and 9 (60.0%) were applicants of color. In this step, applicants of color decreased 5% compared to the 65% of applicants of color in the previous step. In addition, the 60% of applicants of color hired was 76% higher than the 34.1% who applied.



Table 77. Gender of 2016 Full-time Police Officer Applicants Hired – Step 5.

Gender	Frequency	Percent
Male	n = 13	86.7%
Female	n = 2	13.3%
Missing	n = 0	0.0%
Total	n = 15	100.0%

Table 78. Ethnic Group of 2016 Full-time Police Officer Applicants Hired – Step 5.

Ethnic Group	Frequency	Percent
White	n = 6	40.0%
Hispanic / Latino	n = 0	0.0%
Black / African American	n = 4	26.7%
Asian / Pacific Islander	n = 5	33.3%
American Indian / Alaskan Native	n = 0	0.0%
Missing	n = 0	0.0%
Total	n = 15	100.0%

Summary of Applicants Hired from Postings Open for External and Internal Applicants

Table 79 summarizes the frequencies (n) and percentages of female applicants and applicants of color hired for full-time police officer from the 2009 through 2016 postings open to external and internal applicants presented in the previous tables. The highest number and percentage of females hired was in the 2009 posting, with 5 and 33.3%. The lowest number and percentage of females hired was in the 2010 posting, with none hired. The highest numbers of applicants of color hired, were in both the 2012 and 2014 postings, with 11 hired. The highest percentage of applicants of color hired was in the 2016 posting, with 60.0%. The lowest number and percentage of applicants of color hired was in the 2011 posting, with none hired.



Table 79. Number and Percent of Female and Applicants of Color Hired for Full-time Police Officer from 2009 to 2016 Postings Open to External and Internal Applicants.

Year	2009	2010	2011	2012	2013	2014	2015	2016
Female	n= 5	n= 0	n= 1	n= 4	n= 3	n= 4	N/A	n= 2
Applicants	33.3 %	0.0 %	20.0 %	14.8 %	23.1 %	20.0 %		13.3 %
Applicants of	n= 2	n= 1	n= 0	n=11	n=6	n=11	N/A	n= 9
Color	13.4 %	20.0 %	0.0 %	40.7 %	46.2 %	55.0 %		60.0 %

N/A – There was not a posting open to all applicants in 2015.

Table 80 presents the applicants of color by the frequency (n) and percentage of applicants hired for each of the five ethnic groups for full-time police officer from the 2009 to 2016 postings for external and internal applicants. At least 1 Hispanic / Latino was hired in five of the seven years, at least 1 Black / African American was hired in five of the seven years, at least 1 Asian / Pacific Islander was hired in four of the seven years, and at least 1 American Indian / Alaskan Native was hired in two of the seven years. The highest number of the four non-White groups was for Blacks / African Americans, with 7 in the 2012 posting. The highest percentage of the four non-White groups was for Asians / Pacific Islanders, with 33.3% in the 2016 posting.

Table 80. Number and Percent of Full-time Police Officers Hired from 2009 to 2016 Postings Open to External and Internal Applicants by Ethnic Group.

Year	2009	2010	2011	2012	2013	2014	2015	2016
White	n = 11	n = 4	n = 5	n = 16	n = 7	n = 9	N/A	n = 6
	73.3 %	80.0 %	100 %	59.3 %	53.8 %	45.0 %		40.0 %
Hispanic / Latino	n = 1	n = 1		n = 3	n = 1	n = 2	N/A	
	6.7 %	20.0 %		11.1 %	7.7 %	10.0 %		
Black / African	n = 1			n = 7	n = 3	n = 4	N/A	n = 4
American	6.7 %			25.9 %	23.1 %	20.0 %		26.7 %
Asian / Pacific				n = 1	n = 1	n = 3	N/A	n = 5
Islander				3.7 %	7.7 %	15.0 %		33.3 %
American Indian					n = 1	n = 2	N/A	
Alaskan Native					7.7 %	10.0 %		
Missing	n = 2						N/A	
	13.3%							
Total Hired	n = 15	n = 5	n = 5	n = 27	n = 13	n = 20	N/A	n = 15
	100 %	100 %	100 %	100 %	100 %	100 %		100 %

N/A – There was not a posting open to all applicants in 2015.



2011 to 2015 Postings for Full-time Police Officer Open Only for Internal Applicants

There were six postings from 2011 to 2015 for the position of full-time police officer open only to internal employees. Minimum qualifications for the six postings included meeting the state regulatory board's eligibility requirements for a police officer, a valid driver's license with a good or an acceptable driving record, and current employment by the agency in a different position. Desired qualifications varied for the six postings. All six postings included prior police experience. Five postings included a degree and relevant volunteer experience. One posting included prior experience in non-paid non-sworn police related positions, previous security experience, fluency in languages other than English, and experience working with diverse populations. Three postings had first aid certification as a minimum qualification and three postings had it as a desired qualification.

The steps in the application process for all six internal only postings included a minimum and desired qualifications review, an interview, a background check, a medical examination, a drug screen, and a psychological evaluation. Three of the six internal only postings had a written test and three postings did not have a written test. There was only one posting that had the second interview after the background investigation. The other five postings had the second interview before the background investigation.

All six postings for the internal only postings are grouped together in reporting the frequencies and percentages by gender and ethnic group due to the low number of applicants for each posting. There were 39 applications received for all six internal postings. Table 81 presents the frequency (n) and percent for gender. Of the 39 applicants, 34 (87.2%) were male and 5 (12.8%) were female. Table 82 presents the frequency (n) and percent for each of the five



ethnic groups. Of the 39 applicants, 24 (61.5%) were White and 15 (38.5%) were applicants of color.

Table 81. Gender of 2011 to 2015 Full-time Internal Police Officer Applicants.

Gender	Frequency	Percent
Male	n = 34	87.2 %
Female	n = 5	12.8 %
Missing	n = 0	0.0 %
Total	n = 39	100.0 %

Table 82. Ethnic Group of 2011 to 2015 Full-time Internal Police Officer Applicants.

Ethnic Group	Frequency	Percent
White	n = 24	61.5 %
Hispanic / Latino	n = 1	2.6 %
Black / African American	n = 5	12.8 %
Asian / Pacific Islander	n = 8	20.5 %
American Indian / Alaskan Native	n = 1	2.6 %
Missing	n = 0	0.0 %
Total	n = 39	100.0 %

There were 14 applicants hired from the six postings for internal employees only. Table 83 presents the frequency (n) and percent for gender. Of the 14 applicants hired, 10 (71.4%) were male and 4 (28.6%) were female. Females who were hired, 28.6%, were more than two times higher than females who applied, 12.8%. Table 84 presents the frequency (n) and percent for each of the five ethnic groups. Of the 14 applicants hired, 12 (85.7%) were White and 2 (14.3%) were applicants of color. Applicants who were hired, 14.3%, were 37% less than applicants of color who applied, 38.5%.

Table 83. Gender of 2011 to 2015 Full-time Internal Police Officer Applicants Hired.

Gender	Frequency	Percent
Male	n = 10	71.4 %
Female	n = 4	28.6 %
Missing	n = 0	0.0 %
Total	n = 14	100.0 %



Table 84. Ethnic Group of 2011 to 2015 Full-time Internal Police Officer Applicants Hired.

Ethnic Group	Frequency	Percent
White	n = 12	85.7 %
Hispanic / Latino	n = 0	0.0 %
Black / African American	n = 1	7.15 %
Asian / Pacific Islander	n = 1	7.15 %
American Indian / Alaskan Native	n = 0	0.0 %
Missing	n = 0	0.0 %
Total	n = 14	100.0 %

Comparison of Applicants Hired from Postings Open for External and Internal Applicants to Postings Open Only for Internal Applicants

Table 85 compares the percentages of female applicants and applicants of color hired for postings open for external and internal applicants, to the postings open for only internal applicants. For the postings open for all applicants, 19 (19.0%) of the applicants hired were female, and for the postings open only for internal employees, 4 (28.6%) of the applicants hired were female. The percentage of females hired from the postings open only for internal applicants was 1.5 times higher than the percentage of females hired from the postings open for both external and internal applicants. Of the postings open to all applicants, 40 (40.0%) of the applicants hired were applicants of color, and of the postings open only to internal employees, 2 (14.3%) of the applicants hired were applicants of color. The percentage of applicants of color hired from postings open only for internal applicants was almost one-third lower than the percentage of applicants of color hired from postings open for both external and internal applicants.



Table 85. Number and Percent of Female Applicants and Applicants of Color Hired for Full-time Police Officer by Type of Posting (Open to all Applicants and Internal Only Applicants).

Type of Posting	Posting Open to All Applicants	Posting Open to
	(External and Internal Applicants)	Internal Applicants Only
	from 2009 to 2016	from 2011 to 2015
Female	n = 19	n = 4
Applicants	19.0 %	28.6 %
Applicants of Color	n = 40	n = 2
	40.0 %	14.3 %

Full-time Police Officer Separation Data

There are three broad categories of separation used by the case agency: *Resignation*, *Termination*, and *Retirement*. These three categories parallel to the three broad categories cited in Chapter II: Voluntary Separation, Involuntary Separation, and Retirement. Resignations are divided into two reasons: *General Resignation* and *Resignation – Other Position*. The definition of a general resignation includes an officer who left the case agency or law enforcement for personal reasons, or who left for a position not in law enforcement. Resignation – other position is defined as an officer who left the case agency for another law enforcement agency.

Termination is divided into three reasons: Failed Probation, Medically Disqualified, and Misconduct. All full-time officers are given a probationary period from their date of hire as a full-time officer. The one-year probationary period includes academy training, field training, and the officer assigned to work a patrol area. The agency's academy is currently 10 weeks of training that provides officers with a solid base to perform their jobs. The classroom and skills training include firearms, use of force, scenarios, active shooter, cultural awareness, community outreach, and agency expectations. Once an officer passes their academy training, they go through four phases of field training, concluding with being shadowed by a field training officer. The field training program takes about 14 weeks. Any of the phases in the officer's field training



program may be extended as deemed necessary. After passing the field training program, the officer is assigned to work a patrol area. Failed probation is defined as an officer who is terminated before their probationary period has expired. This includes an officer not passing the academy; an officer passing the academy, but not successfully completing the field training; or an officer passing both the academy and the field training but is terminated during the time the officer is on their own before their probation period expires.

Medically disqualified is defined as an officer who can no longer perform the job of police officer and who does not qualify for a pension disability. Misconduct is defined as an officer who is terminated due to a violation of department policy. Retirement is divided into two reasons: *Retirement – Normal* and *Retirement – Disability*. Normal retirement is defined as an officer who retires under their pension. Disability retirement is defined as an officer who has a disability that qualifies under the pension.

As discussed in chapter II, Review of Literature, Wareham et al. (2015) calculated the turnover rates by dividing the number of full-time sworn officers for each turnover indicator by the total number of full-time sworn officers, and then multiplying that number by 100 to get turnover rates in percentages (p. 350). The four turnover indicators used by Wareham et al. are voluntary turnover (general resignations and resignations – other law enforcement position) retirement (non-medical retirements), involuntary turnover (medical/disability retirements, probationary rejections, dismissals, and medical disabilities not covered by their pension), and total turnover rate (the three previous categories added together).

The agency turnover rates were calculated using the same method as Wareham et al. (2015) used in their study. Table 86 presents the number of full-time officers who separated from the agency and the turnover rates for each year that separation data was available for the



entire year, from 2010 through 2016. Officers who separated in 2017 are not included in the agency turnover rates because the 2017 data only includes officers who separated from the agency through May of 2017. The lowest turnover rate is 1.54% for 2010, and the highest turnover rate is 12.50% for 2011.

Table 86. Number of Full-time Police Officer Separations by Turnover Indicator and Total Turnover Rates for each Year from 2010 to 2016.

Turnover Indicator	2010	2011	2012	2013	2014	2015	2016
Resignations	0	2	1	2	5	6	5
Retirements	0	5	3	2	3	0	0
Involuntary Separations	1	1	0	2	1	7	3
Total Number of Officers	1	8	4	6	9	13	8
Total Turnover Rate	1.54%	12.50%	6.06%	7.06%	10.00%	11.93%	7.14%

The average turnover rate for 49 officers who separated from 2010 through 2016 is 8.29%. Table 87 presents the total number of separations and average turnover rates, broken down into the four categories of turnover indicators used by Wareham et al.'s (2015) study. The turnover rate for the 21 resignations is 3.55% and includes the case agency's general resignations and resignations for another law enforcement position. The turnover rate for the 13 retirements is 2.20% and includes non-medical retirements. The turnover rate for the 15 involuntary separations is 2.54% and includes medical/disability retirements, dismissals due to misconduct, probationary rejections, and medical/disability terminations not covered by their pension. The turnover rates presented in Tables 86 and 87 are compared to Wareham et al.'s results in chapter VI.



Table 87. Number and Average Turnover Rate for the Total Number of Officers who Separated from 2010 to 2016 by Turnover Indicator.

Turnover Indicator	Frequency	Turnover Rate
Resignations	21	3.55%
Retirements	13	2.20%
Involuntary Separations	15	2.54%
Total Turnover	49	8.29%

The remaining separation data includes the 53 full-time officers for all ranks who separated from the case agency from January 2010 through May of 2017. Table 88 presents the officers who separated under the three broad categories of separation. Of the 53 officers who separated from the agency, 23 (43.4%) resigned (voluntary separation), 14 (26.4%) were terminated (involuntary separation), and 16 (30.2%) retired. Tables 88 also presents the breakdown of the specific reasons under each of the broad categories of separation. Of the 23 officers who resigned voluntarily, 4 (7.5%) were for general resignations, which means they resigned from the agency or law enforcement for personal reasons or left for a position not in law enforcement. The remaining 19 officers (35.9%) left for another law enforcement position. Of the 14 officers who were terminated involuntarily, 8 (15.0%) failed probation, 3 (5.7%) were medically disqualified, and 3 (5.7%) were dismissed for misconduct. Of the 8 officers who failed probation, 6 officers did not pass their FTO training and 2 officers passed their FTO training but were terminated before their probation expired due to performance or conduct issues. Of the 16 who retired, 13 (24.5%) left with normal retirement and 3 (5.7%) left with a disability retirement.



Table 88. Full-time Police Officers who Separated from 2010 to mid-2017 by Reason of Separation and Type of Separation.

Type of Separation	Resignations	Terminations	Retirements	Total	Total
	(Voluntary	(Involuntary		Number	Percent
	Separation)	Separation)		Separated	Separated
General Resignation	n= 4			n= 4	7.5%
Resignation -	n=19			n=19	35.9%
Other Position					
Failed Probation		n= 8		n= 8	15.0%
Medically		n= 3		n= 3	5.7%
Disqualified					
Misconduct		n= 3		n= 3	5.7%
Retirement - Normal			n=13	n=13	24.5%
Retirement - Disability			n= 3	n= 3	5.7%
Total Separated	n=23	n=14	n=16	n=53	n=53
	43.4%	26.4%	30.2%	100.0%	100%

Table 89 presents the agency data and the six categories that Reaves used in his 2012 BJS Report of the 2008 CSLLEA data. The highest percentage of separations was for general resignations and resignations for another law enforcement position, with 43.4%. The second highest percentage of separations was for normal retirements, with 24.5%. The third highest percentage of separations was for termination for failed probation, with 15.0%. Last are termination for misconduct, termination for a medical / disability not covered by retirement, and a disability retirement, with 5.7% for each of these three categories. The agency percentages for categories of separation are compared to Reaves' (2012) results in chapter VI.

Table 89. Percent of Reason of Separation from 2010 to mid-2017 using Reaves' (2012) and Case Agency Categories.

Reave's (2012) Categories in 2008 CSLLEA data	Agency Categories for 2010 to 2017 data	Percent of Total Turnover
Resignation	General Resignation and Resignation - Other Position	43.4%
Non-medical Retirement	Retirements - Normal	24.5%
Dismissal	Termination - Misconduct	5.7%



Table 89. cont.

Reave's (2012) Categories in	Agency Categories for 2010 to	Percent of Total Turnover
2008 CSLLEA data	2017 data	
Probation Rejection	Termination - Failed Probation	15.0%
Medical/Disability	Retirement - Disability	5.7%
Retirement		
Other Separation	Termination - Medical /	5.7%
	Disability not covered by	
	Pension	
Total of All Separations	Total of All Separations	100.0%

Table 90 presents the officers who separated from 2010 to 2017 by reason of separation, with a breakdown for gender and ethnic group for the 53 officers. Of the 4 officers who left with a general resignation, 1 was male and 3 (75.0%) were female, and 3 officers were White and 1 (25.0%) was an officer of color. Of the 19 officers who left for another law enforcement position, 18 were male and 1 (5.3%) was female, and 15 were White and 4 (21.1%) were officers of color. Of the 8 officers who were terminated because they failed probation, 6 were male and 2 (25.0%) were female, and 4 were White and 4 (50.0%) were officers of color. Of the 3 officers who were terminated because of a medical disqualification, all 3 were male and all 3 were White. Of the 3 officers who were terminated for misconduct, all 3 were male and all 3 were officers of color. Of the 13 officers who retired under a normal retirement, all 13 were male, and 11 were White and 2 (15.4%) were officers of color. Of the 3 officers who retired under a disability retirement, all 3 were male, and 1 was White and 2 (66.7%) were officers of color.



Table 90. Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Reason of Separation, Gender, and Ethnic Group.

Type of Separation	Total	Gender	Ethnic Group	
	Separated		1	
General Resignation	n= 4	1 Male	3 White	
_	7.5%	3 Female	1 Asian / Pacific Islander	
Resignation - Other Position	n=19	18 Male	15 White	
	35.9%	1 Female	1 Hispanic / Latino	
			1 Black / African American	
			1 Asian / Pacific Islander	
			1 American Indian / Alaskan Native	
Failed Probation	n=8	6 Male	4 White	
	15.0%	2 Female	3 Black / African American	
			1 Hispanic / Latino	
Medically Disqualified	n=3	3 Male	3 White	
	5.7%			
Misconduct	n=3	3 Male	1 Hispanic / Latino	
	5.7%		1 Black / African American	
			1 American Indian / Alaskan Native	
Retirement - Normal	n=13	13 Male	11 White	
	24.5%		1 Hispanic / Latino	
			1 American Indian / Alaskan Native	
Retirement - Disability	n=3	3 Male	1 White	
	5.7%		1 Hispanic / Latino	
			1 Black / African American	
Total	53	53	53	
	100.0%			

Table 91 presents the number and percentage of full-time officers who separated with 5 or less years of service and more than 5 years of service. There were 32 (60.4%) officers with 5 or less years of service and 21 (39.6%) officers with more than 5 years of service. Table 92 presents the number and percentage of full-time officers who separated from the agency by years of service. Almost one-third of the officers, 17 (32.1%), had less than 1 year of service, 10 (18.9%) of the officers had 1-3 years of service, 5 (9.4%) of the officers had 3-5 years of service, and 21 (39.6%) officers had more than 5 years of service.



Table 91. Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Five or Less Years of Service and More Than 5 Years of Service.

Years of Service	Frequency	Percent
5 or less years	32	60.4%
More than 5 years	21	39.6%

Table 92. Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Years of Service.

Years of Service	Frequency	Percent
Less than 1 year	17	32.1%
1-3 years	10	18.9%
3-5 years	5	9.4%
More than 5 years	21	39.6%
Total	53	100.0%

Table 93 presents the years of service before separation, with a breakdown of the number for gender and ethnic group for the 53 officers. Of the 17 officers who separated with less than 1 year of service, 13 were male and 4 (23.5%) were female, and 10 were White and 7 (41.2%) were officers of color. Of the 10 officers who separated with 1-3 years of service, all 10 were male, and 6 were White and 4 (40.0%) were officers of color. Of the 5 officers who separated with 3-5 years of service, 4 were male and 1 (20.0%) was female, and all 5 were White. Of the 21 officers who separated with more than 5 years of service, 20 were male and 1 (4.8%) was female, and 16 were White and 5 (23.8%) were officers of color.

Table 93. Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Years of Service, Gender, and Ethnic Group.

Years of Service	Total Number	Gender	Ethnic Group	
Less than 1 year	n=17	13 Male	10 White	
	32.1%	4 Female	1 Hispanic / Latino	
			3 Black / African American	
			2 Asian / Pacific Islander	
			1 American Indian / Alaskan Native	



Table 93, cont.

Years of Service	Total Number	Gender	Ethnic Group
1-3 years	n=10	10 Male	6 White
-	18.9%		1 Hispanic / Latino
			2 Black / African American
			1 American Indian / Alaskan Native
3-5 years	n= 5	4 Male	5 White
	9.4%	1 Female	
More than 5 years	n=21	20 Male	16 White
	39.6%	1 Female	3 Hispanic / Latino
			1 Black / African American
			1 American Indian / Alaskan Native
Total	n=53	n=53	n=53

Table 94 presents the number of officers who left the agency by years of service and the three broad types of separation: Resignation (Voluntary separation), Termination (Involuntary separation), and Retirement. Of the 17 officers who separated with less than 1 year of service, 7 (41.2%) resigned and 10 (58.8%) were terminated. Of the 10 officers who separated with 1-3 years of service, 8 (80.0%) resigned and 2 (20.0%) were terminated. Of the 5 officers who separated with 3-5 years of service, all 5 (100.0%) resigned. For the 21 officers who separated with more than 5 years of service, 3 (14.3%) resigned, 2 (9.5%) were terminated, and 16 (76.2%) retired.

Table 94. Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Years of Service and Type of Separation.

Years of Service	Resignations	Terminations	Retirements	Total Separated
Less than 1 year	7	10	0	17
1-3 years	8	2	0	10
3-5 years	5	0	0	5
More than 5 years	3	2	16	21
Total	23	14	16	53

Table 95 presents the 53 officers who left the agency by the specific reason of separation and years of service. Of the 17 officers who separated with less than 1 year of service, 3 (17.6%)



resigned under a general resignation, 4 (23.5%) left for another law enforcement agency, 8 (47.1%) failed probation, and 2 (11.8%) were medically disqualified. Of the 10 officers who separated with 1-3 years of service, 8 (80.0%) left for another law enforcement position, and 2 (20.0%) were terminated because of misconduct. Of the 5 officers who separated with 3-5 years of service, all 5 resigned for another law enforcement position. Of the 21 officers who separated with more than 5 years of service, 1 (4.75%) resigned under a general resignation, 2 (9.5%) left for another law enforcement agency, 1 (4.75%) was medically disqualified, 1 (4.75%) was terminated because of misconduct, 13 (62.0%) retired under a normal retirement, and 3 (14.25%) retired on a disability retirement.

Table 95. Full-time Police Officers who Separated from the Agency from 2010 to mid-2017 by Reason of Separation and Years of Service.

Type of Separation	<1 year	1-3 years	3-5 Years	>5 years	Total
General Resignation	3	0	0	1	4
Resignation - Other Position	4	8	5	2	19
Failed Probation	8	0	0	0	8
Medically Disqualified	2	0	0	1	3
Misconduct	0	2	0	1	3
Retirement - Normal	0	0	0	13	13
Retirement - Disability	0	0	0	3	3
Total	17	10	5	21	53

Conclusion

This chapter presented the data analysis for the composition, applicant, and separation data. It is evident from the results of the applicant and composition data presented that representation of females and officers of color has increased both in frequencies and percentages for those who applied and those who were hired. This chapter also presented the results from the separation data, including turnover rates, reasons for separation, length of service before separation, and gender and race/ethnicity of officers who separated. The results of the agency's



separation data are compared to Wareham et al.'s (2015) turnover rates and Reaves' (2012) separation rates in Chapter VI. Chapter V discusses the analysis of the interviews. The last chapter, Chapter VI, presents the discussions and conclusions for the four data sets.



CHAPTER V

COLLECTION AND ANALYSIS OF QUALITATIVE INTERVIEW DATA

This chapter presents the findings from the semi-structured interviews in this mixed methods case study. It discusses recruiting, hiring, and retention from the perspectives of seven agency personnel responsible for the implementation and oversight of recruiting, hiring, and retention practices at the case agency. Agency administration identified key personnel to be interviewed. This chapter identifies and discusses themes in the responses from the seven respondents interviewed.

Description of Interview Data

In conjunction with the results from the three data sets presented in Chapter IV, the interview data results are used to answer research sub-questions #1-4, previously presented in Chapter I and III.

Sub-question #4

What practices are used by the agency for recruiting, hiring, and retaining diversity?

Sub-question #5

What factors are important for recruiting, hiring, and retaining diversity within the agency?

Sub-question #6

What characteristics are important in leadership for recruiting, hiring, and retaining diversity within the agency?



Sub-question #7

What modifications in the traditional recruiting, selection, and hiring processes are needed to recruit and hire for a diverse police agency?

This researcher used the interview guide in Appendix A to obtain the respondent's own words to describe the processes of recruiting, hiring, and retention. Additional questions were asked to clarify responses, check for understanding, encourage subjects to elaborate on their responses, follow up on responses, or explore an unexpected topic.

This researcher started each interview with the following introduction:

The purpose of this research is to describe how your agency recruits, hires, and retains diversity for your full-time police officers. I would like to ask you about this and your role in hiring, recruiting, and retention processes. The first part of the interview will ask you questions about recruiting and hiring and the second part of the interview will ask you questions about retention.

Prior literature on representative bureaucracy presented in Chapter II was used to construct the interview questions. There was a total of nine main questions and three sub-questions on recruiting and hiring, five questions on separation and retention, and one closing question (See Appendix A).

Process of Identifying Themes from Analysis of the Interview Data

It is evident from the interview data that this agency prioritizes being reflective of its communities and engaging in community-oriented and problem-oriented policing. Respondents in the interviews indicated that the increases in the representation of female officer and officers of color within their agency were a result of a combination of different factors. These factors include the Human Resources Office having a formal Affirmative Action plan and being an



Equal Opportunity Employer; the command staff setting hiring goals, targeting its recruiting efforts, and implementing changes in the application process; and the chief taking an active role in recruiting and hiring practices. At least five of the seven respondents needed to have discussed an element of a category or theme in their responses in order to be considered a theme. After analyzing the data, nine themes were found to meet the criteria. As indicated below, one theme was in response to an interview question directly asking about the theme, four themes were influenced by the interview questions, and four themes emerged from the interview data.

The following theme was in response to an interview question asking directly about the theme and was also present in response to other interview questions:

- The importance of the agency being reflective of its communities

 The following four themes were influenced by the interview questions:
- The active role of the chief in diversifying the agency
- The impact of external factors on the applicant pool
- The influence of organizational factors on recruiting, hiring, and retention
- The positive influence of professional organizations and associations on increasing recruiting, hiring, and retention of female officers and officers of color

The following four themes emerged from the interview data:

- The impact of changes in the hiring process on increasing diversity
- The influence of administrative discretion on decisions made regarding recruiting and hiring practices
- The relationship between the agency being responsive to its communities through a commitment to community policing and recruiting, hiring, and retaining diversity



• The perception that some diverse applicants hired were not qualified

Five of the seven respondents discussed one theme, the perception that some diverse applicants were not qualified, and all seven respondents discussed at least one element of the other eight themes. Elements included both positive and negative perceptions. Some of these themes were also represented in more than one question. Each theme and associated question(s) from the interview are discussed below. Some of these themes are related to other themes which are also noted below.

Themes for Recruiting, Hiring, and Retention

The importance of the agency being reflective of its communities. This theme, the importance of the agency being reflective of its communities, was in response to an interview question directly asking about this theme and also appeared in response to several other interview questions. In response to the question, "Do you think it is important for your agency to reflect community characteristics? Why or why not?", all respondents stated it is important for their agency to reflect female officers and officers of color. In response to why it is important to be representative of the communities they serve, two of the responses illustrated the impact of incidents involving the police and communities of color, "I don't think you cannot be in this day and age", and "of course, look at all the trends now, the agencies that are having critical incidents that aren't reflective of their community obviously are paying more of a price for that".

Two respondents raised the concern that it is not possible to send an officer of color to all calls involving a member of his/her community. Another respondent also brought this to attention, stating that people "who have this mindset miss the point because if a white officer answers the call from a person of color, that person of color knows that their community is represented within their agency". This respondent added that one reason why recruiting for

diversity is important is that "we get more credibility from that community because it shows we are serious and committed to hiring officers from their community" and "that gets all of us credibility when we have to have the hard conversations". Another respondent stated that "if protecting the community is our fundamental measure as cops", this cannot be accomplished if officers are not culturally and demographically reflective of the people they serve. This respondent also added that they have had great success across many levels of diversity which reinforces that they are more effective in delivering police services. Another respondent affirmed their success by stating that the focus on recruiting has resulted in success in reflecting their community.

All respondents stated that there are benefits of increased support, trust, and credibility from the communities as a result of officers getting out into the community. Respondents stated that this not only earns trust and cooperation, but also educates the public on the agency's role and function. One respondent elaborated "it's a way to build community trust when you're reflective of who you serve, they go hand in hand", and added that "over the course of the next five, 10, and 15 years, that diversity is going to continue up through the ranks". Another respondent stated "... to be able to police the community you live in, you have to understand, and having a diverse department helps with that, because you're learning, I may not be from this community, but there may be somebody that I work with who is and can help explain traditions or cultural things that I don't understand".

The importance of the agency being reflective of its communities was also evident in the responses to "Has your agency identified goals in achieving diversity?", "What are these goals?", and "What is your agency doing to achieve these goals?" One respondent discussed the difficulty in what measure to use as a goal for the agency to be more reflective of the



community, stating that it is difficult to achieve a reflection of the population, which is 50% for women, because all agencies are having difficulty recruiting women. Instead, one goal for the agency set by the command staff is 50% from underrepresented populations. In addition to underrepresented groups including officers of color, women are also included because women are underrepresented in law enforcement. Another respondent added that another goal they aim for is to hire the same percentage of female officers and applicants of color that applied. Another respondent stated that their Human Resources Office sets hiring goals that includes examining "the demographic make-up and they tell you whether or not you have an identified need" and further explained that it is based on the demographic make-up of the available hiring pool and available work force of surrounding areas, which is a different standard than trying to represent percentages in the community. Another respondent added that "I don't think we've gone far enough. I think we're taking baby steps to try and figure this out. We've had some success, we've had more success than many other agencies have had, we're at least double what most other agencies are in terms of underrepresented populations".

In response to the question "What groups do you target in recruiting?", all respondents stated that their agency targets recruitment for specific groups. Examples provided included Native Americans, Blacks and African Americans, Asians, Latinos and Hispanics, females, and people who are fluent in languages other than English. Most of the respondents stated that they need to make a concentrated effort to recruit more females. One respondent stated that "one of the ways to recruit female officers is to promote female officers, so that people outside the department see that there are opportunities here". Another respondent stated that they have recruited for females in different ethnic groups and added that they reach out to diverse



communities and women, so prospective applicants understand what law enforcement officers do.

The recruiting brochures were also changed. One respondent described how the first brochure showed a white male on the front and later brochures included different races/ethnicities and both males and females. This respondent added that "changing the visuals of what to hand out in terms of what our recruiting literature looks like" sends a message that we want to hire the best people for the job.

The active role of the chief in diversifying the agency. This theme, the active role the chief takes in diversifying the agency, particularly in recruiting and hiring, was identified by all respondents to the different questions in the interview. When asked one question, "Identify people or positions that are important in terms of recruiting and hiring diversity among your agency's full-time officers." and "Why are they important?", all seven respondents identified the chief as important or the most important person or position. Five respondents indicated that the reason their agency is diverse is because of the chief. One respondent stated that the chief has allocated resources to the recruitment team to go to events and travel, marketing, and allows flexibility in adjusting work schedules for officers to attend recruitment fairs and events. The same respondent stated that the "chief is huge and instrumental in making recruiting more engaging" and that the chief "had a major impact in increasing their diversity from about 5% to well over 35%". Three other respondents indicated a similar increase in the composition of their agency due to the chief. Other examples include responses from four other respondents stating that hiring diversity is a priority with the chief; that the chief "sets the tone for recruiting and hiring and this chief is very active in the hiring process"; that the chief is "most important in setting our diversity goals"; and "the chief does not believe in the hiring process to be delegated



down the ladder". As stated in the previous theme, the command staff set the goal of achieving 50% of underrepresented populations in hiring and agency composition. Three other respondents discussed that the percentage of female applicants and applicants of color have increased in the past few years and believed they have met the goal of 50% female applicants and applicants of color hired in the past few years.

Other persons or positions identified as important are the command staff; the Human Resources Office; the hiring managers, which includes an administrator who is in charge of the hiring process, liaisons with the Human Resources Office, and coordinates the steps in the application process, the academy training and field training; and other individuals who are involved in recruiting, the interview panel, and background investigations. Three respondents added that the people responsible for recruiting and hiring reflect the guidance and direction from the command staff.

This theme also became evident in the seven respondents' description of their agency's recruiting and hiring process, goals in diversifying their agency, what their agency is doing to achieve these goals, the steps in their application process, and organizational factors that impact recruitment, hiring, and retention. One respondent summarized this as "the chief has made diversity a priority here, and I think the chief has accomplished it". In addition, all respondents stated that the changes in the application process occurred after the change in command staff in 2012, and since that time, the chief has been instrumental in implementing these changes in the selection process.

The impact of changes in the hiring process on increasing diversity. This theme, the impact of changes in the hiring process on increasing diversity, emerged from the interview data and was mentioned by all respondents to at least one of the interview questions. All respondents



stated that the changes in the application process have resulted in increasing the numbers of female officers and officers of color within their agency. One respondent added that they have not gone below 50% of underrepresented applicants (female applicants and applicants of color) hired for the job postings since these changes were made. In response to the question, "What are the steps in your selection process?", the steps and changes in the application process identified and described by all seven respondents are consistent with the steps in the application data presented in chapter IV.

The changes in the application process are presented in the order they appear in the application process. The first change noted by three respondents was adding to the list of desired qualifications. Examples provided include fluency in languages other than English and experience working with diverse communities.

All respondents noted in their interviews that the written test was eliminated from the application process. Three respondents stated that they found the written test eliminated a good number of applicants of color. One respondent added "Did the people that passed first on the written test perform better than the people that passed 10th on the written test? Nobody can show any correlation between actual performance and where you ranked on the written test".

This respondent expounded on why the written test was eliminated from the application process,

The written test does not tell us much about your skill set, but we spend our day talking to people. We initially made it pass/fail. 80% of the diversity that was in the initial application process was eliminated during the written test. We then moved them onto the next test which was a much better test where you're going to find good police performance, and that's the oral panel ... if you can talk to a panel of strangers, then you probably can talk to a group of strangers.



Another respondent stated

I think it is better now than in the past ... because of [eliminating] the written test, I don't believe we were getting the candidates we were looking for because there were certain groups of people that didn't score as high, so the decision was made to stop with the written test, and just do a larger pool of the first interview, now we're able to get a better pool of candidates who we're looking for.

Four respondents agreed that the interview was a better predictor of police performance than the written test. Three respondents expressed concerns that even though the written test is not the best predictor of police performance, the written test assessed cognitive abilities and the aptitude to learn, and they did not know of a different test to assess these.

After the written test was eliminated, applicants proceed from the qualifications review to the first interview, where they are scored pass/fail instead of ranked by score, as was previously done. Another change discussed by the respondents included diversifying the interview panel because of the perception that diverse candidates were lost in the first interview. This is illustrated by one respondent stating that they were told that panel members had to be subject matter experts and replied that "the subject matter they're experts on is being in the community". This respondent added that they are "still working on this but we now have people of color and broader diversity on the community side and officer side too".

All respondents discussed three changes made to the background process after 2012, and one respondent described the backgrounds as an interesting challenge. Five respondents indicated that every agency is competing for good candidates and applicants accept the first job offer. One respondent explained "sometimes it's a matter of who can get through the hiring process the fastest to make the job offer first". One respondent stated that "historically, we



would do the background and then if someone were to get a pass on the background, then they'd do a final interview". The first change made regarding the background was moving the second interview from after the background investigation to before the background investigation.

Several respondents noted that switching the background and second interview is different than what they have done and what other agencies remain doing. This reduced the number of background investigations, which also reduced the time it took to make a conditional offer to candidates. Although moving the second interview to before the background investigation added more interviews with the command staff, one respondent noted "tell me what else we were doing that was more important than hiring, if we hire the right officers, then our job becomes easier". This respondent added that if they hire a problem, then administration "ends up with Internal Affairs, discipline, and a grievance, arbitration and termination hearings. It's a pay me now or pay me after thing" and "we want to spend the time on the front end and then pay it forward".

Also, another factor discussed by all respondents is that the list to determine who advances on from the second interview to the background can change. Terms used to describe this change were "the list starts over" and the "scores are reset at 0". One respondent stated that the reason why the list can change is "because what the interview panel looks for is will this person likely be a good cop in a very general sense" and "what the second interview looks for is will this person be a good cop here ... they're different sets of criteria". This respondent also identified the five questions asked in the second interview, which include the applicants background growing up and interacting with people, volunteer experience, their definition of community policing and how they have applied it or will apply it at this agency, an example where they had to make an ethical decision, and their experience working with diverse communities. After the second interview, the command staff and the hiring manager then

determine who to background, then after backgrounds, they review the list to determine who to make conditional offers of employment.

The second change in the background process was having officers within the agency conduct the background investigations. Previously, backgrounds were conducted from retired officers from outside the agency, which one respondent described as "thorough, but eliminated candidates who were recruited". Three respondents also stated that this provided opportunities to help retain officers.

The process before the third change was that the backgrounder could eliminate applicants in the background process. This was changed to making recommendations, and if a recommendation is made not to hire, the backgrounder is required to state why not. This change allowed the command staff and hiring manager to make the decision whether or not to hire an applicant instead of the backgrounder making that decision. Several respondents supported this change, adding that if people make a mistake and grow from that mistake, they should not be automatically eliminated. Only one respondent specifically mentioned the psychological evaluation, stating it was problematic because the specific profile of an ideal cop does not include "refugee populations" and that refugees are one group the agency targets in recruiting.

The influence of administrative discretion on decisions made regarding recruiting and hiring practices. It is important to understand what factors contributed to the chief taking an active role in recruiting and hiring practices and in implementing changes in the application process that resulted in the increases in female officers and officers of color. How the chief obtains and uses this discretion to increase their agency's diversity is a key variable. This theme, the influence of administrative discretion on decisions made regarding recruiting and hiring practices, emerged from the responses of all respondents to various questions. These include the



responses to the question on the human resources structure, the descriptions of the steps and changes to the steps in the application process, identifying the important people and positions important in recruiting, hiring, and retaining a diverse agency, identifying goals in achieving diversity, and what the agency is doing to achieve these goals.

Besides the minimum selection standards for police officers set by the state regulatory board, all respondents discussed that their Human Resources Office sets the standards, rules, and regulations for hiring practices at the agency and that the chief makes decisions regarding recruiting and hiring. A significant factor that emerged from the interview data was that their agency is not under civil service rules and regulations and this strongly contributed to the chief making changes in the application process that resulted in the increase of female officers and officers of color. In response to the question, "What type of human resource structure or system do you have for your recruiting and hiring? Please describe this structure", all respondents stated that their Human Resources Office was not similar to most other police agencies, especially for most large agencies. Five respondents specifically stated that their Human Resource Office was not under civil service rules and regulations, one respondent stated public service, and one respondent stated it was not like a traditional police human resources office. Three respondents clarified the influence of civil service versus their structure on hiring diverse applicants, stating that with civil service, there are rankings and an eligibility list created where candidates are selected in the order they are ranked which makes it harder to hire female officers and officers of color. Respondents described that instead of having civil service which uses a ranked order, their agency establishes a pool of candidates, so candidates do not have to be taken in order and this allows for more flexibility in hiring female officers and officers of color.



Five respondents indicated that the structure of the Human Resources Office gives the chief discretion, authority, and flexibility in making decisions on hiring practices, specifically, in implementing changes in the application process that resulted in hiring more female officers and officers of color. One respondent stated that their Human Resources Office "allowed us to do things that are a little bit outside the norm as long as it doesn't put any particular candidate or class of candidates at a disadvantage" and "as long as we can articulate the reasons why we do things, why this candidate scored higher. For example, they speak a second or third language, we are on solid ground, and our agency doesn't challenge us on that". One respondent also stated that "they'll allow us to filter those applications when you are trying to reduce the number of applicants, so we have some say in those supplemental questions". Their Human Resources Office has a formal Affirmative Action plan and is an Equal Opportunity Employer. In addition, one respondent stated that their Human Resources Office would be more active in recruiting, but that their agency "adopted that role internally to be more active in the recruiting piece".

Another factor that provided administrative discretion in making decisions regarding hiring practices was identified by four respondents who stated that their union contract influences promotional practices, however, their union does not influence hiring practices, thus the chief directly influences the steps in the application process.

The relationship between the agency being responsive to its communities through a commitment to community policing and recruiting, hiring, and retaining diversity. All respondents indicated that it is important for their agency to be responsive to its communities through community policing. This theme, the relationship between being responsive to its communities through a commitment to community policing and recruiting, hiring, and retaining diversity, emerged in the responses of five respondents. Interview data indicated that there was a



reciprocal relationship between community policing and recruiting, hiring, and retaining female officers, officers of color, and officers who are fluent in a language other than English. It was evident from four of the respondents that community engagement has helped their recruiting efforts, stating that with this chief "recruitment has been a big push, ... for us to be able to go out to more places and recruit more", "we do a lot of recruiting ... going to community events", "we go out and visit with the community, so they can see us, know what we're about, and in itself, that's even a recruitment tactic, even though we don't have our boards and signs up", and that "one of the big things is the ability to be able to participate in that stuff off and on duty". One respondent added that the chief is "huge on community engagement and being out there in the community" which has helped increase the diversity within their department. Another respondent stated that the "chief is really active in all these communities" and "that's been huge for us and that isn't just about recruiting, that's about making sure that we're responsible to the communities that we serve. But it does help with the recruiting piece too". The same respondent provided the example of where a Latino officer helped the agency develop a recruiting brochure in Spanish and stated that this "shows you're serious about reaching out to their community" and that this is "one tiny thing to show that we've taken one little extra step to show you we want you to be part of our organization". One respondent stated that there is a necessity to be active in the community due to a "lack of cultural and linguistic diversity".

The influence of community policing on recruiting and hiring was also evident in the descriptions provided by respondents for the desired qualities, skills, and traits that applicants should possess, which are also consistent with the traits and skills necessary for officers to engage in community-oriented and problem-oriented policing. In response to the questions, "What qualities, skills, abilities, and traits are you looking for in the people who apply for full-



time police officer at your agency?", all seven respondents indicated that personal character and traits were as important or more important than technical knowledge and skills, such as laws, firearms, defensive tactics, because they can teach specific skills in the academy training. This was illustrated by the traits and skills respondents stated that applicants should possess to successfully perform as police officers at their agency: honesty, integrity, good character, being innovative, having a community service mindset, being a good communicator, having a clean background, problem solving, moral courage, physical courage, ability to stand up to stress, willingness to learn, and being trainable and coachable. Three other respondents added, "anyone can become a police officer, and get trained, but it takes a passion, a calling, to be able to be in a field of service and protect and serve", "we can take smart people and people with integrity and train the skills to be a good cop but it's hard to train integrity, character, and community service mindset", and we want people "who truly want to help out people that are out there to make somebody else's day better". It was interesting that one respondent pointed out that community policing increases recruitment and retention because their agency is doing much more community engagement than they use to and this attracts and retains officers who want to engage in community policing.

All the respondents stated communication skills were important or most important in skills that their officers should possess. Respondents provided examples to illustrate this, including the necessity for applicants to effectively speak to other people in general, calm people who are upset or suicidal, read body language, speak languages other than English, and "communicate on a variety of different levels, face-to-face, good body language, but also have high levels of linguistics and cultural competency". This also illustrates the connection between



the skills necessary to perform community-oriented and problem-oriented policing and recruiting and hiring officers with diverse backgrounds.

In addition to the influence of community policing on recruitment and hiring diverse officers, recruitment also influences community policing. Two respondents illustrate how recruiting efforts increase community policing by educating communities about their role and function by stating "getting out there ... and educating people overall who we are as an agency and what we do, community engagement" and "recruitment is sometimes more than just getting officers, but it's also educating the public in what we do and how we function... so you have to realize that recruitment is a lot more than just hiring and firing and retaining people". One respondent provided an example where recruiting efforts also help to build relationships with their communities, stating that the networking and relationships that they have built by attending schools, job fairs, and community events are important for others to see officers in a positive light.

The impact of external factors on the applicant pool. This theme, the impact of external factors on the applicant pool, was influenced by the question, "What factors impact your agency's recruitment and hiring practices achieving diversity?". All respondents discussed at least two events or factors external to the agency that they perceived as responsible for the decrease in the number of applicants for their agency and for law enforcement agencies in general, including the negative portrayal of police by the news, social media, and protest groups, such as Black Lives Matter. One respondent stated that high profile cases have "a very tough effect in our ability to recruit". Two other respondents stated that the negative perception of law enforcement has hindered the number of applicants.



Five respondents illustrated how the applicant pool for police officers is down for their agency and elsewhere in the United States. One respondent stated that the reduction in the applicant pool has "gotten worse in the last year". One of these respondents added that "the hiring pool is not as deep as it was just a few years ago" and another respondent added that for all agencies, the quality of candidates has decreased. For female applicants, one respondent stated that it was "tough to recruit women everywhere". Another respondent commented on the different generation of people going into law enforcement as contributing to fewer applicants.

As stated previously, respondents identified that the time it takes to get applicants through the application process can result in applicants taking the first offer. One respondent described this as the most important factor in recruiting candidates by stating "we get them into the background phase, we then realize that they're being backgrounded at multiple agencies at the same time, so it becomes a factor that they'll take the first job that they get offered," and if your process is not quick enough you will lose them to another agency. It was interesting that this respondent perceived this to be a social factor rather than an organizational or internal factor, because other agencies that also wish to be reflective of their communities are competing with their agency.

Discussed next are external factors that influence separation and retention. In response to the two questions, "For the full-time officers that have separated from your agency, do you know the reasons why they have left? Please tell me about these reasons.", all respondents perceived that there are a variety of reasons officers have left their agency. All respondents stated that officers have left for more opportunities at other agencies, for example, more opportunities for promotion or work in specialty units. One respondent was "sometimes baffled by the reason why people leave" and gave the example where officers who stated they are leaving because



there is not a lot of opportunity here but left for agencies with less opportunity and less pay. This respondent added that "some people are just moving because they can, I think it's a bit generational". Other reasons provided by respondents included for family or personal reasons, for example, to be closer to home and deciding they do not want to do this job after being hired. One factor that negatively affects officer retention was identified by several respondents stating that some of their officers were recruited by other agencies, with two respondents using the term "poaching" to describe what they perceived to be other agencies taking their officers.

The influence of organizational factors on recruiting, hiring, and retention.

Elements of the theme, the influence of organization factors on recruiting, hiring, and retention, was influenced by the interview questions and was discussed by all respondents throughout the interviews. All respondents stated in their interviews that their agency has great pay, benefits, overtime, and pension and these were factors for increasing recruiting, hiring, and retention. Mentors within the agency were mentioned by two respondents, with one respondent stating, "mentors in the organization become a part of what keeps people here" and another respondent stating that is important to retain officers because they are needed to mentor other officers.

In response to the question, "What else do you think your agency can do to attract diverse qualified candidates?", one respondent suggested signing bonuses that some other agencies have. Another respondent suggested having a full-time person committed to coordinating recruiting and community engagement activities. One respondent expounded further, "There is still stuff we could do, such as a Facebook page so millennials know about our jobs".

Another organizational factor, their union, was discussed in the interviews. In response to the question, "Does your union support or hinder recruiting and hiring female officers and officers of color?", all seven respondents stated that their union supports recruiting and hiring



female officers and officers of color. Respondents indicated that their union has worked hard for good pay, benefits, and contracts to attract candidates and that their union has a good working relationship with management.

In response to the questions, "Do female or officers of color leave at different rates than White males?", all seven respondents stated that they had a fairly low turnover rate overall and that mostly White males have left, and not many officers of color or females have left. One respondent offered reasons why women might leave including shift work, daycare, or are intimidated because of their size. The responses that few women and officers leave are consistent with the results of the separation data presented in chapter IV.

In response to the question, "What does your agency do to retain your full-time officers?", even though all respondents stated that officers leave for more opportunities in other agencies, most of the respondents stated there are opportunities here that help in retention.

Examples included additional assignments or what one respondent described as "collateral duties" to their assigned position, for example, instructor positions, career enrichment opportunities, sending officers to different types of training, offering more overtime, offering some flexibility on work schedules when feasible, and making women feel welcome. One respondent added that "some people do not realize what a privilege it is to work here".

In response to the question, "What else do you think your agency can do to retain officers?", suggestions included offering ride-a-longs for applicants so they understand what the job entails, give officers more opportunities for different or additional assignments, and help with child care issues. One respondent stated they can offer more opportunities only to a point because they cannot pull officers from core assignments. One respondent emphasized the need to do more with mentorship programs.



The perception that some diverse applicants hired were not qualified. The theme, the perception that some diverse applicants hired were not qualified, emerged from the interview data. Respondents indicated that it is important for management to communicate with officers what the hiring process consists of and the rationale behind the hiring process. Four respondents stated that there is a perception that some of the diverse candidates do not meet minimum selection standards or qualifications. In response to the question. "Has there been any challenges to the selection criteria?", respondents stated although there were very few formal complaints, they have heard concerns and perceptions from other officers. These included some people get more chances in passing the academy training and field training, administration can pick whoever they want, that people get hired because of their race/ethnicity, that some applicants got hired here that could not get hired at other agencies, that the same people are on the interview panel, that people not recommended by the background investigator were hired, that the standards were lowered, and there is lack of communication regarding who is hired and why.

One respondent stated that "we also need to make sure that we continue to maintain a standard" and provided two examples. The first example acknowledged that there is a benefit of someone who can speak other languages, however, officers need to be able to speak, read, and write English because others have to be able to understand what they are saying on the radio, and secondly, reports that are poorly written can impact the credibility of the officer and the agency. Five respondents stated officers have left their agency because they were not happy with the processes for hiring, promotion, and/or selection for special positions/assignments. Two respondents explained that they have lost officers who don't want to work with other officers who lack a competency level.



It is not known where these perceptions originate because all applicants have to meet minimum selection standards to become a police officer set by the state regulatory board.

Recruiting and hiring diverse officers increases the hiring pool and is not in opposition to recruiting and hiring competent and qualified applicants. Chapter VI discusses how the role and function of police has changed and expanded and how these changes impact hiring requirements.

The positive influence of professional organizations and associations on increasing recruiting, hiring, and retention of female officers and officers of color. The theme, the positive influence of professional organizations and associations on increasing, hiring, and retention of female officers and officers of color resulted from the response to the questions "What professional organizations have affected your agency in recruiting and hiring diversity? Please describe how." All seven respondents stated that there were positive outcomes from several organizations and associations, including the different racial/ethnic police organizations and associations which one respondent termed "affinity organizations" and organizations and associations for women police and chief law enforcement officers. The following are examples cited by respondents of how these organizations have helped their recruiting: the agency is promoted at the various conferences; the agency sends their officers to the various conferences; applicants have stated in their interview that a particular association told them to apply and/or helped them prepare for the interview; officers who belong to associations tell other members to apply to their agency; and openings at their agency are posted in the various publications of the different organizations. One respondent stated "there's two reasons we're really diverse. Number one are the affinity organizations that have been sending us great recruits, and secondly, and probably even more important than that are the cops that we have here will go back and tell their friends this is a good place to work". Respondents cited examples where some of the



officers of color within their agency recruited others from their community to apply. One respondent added "the cops become your best tool for recruiting".

Conclusion

This chapter discussed the results of the interview data organized by nine themes. The last chapter, Chapter VI, discusses the findings and conclusions for the analysis of the four sets presented in Chapter IV and this chapter, the implications of this study, the contributions of this study to the literature, and compares the results to previous literature presented in Chapter II. Discussed last are the limitations of this study and recommendations for future research.



CHAPTER VI

DISCUSSION AND CONCLUSIONS

The purpose of this study is to describe how one police agency in the Midwest recruits, hires, and retains a representative agency and to describe what factors were important in this process. It is a descriptive, single case study that uses triangulation of multiple methods, both quantitative and qualitative, and multiple data sources to describe how the agency recruits, hires, and retains diversity. The quantitative methods include composition data from 2009 through 2017, application data from 2009 through 2016, and separation data from 2010 through mid-2017. The one qualitative method, interviews of personnel responsible for the implementation and oversight of recruiting, hiring, and retention, provided detailed descriptions from the perspectives of those involved in these practices. These detailed descriptions contribute to our understanding of how this agency increased its representation of female officers, officers of color, and officers who are fluent in languages other than English.

Summary of the Findings

The three quantitative data sets, the composition, applicant, and separation data, provided valuable information on the representation of female officers and officers of color. In addition, the data from the interviews provided in-depth explanations, descriptions, and insights of how the agency accomplished passive representation and the effects of changing demographics within their agency, which provided elements of active representation.



This study identified several factors that increased passive representation of female officers, officers of color, and officers who are fluent in languages other than English. These factors include the Human Resources Office and command staff prioritizing being reflecting of and responsive to the communities it serves, the agency setting goals for hiring and composition of female officers and officers of color (underrepresented groups), and the chief taking an active role in the hiring process, particularly in implementing changes in the application process.

It is evident from the composition, applicant, and interview data that representation of females and officers of color has increased in numbers and percentages for both applications and officers hired since the change of command staff in 2012. These increases are a result of the Human Resources Office and the agency prioritizing being reflective of its communities. This includes setting hiring goals, targeting recruiting efforts, and implementing changes in the application process.

The command staff set goals for their agency to be reflective of its communities. One goal is to have 50% of their officers from underrepresented groups for each job posting and overall composition of the agency. A second goal is that the percentages of female officers and officers of color hired are comparable to the percentages that applied. Respondents identified the role of leadership as a crucial factor for both the increases of female officers and officers of color and their agency engaging in community-oriented and problem-oriented policing. Another important factor is the role of the command staff in gaining support and cooperation from members throughout the agency, including management, supervisors, and officers.

In addition to the minimum selection standards for police officers set by the state regulatory board, all respondents discussed that their human resources office sets the standards, rules, and regulations for hiring practices. The Human Resources Office was described by



interview respondents as an Equal Opportunity Employer that has a formal Affirmative Action plan to reflect the workforce demographics of its communities. An important factor in increasing the hiring of female officers and officers of color is that the agency's employment practices are not under civil service rules. Respondents stated that many police agencies, particularly for large agencies, operate under civil service rules. Respondents clarified the influence of civil service versus their employment practices on hiring diverse applicants. They stated that under civil service rules there are rankings and an eligibility list created where candidates are hired in the order they are ranked. Instead, their agency hires eligible applicants from a pool of qualified candidates, not from a ranked order. Respondents perceived this as an important factor that increased the hiring of female applicants, applicants of color, and applicants who are fluent in languages other than English.

Another factor identified by respondents is that although their union contract influences promotional practices, the contract does not influence hiring practices. Respondents stated that within the framework established by the Human Resources Office and their union contract not influencing hiring decisions, the chief is provided with the authority and flexibility to make decisions regarding hiring practices, specifically by implementing changes to the steps in the application process. Respondents provided the following examples where this flexibility was applied: adding desired qualifications, moving the second interview from after background investigations to before background investigations; eliminating the written test; making the interviews pass/fail instead of using a ranking; having the backgrounds conducted by officers within the agency instead of contracting with investigators outside the agency; and, selecting people from diverse backgrounds to be on the oral boards.



It is also evident from the applicant and interview data that this agency places great emphasis on people skills, specifically, communication skills and possessing a community mindset. The interview data results indicated that there is a reciprocal relationship between community-oriented and problem-oriented policing and recruitment, hiring, and retention.

Although the practices resulted in increasing diversity, interview respondents indicated that there were unintended negative consequences, including perceptions and concerns that some of the diverse applicants were not qualified or competent and that there was increased separation of officers for failed probation.

It is imperative that police agencies examine their hiring requirements to ensure they are predictive of the tasks police perform. Over the past several decades, the role and function of policing has expanded and evolved. These changes include a broader range of tasks, the demands of policing in a pluralistic society due to changing population demographics, the engagement of community-oriented and problem-oriented policing, and changes in the societal and political environment calling for accountability, trust and legitimacy.

As discussed in Chapter II, the theory of representative bureaucracy has evolved since its inception in the 1940s. Groeneveld and Van de Walle (2010) argue "that representative bureaucracy is a multidimensional and changing concept" (p. 239). The responses in the interview data illustrate what Stein (1986), Selden (1997), Meier et al. (1999), and Sowa and Selden (2003) claimed, that public agencies can be more responsive to the public if they are broadly representative of the public they serve. Respondents stating that their agency prioritizes being reflective of their communities also supports the assertion by Meier and Nicholson-Crotty (2006) that passive representation is an "indicator of the organization's commitment to equal



opportunity and access to power and can promote the legitimacy of public bureaucracies" (p. 851).

Diverse police agencies that are reflective of the communities they serve are vital to community-oriented and problem-oriented policing and in repairing fractured relationships by establishing trust and a partnership with the communities. Agencies that engage in community-oriented and problem-oriented policing also attract more female officers and officers of color. One effect of reflective police agencies was a competency effect, identified by interview respondents who stated that female officers and officers of color possess skills that are consistent with community engagement and can help other officers understand different cultures and backgrounds. This is consistent with Sklansky's (2006) argument that changes in police demographics result in special competencies described as "greater understanding of minority communities, and greater credibility in minority communities" (p. 1224). Another effect identified by respondents was that being reflective of its communities improves their agency's credibility. This illustrates an example of Sklansky's community effects where diversity within a police agency positively affects the relations with the community.

It was evident from the separation data that the rates of female officers who resigned and involuntarily separated were lower than the rates for male officers, which is contrary to the most recent research cited. The interview data regarding the perceptions on why officers have left their agency validated the data results of the separation data. In addition, the interview data added detailed information that enriches the data.

The findings from this study have several implications for this agency, the communities it serves, and for other police agencies. The results from the application data indicate that targeted recruitment and changes in the application process increased representation of female officers



and officers of color. The results suggest that the traditional selection and hiring process for police officers may negatively affect female applicants and applicants of color. It is important to note that the changes implemented in the application process still meet the requirements for police officers by the state regulatory board. This agency may use the results of this study to evaluate the future impact of the changes implemented in the recruiting and hiring practices at different stages in an officer's career. This includes assessing the impact on officer retention. The agency may also evaluate the future impact of increasing their diversity on the performance of their officers in being responsive to its communities.

Other police agencies who want to be reflective of their communities and engage in community-oriented and problem-oriented policing can use this case study as an example. This includes committing resources to recruiting efforts, targeting specific groups for recruitment, examining their application process and be willing to change the process to ensure that their qualifications are consistent with the skills needed to engage in community policing, and ensure that diverse officers are welcomed and respected within their agency. Those agencies where the employment practices are under civil service may still implement changes in the application process through civil service procedures dictated by state statute and local ordinances.

The literature cited in Chapter II presented ways in which passive representative bureaucracy may lead to active representative bureaucracy in some form of policy outputs, programs, and decision making that positively impact the interests of female police officers, police officers of color, and communities of color. Bradbury and Kellough (2008) and Kennedy (2014) added that a bureaucracy that is diverse will promote public policy that is more responsive and represent all interests in the formulation and implementation of policy. Benefits from a diverse agency can be the result of direct sources, or indirect sources that influence the



behaviors of others. Examples of active representation provided by respondents include having a symbolic nature of representation; achieving social equity; emphasizing the service aspects regarding the role of police, including communication skills and a community mind-set; increasing credibility, trust, and cooperation from their communities; increasing engagement of community-oriented and problem-oriented policing; improving the attitudes of the public towards the police; increasing officer's awareness, understanding, and acceptance of different cultures, backgrounds and experiences in other officers within the agency; and attracting and retaining officers who wish to engage in community policing.

Contributions of this Study to the Literature

Although this case study includes only one agency, the results of the data provide a richness in detail. This research examines the various practices, factors, and variables of one police agency that recruits, hires, and retains diversity that is reflective of the communities it serves and makes several contributions to the literature. This study benefits from the triangulation of multiple data sources that strengthens reliability and validity and reduces possible response bias and interviewer bias in the interviews. This study examines full-time police officer composition data over a nine-year period from 2009 to 2017 and all applicants for full-time police officer at the case agency over an eight-year period from 2009 through 2016. This includes examining data before and after a change in command staff in 2012 to assess the impact of the changes in recruiting and hiring practices. In addition, this study examines each step in the application process to determine if certain steps in the application process result in different passing rates for males and females and the five categories of race/ethnicity.

The separation data from 2010 through mid-2017 provide information over an eight-year period and is also more current than the most recent 2008 data found in prior literature, which



includes different and/or additional factors that may influence police officer separation from 2008 to mid-2017. The results of the separation data and interview data regarding the reasons officers have left the agency and ways to retain officers can help this agency and other agencies retain officers, especially in reducing early separation of female officers and officers of color.

This study also adds to the literature by conducting qualitative in-depth interviews to capture the perspectives of those responsible for the implementation of recruiting, hiring, and retention practices in the context of a law enforcement agency. Also, the interviews provided data that could not be obtained from only the three quantitative data sets. This includes capturing the respondents' perceptions that provided validation to the three quantitative data sets. This also includes detailed descriptions and explanations of the steps in the application process, information regarding the changes implemented in the application process, and factors perceived to influence recruitment, hiring, and retention in general, and for increasing the representation of female officers and officers of color. This study also includes the complexity and combination of factors involved. This study adds to the literature by increasing our understanding of representative bureaucracy by refining, extending, and offering new insights to the theory, concepts, and prior literature, and contextualizing what factors influence recruiting, hiring, and retaining diverse police agencies.

Discussed next are the results and conclusions for the data analysis of the four data sets presented in Chapter IV and Chapter V: composition data, application data, separation data, and interview data. Then, the results of the data are compared to previous literature presented in chapter II. The chapter concludes with the limitations of this study and recommendations for future research.



Discussion and Conclusions for the Composition Data

Table 4, previously presented in Chapter IV, shows that in 2009, 3.7% of the officers were female and the percentage of female officers increased each year to 17.0% in 2017.

Noteworthy is that the percentage of female officers in 2017 was more than four and one-half times the percentage in 2009. The interview respondents identified that the change in command staff in 2012 was responsible for the increase in their agency's diversity by prioritizing being reflective of and responsive to the community. This led to the 60% increase in the percentage of females between 2013, 10.6%, and 2017, 17.0%. Compared to the percentage of females from the previous year, there was a 13% increase in 2013, a 11% increase in 2014, a 13% increase in 2015, a 24% increase in 2016, and a 3% increase in 2017.

The percentages of female officers can also be compared to the percentages of female officers in the U.S. previously presented in Chapter II. Almost 12% of officers in local police departments in Langton's (2010) 2007 data were female and about 12% in the 2013 LEMAS survey data (Reaves, 2015; Stevens, 2018). The percentages of female officers for 2009 through 2014 were below 12.0% and for 2015, 2016 and 2017, were more than 12.0% with 13.3%, 16.5%, and 17.0%, respectively.

Table 5, previously presented in Chapter IV, shows a 127% increase in officers of color s between 2009, 13.0%, and 2017, 29.5%. In 2009, 13.0% of the agency's officers were officers of color, in 2010, the percentage of officers of color increased to 13.5%, in 2011 and 2012 the percentages decreased, and from 2013 to 2017, the percentages increased each year to 29.5% officers of color in 2017. There was a 214% increase in the percentage of officers between 2013, 10.6%, and 2017, 17.0%. Compared to the percentage of officers of color from the previous year, there was a 13% increase in 2013, an 89% increase in 2014, a 28% increase in 2015, a 14%

increase in 2016, and a 0.7% increase in 2017. The command staff's priority of diversifying the agency and the chief implementing changes in the application process discussed for the increase of females also likely contributed to the increase in officers of color starting in 2013. Adding fluency in languages other than English as a desired qualification in the 2013 through 2016 job postings and working with diverse populations as a desired qualification in the 2014 through 2016 postings also likely contributed to increasing the percentages of officers of color.

The percentages of officers of color within the agency can also be compared to 27% of officers of color in local police departments nationwide from the 2013 LEMAS survey data cited in Chapter II (Reaves, 2015). The percentages of officers for the years 2009 through 2015 were still below 27%, but by 2016 and 2017, had exceeded the national average of 27%, with 29.3% and 29.5%, respectively.

The data in Table 6, discussed in Chapter IV, can be compared to the percentages for the make-up of race/ethnicity in the population that the case agency serves. As discussed in Chapter II, data from the 2010 Census Bureau were used to calculate the percentages for the five categories of ethnic groups used by the case agency. The percentages were 79% White, 5% Hispanic / Latino, 8% Black / African American, 6% Asian / Pacific Islander, and 0.6% American Indian / Native Alaskan. The data results from Table 6 in Chapter IV show that the agency is reflective of the communities it serves. For Hispanics / Latinos, the percentages for officers in 2009, 2010, 2014, 2015, 2016, and 2017 were more than the 5% in the population and the percentages of Hispanics and Latinos in 2011, 2012, and 2013 were just under 5%.

Noteworthy is that the agency was reflective of the city population for Hispanics and Latinos before and after the change in command staff. For Blacks / African Americans, the percentages for officers in 2014, 2015, 2016, and 2017 were more than the 8% in the population. For Asians



/ Pacific Islanders, the percentages for officers in 2016 and 2017 were more than the 6% in the population and the percentages increased after the change in command staff. For American Indians / Native Americans, in 2009, 2010, 2011, 2015, 2016, and 2017, there were more than the 0.6% in the population. Noteworthy is that the agency was reflective of population demographics before the change in command staff and the percentage increased after the change of command staff. One reason for these results in meeting population demographics may be attributed to the human resources office's effort of ensuring that its job postings are inclusive.

These results support research by Hochstedler et al. (1984), Stein (1986), Lewis (1989), Kim and Mengistu (1994), Selden (1997), Zhao and Lovrich (1998), Goode and Baldwin (2005), and Morabito and Shelley (2015), finding that the size of local minority populations is positively associated with officers of color. Possible explanations for this finding include pressure from communities of color to be more reflective of and responsive to the make-up of their communities. Another explanation extends Zhao and Lovrich's (1998) argument regarding African Americans to other communities of color, that the increase of African American officers is influenced by a larger African American population leading to more political power and influence on personnel policies and equitable hiring and promotional practices in police departments. This element was evident in respondent's statements that being reflective of and responsive to their communities is a priority set by the command staff and integrated throughout the agency.

Discussion and Conclusions for the Application Data

There was a total of 13 job postings, seven job postings open to both external and internal applicants and six job postings open only to internal employees. The seven postings open to both external and internal applicants were in 2009, 2010, 2011, 2012, 2013, 2014, and 2016.



The six postings open only to internal applicants were from 2011 through 2015 and are grouped together for analysis because of the small number of applicants.

Discussed first are the seven postings open to external and internal applicants, including the percentages of female applicants and applicants of color for each step of the application process and the differences between the percentages of females and applicants of color who applied and who were hired. This discussion includes the impact of the different steps in the application process on female applicants and applicants of color, the impact of the changes implemented in the application process on female applicants and applicants of color, and the implications of the results. Discussed last are the results and implications of the data results for the six postings open only to internal applicants.

Tables 96-101 compiled data presented in Chapter IV which shows the percentages of female applicants and applicants of color who passed each step of the application process from application to hiring for the seven postings open to external and internal applicants from 2009 through 2016.

Overview of Applicant Data

The following is an overview of the percentages of female applicants and applicants of color, including the differences between the percentage of females and applicants of color who applied and who were hired for each posting. For female applicants, as Table 96 presents, there were two postings that showed a decrease in the percentage hired compared to the percentage who applied. In 2010, 11.5% females applied but no females were hired and in 2016, 14.5% females applied and 13.3% were hired. For the other five postings, the highest increase from application to hiring was in 2009 where 13.0% females applied and 33.3% were hired. In 2011



and 2013, there were about twice the number of females hired, 20.0% and 23.1% compared to females who applied, 10.7% and 11.6%.

In examining the increase of female applicants, in addition to examining percentages, it is important to examine the number or frequency of female applicants presented in Tables 7, 15, 25, 37, 47, 57, and 67 previously presented in Chapter IV. The number of female applicants was similar for the 2009 through 2012 postings, with 52, 56, 46, and 51 female applicants, respectively, increased to 74 in the 2013 posting, increased to 81 in the 2014 posting, and decreased to 46 female applicants in the 2016 posting. There were only 317 applicants who applied for the 2016 posting. This is consistent with interview respondents stating that applications were down for their agency and for other agencies due to social and community pressure. Even though the total number of applicants decreased for the 2016 posting, the 15% of female applicants is only one percent lower than the previous posting. The increase in both the numbers and percentages of female officers are likely attributed to the targeted recruiting efforts and the changes in the application process.

For applicants of color, Table 96 presents the two postings where the percentages of applicants of color hired were lower than the percentages of applicants of color who applied. In 2009, 20.5% applicants of color applied and 13.4% were hired, and in 2011, 16.5% applicants of color applied but no applicants of color were hired. The other five postings showed an increase in applicants hired compared to the percentages who applied. For the posting in 2010, there was an increase of the percentage of applicants of color hired, 20.0%, compared to the percentage who applied, 16.8%. For the postings in 2012 and 2014, there were more than twice the percentages hired, 40.7% and 55.0%, compared to the percentages who applied, 19.8% and



25.8%. For the postings in 2013 and 2016, there were almost twice the percentages of applicants of color hired, 46.2% and 60.0%, compared to the percentages who applied, 25.8% and 34.1%.

Table 96. Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants.

	2009	2010	2011	2012	2013	2014	2016
Application							
Females	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	14.5%
AOC	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	34.1%
Qualifications							
Females	9.7%	7.9%	11.5%	10.7%	10.1%	14.1%	10.4%
AOC	13.9%	5.3%	15.2%	18.0%	31.3%	41.2%	51.9%
Written Test							
Females	None	None	18.5%	11.8%	13.2%	None	None
AOC			7.4%	18.7%	33.4%		
Interview 1							
Females	10.9%	9.1%	11.1%	9.1%	15.4%	15.7%	11.1%
AOC	15.6%	9.1%	5.6%	20.7%	36.8%	41.0%	53.4%
BI, I2, ME,							
DS, PE, H							
Females	33.3%	0.0%	20.0%				
AOC	13.4%	20.0%	0.0%				
I2, BI, ME,							
DS, PE, H							
Females				14.8%	23.1%	20.0%	13.3%
AOC				40.7%	46.2%	55.0%	60.0%

Applicants, AOC=Applicants of Color

Hiring 1 = The following order in the hiring step: Background Investigation (BI), Interview 2 (I2), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H). Hiring 2 = The following order in the hiring step: Interview 2 (I2), Background Investigation (BI), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H).

In examining the increase of applicants of color, in addition to examining percentages, it is important to examine the number or frequencies of applicants of color presented in Tables 8, 16, 26, 38, 48. 58, and 68 in Chapter IV. The number of applicants of color was 82, 82, and 72 for the postings in 2009 through 2011, increased to 101 in the 2012 posting, increased to 165 in the 2013 posting, decreased to 133 in the 2014 posting, and decreased to 108 in the 2016 posting.

This decrease in 2016 was a result in the decrease of total applicants as discussed and the 34% of applicants of color who applied in 2016 was highest percentage for all the postings. Even though the number of applicants of color decreased for the 2014 and 2016 postings, the percentages increased to 55% and 60% respectively. The increase in both the numbers and percentages of officers of color are likely attributed to the targeted recruiting efforts and the changes in the application process.

As previously stated, there was a change in the command staff in 2012. One way to assess the impact of the changes in the application process implemented by the command staff after 2012 is to compare the percentages of female applicants and applicants of color in the steps of the application process in the postings before and after the change of command staff in 2012. When the changes were made also needs to be taken into consideration. The postings that included these changes resulting from the change in command staff were the 2012, 2013, 2014, and 2016 postings. The 2012 posting closed at the end of 2012 and the steps in the application process took place in early 2013. The 2013 posting closed in early 2014 and the steps in the application process took place in 2014. The 2014 posting closed towards the end of 2014 and the steps in the application process took place at the end of 2014. This information was verified by the list and description of the steps and the order of the steps in the job announcements and the application data, the respondent's descriptions of the steps in the interviews, and confirming this information with a member of the management team. The increase in the numbers and percentages are likely attributed to targeted recruiting and the changes implemented in the application process.



The Impact of Changes in the Hiring Process on Increasing Diversity

The following section discusses four changes in the application process that were identified as resulting from the change in command staff in 2012 and their impact on the recruitment and hiring of female officers and officers of color at the case agency. The first change is moving the second interview from after background investigations to before background investigations in the 2012, 2013, 2014, and 2016 postings. The second change is the addition of desired qualifications, fluency in languages other than English in the 2013, 2014, and 2016 postings and experience working with diverse populations in the 2014 and 2016 postings. The third change is elimination of the written test in the 2014 and 2016 postings. The fourth change are two changes in the background investigations that took effect in the 2014 and 2016 postings and are covered as one change. Background investigations were conducted by officers within the agency instead of being contracted with background investigators from outside the agency as was the case in the previous postings. Also, before the 2014 and 2016 postings, background investigators had the ability to eliminate applicants and this was changed to backgrounder investigators making recommendations not to hire, and if they do not recommend an applicant to be hired, they are required to state the reason why they do not recommend an applicant to be hired. Thus, this decision whether or not to hire based on the background was moved to the command staff and hiring manager instead of the background investigator.

The First Change - Moving the Second Interview to Before the Background Investigation

The first change in the application process discussed is moving the second interview from after background investigations to before background investigations in the 2012, 2013, 2014, and 2016 postings. All respondents in the interviews indicated that this change affected who was hired. Terms used to describe this change were "the list starts over" and "scores are reset at 0".



The descriptions from respondents provided clarity and explanations for why this was the case. One respondent stated that "because what the interview panel looks for [in the first interview], is this person likely to be a good cop in a very general sense" and what the second interview "looks for, is will this person be a good cop here ... they're different sets of criteria". The content of the five questions command staff asked the applicants in the second interview, previously discussed in Chapter V, are consistent with interview respondents stating that the qualifications and skills applicants should possess include good communication skills, being ethical, having integrity, having a community mind-set, and having experience working with diverse communities.

Noteworthy is that the respondents' list is also consistent with the skills and qualifications necessary to engage in community policing.

Respondents stated that moving background investigation to after the second interview changed what most agencies historically have done. In addition to increasing female applicants and applicants of color hired, moving the second interview to before background investigations sped up the hiring process because it reduced the number of backgrounds and the time it took to make a conditional offer to candidates.

Noted in bold print in Table 97 below are the postings showing where the second interview is located in the order of the application process. In the 2009, 2010, and 2011 postings, there were 33.3%, 0.0%, and 20.0% female applicants hired, respectively, and in the 2012, 2013, 2014, and 2016 postings, there were 14.8%, 23.1%, 20.0%, and 13.3% female applicants hired respectively. The postings with the second interview after background investigations shows a wider range in the percentages, 0.0% to 33.3%, compared to the postings with the second interview before background investigations, 14.8% to 23.1%.



In the 2009, 2010, and 2011 postings, there were 13.4%, 20.0%, and 0.0% applicants of color hired, respectively, and in the 2012, 2013, 2014, and 2016 postings, there were 40.7%, 46.2%, 55.0%, and 60.0% applicants of color hired, respectively. The postings with the second interview after background investigations show a range of 0.0% to 20.0%, compared to a range of 40.7% to 60.0% in the postings with the second interview before background investigations. The percentage for applicants of color hired increased by more than 20% in the 2012 posting from the 2011 posting, and increased in the 2013, 2014, and 2016 postings. This suggests that the change in the second interview likely influenced the increase of more than 20% in the percentage of applicants of color observed in the 2012 posting. The 2013 posting also added fluency in languages other than English as a desired qualification, so this change may have also influenced the increase of applicants of color at the qualifications step through the remaining steps in the hiring process. There were also three other changes in the application process for both the 2014 and 2016 postings that may also have influenced the increase in applicants of color, adding experience working with diverse populations as a desired qualification, the changes in the background investigations, and eliminating the written test. These results suggest a combined effect of the changes to account for the increase of applicants hired in the 2013, 2014, and 2016 postings.



Table 97. Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – First Change.

	2009	2010	2011	2012	2013	2014	2016
Application							
Females	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	14.5%
AOC	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	34.1%
Qualifications							
Females	9.7%	7.9%	11.5%	10.7%	10.1%	14.1%	10.4%
AOC	13.9%	5.3%	15.2%	18.0%	31.3%	41.2%	51.9%
Written Test							
Females	None	None	18.5%	11.8%	13.2%	None	None
AOC			7.4%	18.7%	33.4%		
Interview 1							
Females	10.9%	9.1%	11.1%	9.1%	15.4%	15.7%	11.1%
AOC	15.6%	9.1%	5.6%	20.7%	36.8%	41.0%	53.4%
BI, I2, ME,							
DS, PE, H							
Females	33.3%	0.0%	20.0%				
AOC	13.4%	20.0%	0.0%				
I2 , BI, ME,							
DS, PE, H							
Females				14.8%	23.1%	20.0%	13.3%
AOC				40.7%	46.2%	55.0%	60.0%

Applicants, AOC=Applicants of Color

Hiring 1 = The following order in the hiring step: Background Investigation (BI), Interview 2 (I2), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H). Hiring 2 = The following order in the hiring step: Interview 2 (I2), Background Investigation (BI), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H).

The Second Change - Addition of Desired Qualifications

As noted previously in Chapter IV, fluency in languages other than English was added in the 2013, 2014, and 2016 postings, and experience working with diverse populations was added in the 2014 and 2016 postings. There were three methods used to assess how these two additional desired qualifications impacted the percentage of applicants of color. The first method is examining the data for the three postings that added the first desired qualification, fluency in other languages besides English, 2013, 2014, and 2016, which also includes the two postings that 200

added the second desired qualification, experience working with diverse populations. As presented in bold print in Table 98 below, for the postings in 2013, 2014, and 2016, the percent of applicants of color who passed the qualifications review increased to 31.3%, 41.2%, and 51.9%, from the percent of applicants of color in the previous step, applications, 25.8%, 25.8%, and 34.1% respectively. These increases for the three postings, 5.5%, 15.4%, and 17.8%, suggests that the first desired qualification influenced the increase observed in the 2013 posting, and the increases of more than three times observed in the 2014 and 2016 posting likely were influenced by the addition of both desired qualifications.

As presented in bold print in Table 98 below, a second way to examine the impact of the addition of desired qualifications is comparing the percentages of applicants of color who passed the qualifications step in the postings that did not include these two desired qualifications, 13.9% in the 2009 posting, 5.3% in the 2010 posting, 15.2% in the 2011 posting, and 18.0% in the 2012 posting, to the percentages of applicants of color who passed the previous step, application, in the same postings, 20.5%, 16.8%, 16.5%, and 19.8%, respectively. In all four postings, the percentages of applicants who passed the qualifications step was lower than the percentages in the previous step, application. This is contrary to the increases observed between the two steps for all three of the postings in 2013, 2014, and 2014. This also suggests that the addition of these two desired qualifications influenced these increases.

Also presented in bold print in Table 98 below, a third way to examine the impact of these desired qualifications is comparing the percentage of applicants of color who passed the qualifications step in the postings where these two desired qualifications were not present, the 2009 posting, 13.9%, the 2010 posting, 5.3%, the 2011 posting, 15.2%, and the 2012 posting, 18.0%, to the postings where the desired qualifications were added, the 2013 posting, 31.3%, the



2014 posting, 41.2%, and the 2016 posting, 51.9%. The range for the percentage of applicants of color is 5.3% to 18.0% for the postings before the two desired qualifications were added compared to the range of 31.3% to 51.9% for the postings where the two desired qualifications were added. This also suggests that these desired qualifications positively influenced the increases observed. Respondents also indicated that there are symbolic and practical effects of officers of color and officers who are fluent in languages other than English by increasing the agency's credibility and improving relationships with communities of color.

Table 98. Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – Second Change.

	2009	2010	2011	2012	2013	2014	2016
Application							
Females	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	14.5%
AOC	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	34.1%
Qualifications							
Females	9.7%	7.9%	11.5%	10.7%	10.1%	14.1%	10.4%
AOC	13.9%	5.3%	15.2%	18.0%	31.3%	41.2%	51.9%
Written Test							
Females	None	None	18.5%	11.8%	13.2%	None	None
AOC			7.4%	18.7%	33.4%		
Interview 1							
Females	10.9%	9.1%	11.1%	9.1%	15.4%	15.7%	11.1%
AOC	15.6%	9.1%	5.6%	20.7%	36.8%	41.0%	53.4%
BI, I2, ME,							
DS, PE, H							
Females	33.3%	0.0%	20.0%				
AOC	13.4%	20.0%	0.0%				
I2, BI, ME,							
DS, PE, H							
Females				14.8%	23.1%	20.0%	13.3%
AOC				40.7%	46.2%	55.0%	60.0%

Applicants, AOC=Applicants of Color

Hiring 1 = The following order in the hiring step: Background Investigation (BI), Interview 2 (I2), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H). Hiring 2 = The following order in the hiring step: Interview 2 (I2), Background Investigation (BI), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H).



The Third Change - Elimination of the Written Test

The 2009, 2010, 2014, and 2016 postings did not include a written test and the 2011, 2012, and 2013 postings had a written test as one of the steps in the application process. Eliminating the written test was one of the changes implemented by the chief in 2012. It is interesting to note that the postings in 2009 and 2010 also did not include a written test which occurred under the previous command staff.

Reasons why respondents stated the written test was eliminated included the written test was not a good indicator of performance, eliminated applicants who possess the skills they are looking for and diverse applicants, and decreased the numbers of the applicant pool going into the first interview. Three respondents expressed concerns that the written test assessed cognitive abilities and the aptitude to learn and added that they did not know of a different test to assess these.

There were two analyses utilized to address the impact of the written test on female applicants and applicants of color. The first analysis examines the data in the three postings that included a written test, 2011, 2012, and 2013. As presented in bold print in Table 99 below, the percentages of females and applicants of color who passed the previous step, the qualifications review, are compared to the percentage of females and applicants of color who passed the written test. In the 2011, 2012, and 2013 postings, the percentages of female applicants who passed the qualifications step, 11.5%, 10.7%, and 10.1%, increased to 18.5%, 11.8%, and 13.2%, respectively, of females who passed the written test. These results demonstrate that female applicants did better than males on the written test. In the 2011 posting, the percentage of applicants of color who passed the qualifications step, 15.2%, decreased by about 50% to 7.4% who passed the written test. In contrast, the percentages increased slightly for the 2012 and 2013



postings to 18.4% and 33.4% from 18.0% and 31.3%, respectively. More than half of the applicants were eliminated by the written test in the 2011 posting. However, for the 2012 and 2013 postings, this was reduced to eliminating only 0.4% and 2.1%, respectively. This is interesting because this is contrary to the perceptions of three respondents who perceived that the written test eliminated applicants of color. In the 2011 posting, almost 50% of the applicants of color from the previous step were eliminated by the written test, however, the percent of applicants of color increased by 4% and 7% in the 2012 and 2013 posting, respectively.

Table 99. Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – Third Change – First Analysis.

	2009	2010	2011	2012	2013	2014	2016
Application							
Females	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	14.5%
AOC	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	34.1%
Qualifications							
Females	9.7%	7.9%	11.5%	10.7%	10.1%	14.1%	10.4%
AOC	13.9%	5.3%	15.2%	18.0%	31.3%	41.2%	51.9%
Written Test							
Females	None	None	18.5%	11.8%	13.2%	None	None
AOC			7.4%	18.7%	33.4%		
Interview 1							
Females	10.9%	9.1%	11.1%	9.1%	15.4%	15.7%	11.1%
AOC	15.6%	9.1%	5.6%	20.7%	36.8%	41.0%	53.4%
BI, I2, ME, DS,							
PE, H							
Females	33.3%	0.0%	20.0%				
AOC	13.4%	20.0%	0.0%				
I2, BI, ME, DS,							
PE, H							
Females				14.8%	23.1%	20.0%	13.3%
AOC				40.7%	46.2%	55.0%	60.0%

Applicants, AOC=Applicants of Color

Hiring 1 = The following order in the hiring step: Background Investigation (BI), Interview 2 (I2), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H). Hiring 2 = The following order in the hiring step: Interview 2 (I2), Background Investigation (BI), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H).



The second analysis to assess the effects of the written test on female applicants and applicants of color is comparing the percentages of female applicants and applicants of color who passed the qualifications step for the postings that did not include a written test, 2009, 2010, 2014, and 2016, to the percentages of female applicants and applicants of color who passed the written test for the postings that included a written test, 2011, 2012, and 2013, which are presented in bold print in Table 100 below. For the postings that did not include a written test, the percentages of female applicants who passed the qualifications step ranged from 7.9% to 14.1% and the percentages of female applicants who passed the written test ranged from 11.8% to 18.5%. These results confirm the findings from the other comparison above, that females do better than males on the written test. For the postings that did not include a written test, the percentages of applicants of color who passed the qualifications step ranged from 5.3% to 51.9% and the percentages of applicants of color who passed the written test ranged from 7.4% to 33.4%. The range for applicants of color who passed the written test is narrower, 26.0%, than the range for applicants of color who passed the qualifications step, 46.6%, however the range for applicants of color who passed the written step falls within the range for applicants of color who passed the qualifications step. This suggests that even though some applicants of color are eliminated, the impact on applicants of color does not appear to be a disproportionate impact. Also, as noted above, the increase to 41.2% in the 2014 posting and 51.9% in the 2016 postings for applicants of color occurred in qualifications step, which were likely influenced by the addition of the desired qualifications.



Table 100. Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants - Third Change – Second Analysis.

	2009	2010	2011	2012	2013	2014	2016
Application							
Females	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	14.5%
AOC	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	34.1%
Qualifications							
Females	9.7%	7.9%	11.5%	10.7%	10.1%	14.1%	10.4%
AOC	13.9%	5.3%	15.2%	18.0%	31.3%	41.2%	51.9%
Written Test							
Females	None	None	18.5%	11.8%	13.2%	None	None
AOC			7.4%	18.7%	33.4%		
Interview 1							
Females	10.9%	9.1%	11.1%	9.1%	15.4%	15.7%	11.1%
AOC	15.6%	9.1%	5.6%	20.7%	36.8%	41.0%	53.4%
BI, I2, ME,							
DS, PE, H							
Females	33.3%	0.0%	20.0%				
AOC	13.4%	20.0%	0.0%				
I2, BI, ME,							
DS, PE, H							
Females				14.8%	23.1%	20.0%	13.3%
AOC				40.7%	46.2%	55.0%	60.0%

Applicants, AOC=Applicants of Color

Hiring 1 = The following order in the hiring step: Background Investigation (BI), Interview 2 (I2), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H). Hiring 2 = The following order in the hiring step: Interview 2 (I2), Background Investigation (BI), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H).

The Fourth Change - Changes in the Background Investigations

The 2014 and 2016 postings included the background investigations being conducted inhouse instead of being contracted from outside the agency. One respondent described the previous process of background investigations being conducted from officers outside the agency as "thorough but eliminated candidates who were recruited" for this agency. Three respondents also stated having background investigation conducted by officers within the agency may help retain officers because it provides officers with an opportunity to perform additional duties.



The previous process also allowed background investigators to eliminate candidates, which was changed to background investigators making recommendations to not hire instead of their ability to eliminate candidates, and if they recommended an applicant not to be hired, they were required to state the reason why. This change moved the decision whether or not to eliminate applicants in the background investigation from the backgrounder to the command staff and hiring manager. This allowed for consideration of the circumstances for each applicant. Several respondents stated that they support this change because if people made a mistake and grow from that mistake, they should not be automatically eliminated.

Table 101. Percent of Females and Persons of Color Hired for Full-time Police Officer by each Step in the Application Process from the 2009 through 2016 Postings Open to External and Internal Applicants – Fourth Change.

	2009	2010	2011	2012	2013	2014	2016
Application							
Females	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	14.5%
AOC	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	34.1%
Qualifications							
Females	9.7%	7.9%	11.5%	10.7%	10.1%	14.1%	10.4%
AOC	13.9%	5.3%	15.2%	18.0%	31.3%	41.2%	51.9%
Written Test							
Females	None	None	18.5%	11.8%	13.2%	None	None
AOC			7.4%	18.7%	33.4%		
Interview 1							
Females	10.9%	9.1%	11.1%	9.1%	15.4%	15.7%	11.1%
AOC	15.6%	9.1%	5.6%	20.7%	36.8%	41.0%	53.4%
BI, I2, ME, DS,							
PE, H							
Females	33.3%	0.0%	20.0%				
AOC	13.4%	20.0%	0.0%				
I2, BI, ME, DS,							
PE, H							
Females				14.8%	23.1%	20.0%	13.3%
AOC				40.7%	46.2%	55.0%	60.0%

Applicants, AOC=Applicants of Color

Hiring 1 = The following order in the hiring step: Background Investigation (BI), Interview 2 (I2), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H). Hiring 2 = The following order in the hiring step: Interview 2 (I2), Background Investigation (BI), Medical Examination (ME), Drug Screen (DS), Psychological Evaluation (PE), Hired (H).



These two changes are difficult to assess because the background investigation is part of the hiring step, so it cannot be isolated from the second interview, medical examination, drug screen, and psychological evaluation. As presented in Table 101, these two changes possibly contributed to the increases observed from the previous step, the first interview, increases of 27% in the 2014 posting and 20% in the 2016 posting for female applicants, and increases of 34% in the 2014 posting and 12% in the 2016 posting for applicants of color.

The Impact of the Changes in the Application Process on the Hiring Goals for the Postings Open to External and Internal Applicants

Another method used to assess the impact of the changes in the application process on female applicants and applicants of color is determining whether the hiring goals identified and described during the interviews were met. As discussed in Chapter V, one of the hiring goals implemented by the current command staff was hiring at least 50% from underrepresented populations, which includes women and people of color. As described in Chapter V, one respondent explained that females are included in "Underrepresented Groups" because females are underrepresented in law enforcement. The percentage presented for Underrepresented Groups hired for each job posting was calculated by adding the percentages of females hired to the percentages of persons of color hired for each job posting. Females of color were counted for both females and persons of color. As presented in bold print in Table 102, the percentages of Underrepresented Groups hired from the 2009 posting was just under 50% with 46.9%, the 2010 and 2011 postings were considerably less than 50% with 20% for both postings, the 2012 posting was 55.5%, the 2013 posting was 69.3%, the 2014 posting was 75.0%, and the 2016 posting was 73.3%. The hiring goal of 50% from Underrepresented Groups was met in the 2012, 2013, 2014, and 2016 postings with 55.5%, 69.3%, 75%, and 73.3%, respectively, was just under the goal in



the 2009 posting with 46.9%, and considerably lower than the goal in both the 2010 and 2011 postings with 20% for both postings. It is important to note that the percentages of females hired is considerably lower than the percentages of persons of color hired.

Respondents stated in the interviews that diversifying their agency was a priority with the current command staff as of 2012. The increases in females and persons of color hired observed in all four of the postings after the change of the command staff, are likely a result of the command staff prioritizing diversity and implementing changes in the application process to meet this goal. It is interesting that the 46.9% of females and persons of color from the 2009 posting was under the previous command staff. As stated previously, the 2012 posting closed at the end of 2012 and the application process took place in 2013 which was after the change in command staff. The second interview was moved to before the background investigation in the 2012 posting, which also likely influenced the increase in Underrepresented Groups by 30% from the 2010 and 2011 postings.

Table 102. Percent of Underrepresented Groups Applied and Hired for Full-time Police Officer from the 2009 through 2016 Postings Open to External and Internal Applicants.

Year	2009	2010	2011	2012	2013	2014	2016
Females							
Applied	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	14.5%
Hired	33.3%	0.0%	20.0%	14.8%	23.1%	20.0%	13.3%
Persons of Color							
Applied	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	34.1%
Hired	13.4%	20.0%	0.0%	40.7%	46.2%	55.0%	60.0%
Underrepresented							
Groups							
Applied	33.5%	28.3%	27.2%	29.8%	37.4%	41.5%	48.6%
Hired	46.7%	20.0%	20.0%	55.5%	69.3%	75.0%	73.3%

Underrepresented Groups include females and persons of color



The second goal identified during the interviews was to hire the same percentages of female officers and officers of color as the percentages who applied. For females hired, the bold print in Table 103 shows that this second goal was met for every posting except for two postings. In the 2010 posting, 11.5% females applied but none were hired, and in the 2016 posting, 14.5% applied and 13.3% were hired, which was a slight decrease of 1.2%. For applicants of color hired, this second goal was not met in the 2009 or 2011 postings where 20.5% and 16.5% applied and 13.4% and 0.0% were hired, respectively. The postings in 2010, 2012, 2013, 2014, and 2016 met this second goal. For the postings from 2012 on, the percentage of applicants of color hired was just less than twice to more than twice that of the percentage who applied. These increases were likely influenced by the changes in the application process.

Table 103. Percent of Females and Persons of Color Applied and Hired for Full-time Police Officer from the 2009 through 2016 Postings Open to External and Internal Applicants.

Year	2009	2010	2011	2012	2013	2014	2015	2016
Females								
Applied	13.0%	11.5%	10.7%	10.0%	11.6%	15.7%	N/A	14.5%
Hired	33.3%	0.0%	20.0%	14.8%	23.1%	20.0%		13.3%
Persons of Color								
Applied	20.5%	16.8%	16.5%	19.8%	25.8%	25.8%	N/A	34.1%
Hired	13.4%	20.0%	0.0%	40.7%	46.2%	55.0%		60.0%
Underrepresented								
Groups								
Applied	33.5%	28.3%	27.2%	29.8%	37.4%	41.5%	N/A	48.6%
Hired	46.7%	20.0%	20.0%	55.5%	69.3%	75.0%	N/A	73.3%

Underrepresented Groups include females and persons of color

The increase in hiring applicants of color is also observed in Table 80 previously presented in Chapter IV, which presents the applicants of color by the frequency (n) and percentage of applicants hired for each of the five ethnic groups for full-time police officer from the 2009 to 2016 postings. At least 1 Hispanic / Latino was hired in five of the seven years, at



least 1 Black / African American was hired in five of the seven years, at least 1 Asian / Pacific Islander was hired in four of the seven years, and at least 1 American Indian / Alaskan Native was hired in two of the seven years. The highest number of the four non-White groups were Blacks / African Americans with 7 in 2012. The highest percentage of the four non-White groups were Asians / Pacific Islanders with 33.3% hired in the 2016 posting, followed by Blacks / African Americans with 26.7% also in the 2016 posting. For Blacks / African Americans hired, there was a 20% increase between the 2009 and 2016 postings. For Asians / Pacific Islanders, there was an increase between the 2012 and 2016 postings with none hired in the 2009, 2010, and 2011 postings. There were only two postings where American Indians / Alaskan Natives were hired, 2013 and 2016. The increases in numbers and percentages of officers of color hired starting in the 2012 postings compared to the previous postings suggests a positive influence. Only one change in the application process was implemented in the 2012 posting, moving the second interview from after the background to before the background, which likely influenced the increase of applicants hired in the 2012 posting. Fluency in languages other than English as a desired qualification was implemented in the 2013 posting, experience working with diverse populations as a desired qualification, the two changes in the background process, moving investigations in-house and removing backgrounders ability to eliminate candidates to making recommendations, were implemented in the 2014 and 2016 postings. These changes likely had a combined influence on the increase of applicants of color hired.

One other change discussed by respondents in the interviews was the change in the makeup of the interview panels in the first interview. Some respondents indicated there was a perception of losing candidates in the first interview as influencing the need to diversify the interview panel. That was illustrated by one respondent stating that they were told that panel members had to be subject matter experts and replied that "the subject matter they're experts on is being in the community". It was difficult to assess the impact of this change because it cannot be determined what panels were used for each posting because there were more than one panel used for each posting. Also, who was selected to be on the panels depended on the availability of work schedules.

The 2011 to 2015 Postings for Full-time Police Officer Internal Only Applicants

There were six postings from 2011 to 2015 for internal employees only for the position of full-time police officer as previously discussed in Chapter IV. All six postings for internal postings are grouped together in reporting the frequencies and percentages by gender and ethnic group due to the low number of applications. There were 39 applications received for the internal postings. As presented in Tables 81 and 82 in Chapter IV, there were 12.8% female applicants and 38.5% applicants of color, with all four non-White Ethic Groups represented.

There were 14 applicants hired from the six postings for internal employees only. As presented in Table 83 and 84 in Chapter IV, there were 28.6% of female applicants hired which was more than twice the percentage of females who applied, 12.8%. There were 14.3% of applicants of color hired, which was almost one-third less than applicants of color who applied, 38.5%.

Table 85 previously presented in Chapter IV compared the frequencies and percentages of female applicants and applicants of color hired for postings open to external and internal applicants to the postings for internal applicants only. For the postings open to all applicants, 19 (19.0%) of the applicants hired were female, and for the postings open only to internal employees, 4 (28.6%) of the applicants hired were female. For the postings open to all



applicants, 40 (40.0%) of the applicants hired were applicants of color, and for the postings open only to internal employees, 2 (14.3%) of the applicants hired were applicants of color. These results suggest that internal only postings benefit female applicants but not applicants of color. Although the percentage of applicants of color hired was lower than the percentage of applicants of color who applied for the postings open only to internal employees, there are benefits of having postings open only to internal employees. Respondents stated in the interviews that their agency may lose applicants to other agencies that offer them the job first. Having internal only postings may be offered between postings open to external and internal postings, thus applicants do not have to wait as long to apply. Also, due to the small size of the applicant pool in the internal only postings, the application process takes considerably less time so this likely reduces applicants receiving an offer from another agency during the time it takes to get through the application process. Another benefit of internal postings is that because applicants work in other positions within the agency, applicants are more likely to understand what officers do in this agency and the opportunities that exist, and applicants are known within the police agency which may also shorten the time it takes to hire applicants and increase retention.

The Impact of the Changes in the Application Process on the Hiring Goals for the Postings Open Only to Internal Applicants

The two hiring goals previously discussed in the postings open to external and internal applicants above were also compared to the six internal only postings from 2011 through 2015. Five of the six postings occurred in 2013 or after, thus these five postings were influenced by the changes in the application process. Table 104 shows that the goal of 50% Underrepresented Groups was met with 51.3%. The second goal, the percentage of who was hired is comparable to



the percentage who applied, was met for females, the 28.6% hired was more than twice that of who applied, 12.8%. However, this goal was not met for officers of color, with 38.5% who applied, but only 14.3% were hired.

Table 104. Percent of Females, Persons of Color, and Underrepresented Groups Applied and Hired for Full-time Police Officer Internal Only Postings from 2011 through 2015.

Type of Posting	Internal Applicants Only Postings from 2011 through 2015
Females	
Applied	12.8%
Hired	28.6%
Persons of Color	
Applied	38.5%
Hired	14.3%
Underrepresented Groups	
Applied	51.3%
Hired	42.9%

Discussion and Conclusions for Separation Data

The discussion on the separation data includes the turnover rates for the case agency, reasons why officers have separated from the case agency, officers leaving the case agency for law enforcement positions in other agencies, female officers and officers of color leaving the case agency, and officers who left the case agency by length of service. The results of the analysis of the separation data are also compared to research cited in Chapter II.

Turnover Rates

As previously discussed in Chapter IV, there are three broad categories of separation used by the case agency: Resignation, Termination, and Retirement. These three categories are parallel to the three broad categories cited in Chapter II: Voluntary Separation, Involuntary Separation, and Retirement used by Wareham et al. (2015).



The most recent research on separation and turnover rates were from Reaves (2012) and Wareham et al. (2015) using the 2003 LEMAS data and 2008 CSLLEA data. The turnover rates for the case agency are compared to the data from those two studies. Since the data from this study includes separation rates from 2010 through 2016, which is more recent than the 2003 and 2008 data, there are likely different or additional factors to explain the case agency's turnover rates, which are discussed below.

The total turnover rates from Table 86 previously presented in Chapter IV shows that the lowest turnover rate was 1.54% in 2010 and the highest rate was 11.93% in 2015. The higher turnover rates of 12.5% in 2011, 10% in the 2014, and 11.9% in 2015, may be a result of the increased number of officers hired in previous postings.

Table 105 below compares the agency's turnover rates using the same turnover indicators as Wareham et al.'s (2015) data previously discussed in Chapter II. The total turnover rate for the case agency, 8.29% is about 2.5% lower than both the total turnover rates for the 2003 LEMAS, 10.82%, and 2008 CSLLEA data, 10.76%. It is interesting that the case agency had less than half the percentage of resignations, 3.55%, compared to the 2003 LEMAS data, 7.79% and the 2008 CSLLEA data, 7.31%. The percent of normal retirements, 2.2%, was more than two times higher compared to the .92% and .90% for the 2003 and 2008 data. This could be due to some officers who reached retirement age, and some officers may have retired earlier then they planned due to changes in the pension plan in the past few years which reduced benefits for early retirement. The involuntary separations, 2.54%, was comparable to the 2003 and 2008 data.

The case agency's turnover rates presented in Table 105 are also consistent with Reaves (2012) statistics using the 2008 CSLLEA, with a turnover rate of 7.4% for state and local officers



and Stevens (2018) rates of 5% for large agencies and 5% for large agencies previously discussed in Chapter II. The total turnover rates presented in Table 105 for the case agency shows that this is consistent with the total turnover rates presented in Table 86, which are less than or about 7.0% in 2010, 2012, 2013, and 2016. There were three years where the total turnover rates were higher, 12.50% in 2011, 10.00 in 2014 and 11.93% in 2015. The average turnover rate, 8.29% is less than 1.0% higher than Reaves 7.4%.

Table 105. Agency Turnover Rates for Officers who Separated from 2010 to 2016 and the 2003 LEMAS Data and 2008 CSLLEA Data (Wareham et al., 2015) by Turnover Indicators.

Turnover Indicator	Agency Turnover Rate	2003 LEMAS Data	2008 CSLLEA Data
Resignations	3.55%	7.79%	7.31%
Retirements	2.20%	.92%	.90%
Involuntary Separations	2.54%	2.11%	2.55%
Total Turnover	8.29%	10.82%	10.76%

As discussed previously in Chapter II, the findings from Wareham et al.'s (2015) study found that geographic location, type of agency, and size of agency influenced different kinds of turnover (p. 365). As the data in Table 106 presents, there is a wide range of mean turnover rates which are dependent on geographic location, urban status, type of agency, and size of agency. The last column indicates that the mean turnover rate for the case agency, 8.29%, lies within the ranges for both the 2003 LEMAS Survey data and the 2008 CSLLEA data for region and division, urban status, and type of agency. However, the 8.29% mean turnover rate for agency size is higher than the highest percentage for both the 2003 LEMAS Survey data and the 2008 CSLLEA data.

The mean turnover rate for resignations at the case agency, 3.55%, is lower for region and division, and urban status, within the range for type of agency, and higher for agency size than the 2003 LEMAS data; and within the range for region and division, and type of agency,



lower for urban status, and just higher for agency size than the 2008 CSLLEA data. For retirements, the rate for the case agency, 2.20%, is more than two times higher for the Midwest region and division than both the 2003 and 2008 data, higher for the rates for urban status than both data sets, lies within the range for type of agency than the 2003 data, but is higher than the 2008 data, and is higher for agency size than both sets of data. The agency's mean turnover rate for involuntary separation, 2.54%, is higher than the region and division than both the 2003 and 2008 data, within the range for urban status than both data sets for type of agency, higher than the 2003 data and within the range than the 2008 data, and more than 1.0% higher for agency size than both data sets.

Overall, the agency's rates for resignations are consistent with the 2003 and 2008 data. However, there are some differences, which may be explained by the time frame covered by the agency data, 2010 through 2016. As state previously, the most recent data found for separation rates were the 2008 data. Retirements are higher but may be a result of officers reaching retirement age. Also, there were changes in the pension plan the past few years which reduced benefits for early retirement, where officers who could retire may have retired early. Involuntary separations were higher for the case agency compared to the Midwest region and division for the two data sets, consistent for urban status and type of agency, and higher for agency size. The higher rate of failed probation likely contributed to the higher rate, including officers not passing FTO training or the period while officers are on their own during the remainder of the probationary period. The total mean turnover rate of 8.29% is consistent with region and division, urban status, and type of agency, but about 2% higher for agency size. As explained previously, this may be due to officers who have reached retirement age or who retired due to the changes in the pension plan.



Table 106. Mean Turnover Rates for the Case Agency, 2003 LEMAS Survey, and the 2008 CSLLEA Data for Geographic Location, Urban Status, Type of Agency, and Size of Agency.

Type of	Midwest	Rural,	Municipal,	Medium	Case
Turnover	Region and	Suburban,	County, and	(50-99	Agency
	U.S.	and Metro	State Agencies	Officers)	Mean
	Division of	Agencies		and Large	Turnover
	Case			(100-299	rate for
	Agency			Officers)	2010-2016
				Agencies	
2003 LEMAS					
Resignations	6.1 to 9.4	4.8 to 10.6	2.2 to 8.5	2.9 to 3.1	3.55
Retirements	.3 to .9	.7 to 1.3	.9 to 2.8	1.3 to 1.8	2.20
Involuntary	1.1 to 1.4	1.4 to 2.7	.9 to 2.2	1.3 to 1.4	2.54
Separation					
Total Turnover	7.5 to 11.5	7.6 to 14.1	5.9 to 11.6	5.5 to 6.3	8.29
2008 CSLLEA					
Resignations	3.0 to 7.8	4.0 to 9.8	1.8 to 7.5	3.2 to 3.5	3.55
Retirements	.5 to 1.0	.8 to 1.2	.8 to 2.1	1.2 to 1.6	2.20
Involuntary	.5 to 2.1	1.8 to 3.6	.9 to 2.7	1.4 to 1.5	2.54
Separation					
Total Turnover	4.1 to 10.6	6.9 to 14.2	4.9 to 11.1	6.2 to 6.4	8.29

This data was compiled from data in Wareham et al.'s study (2015) and results from separation data presented in Chapter IV.

All rates are percentages.

Table 107 compares the 2008 CSLLEA data for type of separation and size of agency from Table 3 in Chapter II to the agency's percentages for types of separation. Reaves' (2012) calculations of officer separations in 2008 included 54% for resignations, 23% for non-medical retirements, 10% for dismissals, 5% for probationary rejections, and 5% for medical or disability retirements, and 4% for other, which includes disability and death (Reaves, 2012; Stevens, 2018, p. 6).

Comparing the agency data for types of separation to Reaves' (2012) data, 43.4% of the separations in the agency from 2010 through mid-2017 were resignations, which is lower than Reaves' data of 51% to 58%. The 24.5% retirements were just below Reaves' data for agencies with 100-499 officers, 26%, and higher than agencies with 25-99 officers, 18%. As explained 218



previously, this may be due to officers who have reached retirement age or who retired due to the changes in the pension plan.

Involuntary terminations were divided into four categories. The first category, 5.7% dismissals due to misconduct at the agency, was lower compared to the 9-11% in Reaves' (2012) data. The second category, 5.7% medical / disability retirements at the agency, was consistent with the 5% in Reaves' data. The third category, 5.7% in the "Other" category, was higher than the 3-4% in Reaves' data. The last category, probationary rejections, 15.0%, was about three times higher than the 5-6% in Reaves' data.

For the 8 officers who failed probation, 6 officers did not pass their FTO training and 2 officers passed their FTO training but were terminated before their probation expired due to performance or conduct issues. It is important to note that it is difficult to interpret this finding due to the small number of officers. This higher rate of failed probation may be an unintended consequence of the changes in the application process. One possible explanation for this higher rate may be that the recruiting and hiring process may impact applicants passing the academy and field training. Perhaps, there is a gap between qualifications for hiring and those in the academy and field training. Perhaps something was eliminated in the application process that provided screening of applicant's ability to be successful in the academy and field training. Referring to Table 90 in Chapter IV, also concerning is the number and percentage of female officers, 2 and 25%, and the number and percentage of officers of color, 4 and 50%, who failed probation. Recommendations include assigning a mentor to all officers hired to offer support before, during, and after the academy and FTO period. Offering information sessions and a ridealong with an officer for those interested in applying at their agency can also help in providing realistic expectations of the job. Future research could include examining the aspects



emphasized in the academy and FTO training compared to the qualifications and skills emphasized in the recruiting and hiring process.

Table 107. Type of Separation for the 2008 CSLLEA Data (Reaves, 2015) and Case Agency.

Type of Separation	Agencies of All	100-499	25-99 Officers	Agency
	Sizes	Officers		Turnover
Resignation	54%	51%	58%	43.4%
Non-medical	23%	26%	18%	24.5%
retirement				
Dismissal	10%	9%	11%	5.7%
Probationary	5%	6%	5%	15.0%
rejection				
Medical/disability	5%	5%	5%	5.7%
retirement				
Other – Disability	4%	4%	3%	5.7%
and death				

This data was compiled from data in Reaves (2012) report (p. 7). Total percentages for first two columns total 101% due to rounding.

Police Officers who Left for another Law Enforcement Agency (Lateral Moves)

Regarding lateral moves, using the 2008 CSLLEA data on officers who left the agency for another law enforcement position, about 45% left small agencies and about 24% left large agencies (Stevens, 2018; Reaves, 2012, p. 234). Table 88 presented in Chapter IV shows the number and percentage of officers who resigned – other position in bold print, 19, 35.9%. This falls in between Reaves' (2012) data, 24% of officers who left for another law enforcement position from large agencies and 45% of officers who left from small agencies. This suggests that the case agency is not losing officers to other agencies compared to other agencies in the U.S. However, as stated by respondents in the interviews, there was the perception that other agencies are "poaching" their agencies away.



Female Officers and Officers of Color Who Left Policing

Regarding women leaving policing, Fyfe and Kane (2006) examined attrition rates for female officers in the U.S. and found that female officers leave policing at higher rates than do male officers. They found that female officers were more likely to resign voluntarily than male officers (4.3 percent versus 3.0 percent, respectively) and that female officers were terminated involuntarily more often than male officers (1.2 percent versus 0.6 percent, respectively). As presented in Chapter IV, there were a total number of 49 officers who separated from the agency from 2010 through 2016. The 2017 separations were not included in these calculations because the data in 2017 did not include an entire year. Table 108 presents the turnover rates for male and female resignations and terminations by frequency, agency turnover rate, and Fyfe and Kane's turnover rates.

One significant finding was that the turnover rate for female officers who resigned, 0.68%, was 84% lower than Fyfe and Kane's (2006) rate of 4.3% and the turnover rate for male officers who resigned, 7.61%, was 154% higher than Fyfe and Kane's turnover rate of 3.0%. Another significant finding was that the turnover rate for female officers who were involuntarily terminated, 0.17%, was 86% lower than Fyfe and Kane's rate of 1.2%, and the turnover rate for male officers who were involuntarily terminated, 8.12%, was 1,253% higher than Fyfe and Kane's turnover rate of 0.6%. Contrary to Fyfe and Kane's results, the agency's results show that females are less likely to both resign voluntarily and be terminated involuntarily than male officers. These results for the agency are consistent with respondents who stated in their interviews that they did not perceive that female officers leave at higher rates than male officers. The data from this agency is more current, 2010 to 2016, than Fyfe and Kane's data, which could account for the contrast. A possible explanation for these differences is that there is a different



social climate for the agency than existed at the time of the 2006 data. Also, an agency that practices community policing and promotes being reflective of and responsive to its communities, are more welcoming and accepting of female officers. Thus, the agency is more successful at attracting and retaining females and has fewer females resign or terminated.

Table 108. Number and Percent of Agency Turnover Rates and Fyfe and Kane's (2006) Study by Female and Male Resignations and Involuntary Terminations.

Turnover Indicator	Agency	Agency	Fyfe & Kane (2006)
	Frequency	Turnover Rate	Turnover Rate
Female Resignations	4	.68%	4.3%
Male Resignations	45	7.61%	3.0%
Female Involuntary Terminations	1	.17%	1.2%
Male Involuntary Terminations	48	8.12%	.6%

Using the data on the 53 officers who separated, previously presented in Table 90 in Chapter IV, the 6 females who separated were 11.3% of the total number and the 16 officers of color who separated were 30.2% of the total number. Martin (1994) concluded that the higher rate of women leaving law enforcement was due to women experiencing a "more hostile work environment" than males do (p. 398). Stevens (2018) added that "female officers who are single parents are also confronted with different problems than males in the workplace and are treated differently than male officers by their commanders as well" (p. 235). The data for this agency showing that few females leave, does not support females experiencing a hostile work environment. As stated by several respondents in the interviews, the senior command staff made an effort to welcome females. Stevens also reports that "minorities resign at higher rates than do white male officers" (p. 235). As stated in the interviews, respondents stated that only a few officers of color leave and that they perceive that overall, their retention rate is good.



Length of Service and Separation

Regarding the relationship between length of service and attrition, Koper et al. (2002) found that two-thirds of the officers who leave both large and small agencies have 5 or less years of service (p. 47). Stevens (2012) also added that "officers with intermediate lengths of service (i.e., 6 to 14 years of service) are the least likely to leave both large and small agencies" (pp. 235-236). The percentage of officers who separated from the case agency with five or less years, 60.4%, is the total of the first three rows in Table 92 presented in Chapter IV and is about 6% lower than the two-thirds found by Koper et al. (2002) cited in Chapter II. The percentage of officers who separated from the agency with more than 5 years of service, 39.6%, is about 6% higher than Koper et al.'s results of one-third, which is likely due to the increase of retirements observed in the agency.

Important Factors in Recruiting, Hiring, and Retaining Diversity

The changes in the application process were previously discussed under the applicant data, which is one important factor in increasing diversity within the agency. Other factors that respondents identified as important in recruiting, hiring, and retaining diversity are discussed in this section.

Importance of the Agency being Reflective of its Communities

The interview data indicates that a representative police agency has many benefits. All respondents stated in response to different questions, including one question asking directly about this theme, that it was important for their agency to be reflective of its community.

Respondents also discussed the importance of targeted recruitment for specific groups, including females, Native Americans, Blacks and African Americans, Asians, and Latinos and Hispanics.



Several respondents stated that reaching out to diverse communities and females educates prospective applicants on what the police do.

Respondents also indicated that being representative of their communities helps educate communities of color on the role and function of their agency and increases support, trust, and credibility from the communities. This is crucial because of high profile critical incidents involving the police and communities of color. The priority of increasing diversity within their agency was also evident in the setting of the hiring and composition goals discussed previously. The data suggest that being representative of its communities helps in being responsive to its communities, which is discussed next.

The Relationship Between the Agency being Responsive to its Communities Through a Commitment to Community Engagement and Recruiting, Hiring, and Retaining Diversity

All respondents indicated that it is important for their agency to be responsive to its communities through community policing. Interview data indicated that there was a reciprocal relationship between community policing and recruiting, hiring, and retaining female officers, officers of color, and officers who are fluent in languages other than English. As previously presented in Chapter V, respondents provided examples and descriptions on community engagement being a priority with their chief, and how attending community events has helped their recruiting efforts.

The influence of community policing impacting recruiting and hiring was also evident in the descriptions provided by respondents for the desired qualities, skills, and traits that applicants should possess. It is important to note that these desired skills are consistent with the traits and skills necessary for officers to engage in community-oriented and problem-oriented policing. All respondents indicated that personal character and traits were as important or more important than



applicants should possess to successfully perform as police officers at their agency. The list includes honesty, integrity, good character, being innovative, a community service mindset, being a good communicator, a clean background, problem solving, moral courage, physical courage, ability to stand up to stress, willing to learn, trainable, and coachable. Respondents added, "anyone can become a police officer, and get trained, but it takes a passion, a calling, to be able to be in a field of service and protect and serve", "we can take smart people and people with integrity and train the skills to be a good cop but it's hard to train integrity, character, and community service mindset", and we want people "who truly want to help out people that are out there to make somebody else's day better". It was interesting that one respondent pointed out that community policing increases recruitment and retention because their agency is doing much more community engagement than they use, and this may appeal to officers who want to practice community policing. Thus, it would be a reason to stay with this agency.

All the respondents stated communication skills were also important. Respondents provided examples to illustrate this, including the necessity for applicants to effectively speak to other people in general, calm people who are upset or suicidal, read body language, speak languages other than English, and "communicate on a variety of different levels, face-to-face, good body language, but also have high levels of linguistics and cultural competency". In addition to the influence of community policing on recruitment and hiring diverse officers, recruitment also influences community policing. Three respondents illustrated how recruiting efforts increase community policing by educating communities about their role and function and helps build relationships with their communities by attending schools, job fairs, and community events. Respondents also emphasized the importance of others to see officers in a positive light.



The Impact of External Factors on the Applicant Pool

All respondents stated there were factors external to the agency that decreased the number of applicants for their agency and other agencies in the U.S. This was evidenced by the decrease in the total number of applicants in the 2014 and 2016 postings, after the 25% increase to 640 applicants in the 2013 posting from 511 in the 2012 posting, there was a 19% decrease to 516 applicants in the 2014 posting, and an additional decrease of 39% to 317 applicants in the 2016 posting. External demands from "community or special interest groups within the community can pressure law enforcement agencies to diversify their workforce" affirmed by Schuck (2014, p. 59) is illustrated by respondents identifying the negative portrayal of police by the news, social media, and protest groups, such as Black Lives Matter as decreasing the number of applicants for police agencies in general, as well as for their agency. This is consistent with five respondents who indicated the applicant pool for police officers is down for their agency and elsewhere in the U.S., including three respondents who stated that "the hiring pool is not as deep as it was just a few years ago", the quality of candidates has decreased, and it was "tough to recruit women everywhere".

Respondents indicated that the time it takes to get applicants through the application process can result in losing applicants who accept the first offer. This is illustrated by one respondent who stated, "we get them into the background phase, we then realize that they're being backgrounded at multiple agencies at the same time, so it becomes a factor that they'll take the first job they get offered" and if your process is not quick enough, you will lose them to another agency. This agency is competing with other police agencies for candidates. There are about 65 other agencies in general proximity to the case agency that applicants may also apply.



All responses discussed external factors that influence officer separation and retention in their agency. When the respondents were asked if they knew the reasons why officers left their agency, all respondents stated that there were a variety of reasons and all respondents stated that some of the officers left because there were more opportunities at other agencies. Other reasons provided include for family or personal reasons, for example, to be closer to home, and deciding they do not want to do this job after being hired. One factor that negatively affects officer retention was that several respondents indicated they lost officers who were recruited by other agencies, with two respondents using the term "poaching" to describe what they perceived other agencies taking their officers.

The Influence of Organizational Factors on Recruiting, Hiring, and Retaining Diversity

All respondents provided examples and descriptions of how organizational factors have increased recruiting, hiring, and retention. All respondents stated in their interviews that their agency has great pay, benefits, overtime, and/or pension and these were factors for increasing recruiting, hiring, and retention. All respondents asserted that the changes in the application process have increased the number of female officers and officers of color within their agency. Mentors within the agency were mentioned by two respondents, with one respondent who stated that "mentors in the organization become a part of what keeps people here" and another respondent who stated that is important to retain officers because they are needed to mentor other officers. In response to the question, "What else do you think your agency can do to attract diverse qualified candidates?", respondents suggested signing bonuses, having a full-time person committed to coordinating recruiting and community engagement activities, and using social media to attract millennials.



All respondents stated that their union, another organizational factor, supports recruiting and hiring female officers and officers of color, that their union has worked hard for good pay, benefits, and contracts to attract candidates, and that their union has a good working relationship with management. Respondents indicated that their union contract impacts promotional practices but not hiring practices. This is important because it appears to contribute to the chief taking an active role in the hiring process, particularly implementing changes in the application process. This factor also appears to work in combination with the other factors discussed previously, the state regulatory board that sets the minimum selection standards, their Human Resources Office that sets the hiring practices, and their agency not being under civil service rules. Also emphasized was the role of the union in ensuring that management communicates the hiring process and the rationale behind the hiring process with officers. Respondents also stated that even though they have heard negative comments from officers regarding the hiring process, there have been few formal complaints with the union.

All respondents perceived that they have a low turnover rate overall. Of the officers who left, respondents perceived that most of them were White males and just a few were female officers and officers of color. One respondent offered reasons why women might leave including shift work, daycare, or are intimidated because of their size. The responses that few women and officers leave are consistent with the results of the separation data presented in chapter IV.

All respondents stated that even though officers leave for more opportunities in other agencies, most of the respondents stated there are opportunities here that help in retention.

Examples included additional assignments, instructor positions, career enrichment opportunities, sending officers to different types of training, offering more overtime, and offering some



flexibility on work schedules, when feasible. One respondent added that "some people do not realize what a privilege it is to work here".

Respondents provided suggestions that their agency could use to retain officers, including, offering ride-a-longs for applicants so they understand what the job entails, give officers more opportunities for different or additional assignments, help with child care issues, offer signing bonuses, and having a full-time person committed to coordinating recruiting and community engagement activities. One respondent stated that their agency can offer more opportunities only to a point because they cannot pull officers from core assignments. One respondent emphasized the need to do more with mentorship programs, including asking current officers to volunteer as mentors for new officers before they start the academy training, and for veteran officers to help them throughout their careers, including help in getting promoted.

The Perception that Some Diverse Applicants Hired were Not Qualified

The interview data indicated that respondents have heard comments from other officers that some officers hired at this agency were not qualified. These comments included: administration can pick whoever they want despite the order of applicants after the first interview, some officers have left because they were not happy with the processes for hiring, the standards were lowered, some applicants hired at this agency could not get hired at other agencies, and some people get more chances in passing the academy training and field training, It is not known where these perceptions originate because all applicants have to meet minimum selection standards set by the state regulatory board to become a police officer. Recruiting and hiring diverse officers is not in opposition to hiring qualified and competent officers. Instead, the pool of qualified and competent police officer applicants is enhanced.



The role and function of policing has changed and expanded over the past several decades. These expanded and additional tasks police officers are expected to perform include: a broader range of tasks, the demands of policing in a pluralistic society due to the changing population demographics, the complex skill set required for officers to engage in community-oriented and problem-oriented policing, and changes in the societal and political environment calling for more accountability, trust, and legitimacy. This urges police agencies to examine their hiring requirements to ensure they are predictive of the tasks officers perform. Changes in the application process that resulted in increasing the diversity within the agency suggest that the traditional application process for police officers may unnecessarily disqualify applicants from underrepresented groups.

The questions in the first interview are different than the questions asked in the second interview, as one respondent described the first interview as assessing whether applicants possess the qualities and skills to be a police officer in general, while the second interview assesses whether applicants possess the qualities and skills to be a police officer at this agency. This explains what respondents described as "the list starts over". The second interview addresses the skills necessary for community-oriented and problem-oriented policing which is a part of the agency's two priorities, being reflective of and responsive to their communities. These different and/or additional skills sought in applicants may account for some officers not understanding what the hiring process is screening for and may explain why some officers perceive that the agency has lowered the standards or hires officers that do not get hired by other agencies.

The perception that some people get more chances in passing the academy and field training may be a result of applicants having difficulty with the technical skills portion of the academy and FTO process. It is not only important that the academy and field training reflect



the skills and qualifications sought in the application process, but there is an additional need of working with officers during the academy and field training to develop their technical skills.

Offering mentors can also help in understanding expectations and support during the first year.

The Influence of Administrative Discretion on Decisions made Regarding Recruiting and Hiring Practices

Previously discussed were the factors that influenced the chief taking an active role in the hiring, process, especially in implementing changes to the steps in the application process that resulted in increasing the representation of female officers, officers of color, and officers who are fluent in languages other than English. Besides the minimum selection standards for police officer set by the state regulatory board, all respondents discussed that their Human Resources Office sets the standards, rules, and regulations for hiring practices at the agency.

An important finding from the interview data is that the chief takes an active role in the hiring process. The discretion allowing the chief to make decisions regarding recruiting, hiring, and retention practices is a key element in changing the traditional application process to hire more female officers and officers of color. It is imperative to understand what factors contributed to the flexibility and authority by the chief to take such an active role in the hiring process. Respondents discussed that there are certain requirements that Human Resources mandate, however their agency's Human Resources Office allows flexibility in recruiting and hiring. One important factor that contributed to the chief taking an active role in the hiring process is that their Human Resources Office allows the agency flexibility regarding decisions made on recruiting and hiring practices as long as the agency can articulate the reasons why. Examples provided by respondents included adding desired qualification for fluency in languages other than English, using supplemental questions to reduce the number of applicants,



moving the background investigations to before the second interview, eliminating the written test, making the interviews pass/fail instead of using a ranking, having the backgrounds conducted by officers within the agency instead of contracting with investigators outside the agency, and selecting people from diverse backgrounds to be on the oral boards.

In addition to the Human Resources Office rules that govern the hiring process, a second important factor is that their union influences promotional practices through the Human Resources office, however, their union does not influence hiring practices. A third important factor is that their agency's employment practices are not under civil service rules. This is interesting because most police agencies in the U.S. are under civil service. Respondents described that instead of having civil service which selects the top eligible candidates in ranked order, their agency establishes a pool of eligible candidates, so candidates do not have to be taken in order. This allows for more flexibility in hiring female officers and officers of color.

The Influence of Affirmative Action Programs

Prior research on the influence of Affirmative Action programs cited in Chapter II found African American representation in police agencies in the U.S. has slowly increased since early efforts to encourage social equity. The NCWP (2003b) and Schuck (2014) found that consent decrees, hiring quotas, and court-orders led to greater representation for women. Hochstedler (1984), Martin (1991), Selden (1997), and Zhao and Lovrich (1998), also found that representation of officers of color has increased as a result of efforts to encourage social equity, hiring quotas, and affirmative action policies and programs. Walker (1985) found court-ordered quotas was significant in increasing the hiring of Black officers, and Lewis (1989) found that consent decrees was one of the most powerful variables in increasing Black percentages among Black officers. Zhao and Lovrich (1998) found that one of key elements of Affirmative Action



programs is for agencies to specify "goals and timetables keyed to pools of qualified applicants" (p. 269). Walker (1985) also found that the progress in Affirmative Action may be influenced by the "quality of leadership in police departments and/or elected or appointed officials" if the political power translates a commitment to Affirmative Action.

The agency has a statement in their job postings that their agency is an Equal Opportunity employer and has a formal Affirmative Action plan. This agency is not under any consent decrees, nor do they have quotas, so the high numbers and percentages of female officers and officers of color are impressive. The results from this study are consistent with prior literature discussed in Chapter II that found Affirmative Action programs increase representative bureaucracy. The data also suggests that the results are likely in combination with other factors discussed, including, commitment from the Human Resources Office and the command staff to increase their agency's diversity and be responsive to their communities.

The Influence of Structure of Government/Autonomous Personnel Bureaus/Civil Service

The literature in Chapter II also identified structure or type of government by itself and in combination with autonomy of its personnel bureau in influencing compliance with Affirmative Action programs and employment of people of color in the U.S. (Stein, 1986). The respondents stated that their command staff prioritizes being reflective of and responsive to their communities, which includes diversifying their agency and engaging in community-oriented and problem-oriented policing. This also demonstrates that the agency is responsive to external political and social forces and the make-up of its communities, and in gaining the trust and cooperation from the different communities it serves. The structure of government also influences the structure of human resources. Stein described civil service with the aim of substituting political hiring with selection and merit, however, she noted that the restrictive



qualifications that define the merit principle "prevent many minority applicants from obtaining positions with local government" (p. 697). She noted that professional autonomous bureaus were resistant to direction from the outside and the presence of a civil service commission had a significant negative relationship on employment of people of color in large city work forces (p. 702). The results of the data support Stein's findings, that their Human Resources Office not being under civil service rules, was a factor in increasing the agency's representation of female officers and officers of color. Not being under civil service rules is different than for most other police agencies, especially large agencies. This factor was important in the chief's ability to implement changes in the application process that resulted in increasing the numbers and percentages of female officers and officers of color.

The Influence of Professional Associations

All seven respondents stated that there were benefits from several organizations and associations, including the different racial/ethnic police organizations and associations, organizations and associations for women police and chief law enforcement officers, and athletic teams. The following are examples cited by respondents of how these organizations have assisted their recruiting. Examples include the agency being promoted at the various conferences, the agency sending officers to various conferences, hearing applicants state in their interview that a particular organization or association told them to apply and/or helped them prepare for the interview, officers who belong to associations tell other members to apply to their agency, and openings at their agency are posted in the various publications of the organizations and associations. This is consistent with the purpose and results of two professional associations. cited in Chapter II. The National Black Police Association (NBPA) was started to bring together African American police associations with the goal of improving police agencies and



relationships between communities of color and the police. The National Organization of Black Law Enforcement Executives (NOBLE) was formed to bring Black police executives together to address crime in urban areas (H. Greene, 2003). H. Greene described how such organizations have brought about change, "these organizations address numerous issues confronting officers and black citizens, including brutality, crime, discrimination, police-community relations, and recruitment" (p. 212). H. Greene also described how NOBLE has initiated efforts in increasing recruitment of people of color and promoting career fairs. As discussed above, respondents provided examples of how several associations positively impact their recruiting and hiring diversity which adds to the literature regarding how professional associations and organizations can increase recruiting and hiring female officers and officers of color.

The Influence of its Union

There were numerous studies cited in Chapter II that showed mixed results regarding the influence of unions on the representation of female officers and officers of color. Schuck (2014) found that collective bargaining had a direct negative impact on gender diversity. Walker (1985) stated that police unions opposed ways to improve police-community relations and have represented White officers who challenged Affirmative Action programs and plans in courts. Riccucci (1986) found that the presence of a union had a limited effect on female employment within municipal police departments. Kim and Mengistu (1994) found in their study that collective bargaining did not significantly influence female and Asian police office employment and that police unions did not promote hiring Black police officers but did Hispanic officers (p. 171). Warner et al. (1989) did not find a correlation between collective bargaining and female representation in their study. Using a more diverse and larger sample of agencies, Sass and Troyer (1999) found less female representation with unionization in the early 1980s, but not for



the late 1980s and early 1990s. Morabito and Shelley (2015) found that the presence of collective bargaining was significantly and positively associated with increased female representation, but not for officers of color (pp. 341, 345 & 345).

All respondents stated their union supports recruiting and hiring female officers and officers of color. The respondents indicated that their union contract does not influence hiring practices which is another indicator that their union supports hiring diversity. Respondents also stated that their union has worked hard for good pay, benefits, and contracts to attract and retain officers and that it was important for the command staff and management to communicate the rationale behind the hiring process with officers. The results of the data for this agency support some of the findings that unions increase female representation. However, the results for this agency are not consistent with the prior literature where unions oppose police-community relations and hiring women and officers of color or challenge Affirmative Action programs.

The Active Role of Leadership in Diversifying their Agency

As discussed in Chapter V, all respondents indicated that the change in command staff in 2012 and the active role of the chief in the hiring process are responsible for increasing diversity within their agency. Leadership includes the command staff and management within the agency. The current command staff set goals in hiring, prioritized hiring for diversity, targeted specific groups in recruiting and hiring, and implemented changes in the application process to meet these goals.

The following are characteristics important in the command staff and management in recruiting, hiring, and retaining diversity that were identified by respondents during the interviews: engaging in community-oriented and problem-oriented policing, prioritizing being reflective of and responsive to the communities they serve, the relationship between community



engagement and recruiting, hiring, and retaining diversity, and the discretion used by the command staff regarding hiring decisions. Warner et al. (1989) found that the attitudes of hiring agents influenced equal opportunity employment. The results from the interview data support this finding, where several respondents stated that the chief takes an active role in the hiring process and their agency prioritizes being reflective its communities by implementing changes in the application process and being responsive to their communities by committing to communityoriented and problem-oriented policing throughout the agency. Eisenberg, Kent, and Wall (1973) found that the chief of police was identified as "most frequently designated as having the major responsibilities for establishing personnel policies for sworn police personnel" (p. 13). Zhao and Lovrich (1998) found that a significant variable in predicting representation of African American officers in U.S. municipal police agencies is that the police chief has "significant power in the hiring procedure" (p. 274). H. Greene (2003) asserts that police chiefs can influence policing because they have the power to set priorities within their agencies, as well as have an impact on policy, which includes restricting the use of force, recruiting officers of color and practicing community-oriented policing (p. 212). Many Black police chiefs and Blacks in administrative and command positions espouse community-oriented policing "because it provides an opportunity to remove the historical barriers between police and the black community" (p. 212).

This agency has one member of command staff who is a person of color, which is a contributing factor in increasing diversity within the agency. This is consistent with the literature cited in Chapter II regarding role acceptance of administrators, "administrators who perceive themselves as possessing significant discretion and who assume the role of minority representative in their agencies are more likely to enact policy outcomes that favor minority



interests" (Sowa & Selden, 2003, p. 700). This also supports Sowa and Selden's conclusion that this is one mechanism that strengthens the relationship between passive and active representation. The influence on decisions that increased the representation of female officers and officers of color within the agency included prioritizing and setting goals to be reflective of their communities. This includes implementing changes in their application process and being responsive to their communities by engaging in community-oriented and problem-oriented policing. This is also consistent with Bradley and Kellough (2008), who predicted that public administrators of color "will have similar attitudes to minority citizens on issues of critical import and relevance to those citizens, and those attitudes, in turn, will influence policy decisions" (p. 697). They found that African American administrators are more likely than White administrators to adopt an African American representative role. Selden (1997) and Bradbury and Kellough (2008) also point out that even though bureaucrats of color were a significant predictor of the minority representative role, White bureaucrats can also take the role of minority representative.

The chief did not and cannot diversify the agency without support from throughout the agency, including the command staff, management, supervisors, and officers. The interview data suggest that the chief possesses important leadership qualities. This includes leading by example, which was illustrated by getting out in the community, engaging in community-oriented and problem-oriented policing, demonstrating the benefits of a reflective agency, and gaining support from the rest of the command staff, management, supervisors and officers regarding the importance of diversifying the agency.

Zhao and Lovrich (1998), add that any changes in personnel policies are due to several factors, including political, organizational and legal factors (p. 267). In addition to the police



chief being important, the administration under the chief is important in diversifying police agencies. The findings from the interviews are consistent with Van der Lippe et al. (2004) that found that top managers are instrumental in "breaking down the barriers which women encounter" (p. 398). The findings are also consistent with Gustafson (2013) that found in municipal police agencies in the U.S., the presence of people of color in police leadership positions was one of the most significant factors in affecting diversity in line officers.

Limitations of this Study

There are six limitations of the study. The first limitation is that because there is only one case, there is a lack of comparison (Bachman & Schutt, 2014, p. 307). This case study focuses on one law enforcement agency in the Midwest. Purposive sampling was used to select the case agency because it is important to select an agency that intentionally recruits and hires female officers and officers of color and is recognized for a commitment to hiring for diversity. Thus, this agency may not be representative of other police agencies and there is a lack of external generalizability to other agencies, settings, types of agencies, and geographic locations. This case is offered as an example in how other law enforcement agencies can increase the representation of female officers and officers of color. A second limitation is that it is hard to test for causal explanation with one setting because any explanations provided will be tailored to fit the particular setting and case (p. 308). A third limitation is that the small frequencies of female applicants and applicants of color may limit conclusions drawn from the data. The fourth limitation is that the interview data results obtained from seven agency personnel are limited in scope by including mostly the views of command staff, management, supervisors, and a limited viewpoint from officers at the ground level of the agency. The fifth limitation is that this study



does not include the perceptions from officers who separated from the agency. The sixth limitation is possible response bias in the interview questions.

Recommendations for Future Research

Purposive sampling was used to select the case for this study because it was important to select an agency that intentionally recruits and hires diversity. Thus, as previously stated, the case selected may not be representative of other law enforcement agencies. Since this study was a case study with only one agency, the results cannot be generalized to other agencies, including types of agencies, other geographic locations, and size of agencies. Suggestions for future research include examining the composition, application, and separation data in other agencies and geographic locations to increase external validity.

Other suggestions for future research include surveying officers on their experiences in recruiting, hiring, and retention. It would be useful to conduct a longitudinal study to assess whether female officers and officers of color are promoted in the future. In addition, it would be useful to assess whether the representation of female officers and officers of color in supervisor, management, and command staff positions impact recruiting, hiring, and retention of female officers and officers of color. Future research could also include a longitudinal study to assess the separation and retention of female officers and officers of color over several years. Officers who have separated from this agency and other agencies also could be surveyed on why they left and offer suggestions for retaining officers. Also, future research could include examining the aspects emphasized in the academy and FTO training compared to the qualifications and skills emphasized in the recruiting and hiring process.

It would also be useful to survey citizens on their attitudes towards the police to assess how a reflective agency is more responsive to the community and increases trust, legitimacy, and



cooperation from its communities. Officers could be surveyed to determine if officer attitudes towards the community have changed. Future research could also include what aspects of the police subculture are influenced by increasing diversity within an agency and what aspects of police subculture influence recruiting, hiring, and retaining female officers and officers of color.

Conclusion

This case study has presented how the agency has achieved success by taking concrete steps to increase its representation of female officers, officers of color, and officers who are fluent in languages other than English. This was accomplished by setting composition and hiring goals, targeting recruiting efforts, updating their recruiting materials and job postings to reflect a commitment to diversity and community policing, implementing changes in the application process that differ from the traditional selection process, diversifying the make-up of their interview panels, and welcoming and respecting their diverse officers. This case study serves as an exemplary model for other police agencies to increase diversity within their agencies.



APPENDIX



APPENDIX A

Interview Guide

Create Context: The purpose of this research is to describe how your agency recruits, hires, and retains diversity for your full-time police officers. I would like to ask you about this and your role in hiring, recruiting, and retention processes. The first part of the interview will ask you questions about recruiting and hiring and the second part of the interview will ask you questions about retention.

The following are guiding questions for the discussion on recruiting and hiring:

- o It seems to me that there are some people or positions that are important in setting and realizing diversity goals for your agency. Could you identify people or positions that are important in terms of recruiting and hiring diversity among your agency's full-time officers? Why are they important?
- What is your primary role and responsibility in recruiting and hiring?
- Tell me about your recruiting and hiring process. Has your agency identified goals in achieving diversity in recruiting and hiring your full-time officers? If yes, what are these goals? What is your agency doing to achieve these goals?
 - What groups do you target in recruiting?
- What qualities, skills, abilities, and traits are you looking for in your full-time police officer applicants?
- What are the steps in your selection process?



- What type of human resource structure or system do you have for your recruiting and hiring?
 Please describe this structure and its goals. How has it impacted recruiting and hiring female officers and officers of color within your agency?
- O Do you think it is important for your agency to reflect community characteristics? Why or why not?
- What factors do you think impact your agency's recruitment and hiring practices in achieving diversity?
 - What professional associations have affected your agency in recruiting and hiring diversity? Please describe how.
 - Does your union support or hinder recruiting female officers and officers of color? Have there been any challenges to recruiting or hiring practices?
- What else do you think your agency can do to attract qualified applicants for full-time officers? Female officers? Officers of color?

The following are guiding questions for the discussion on retention:

- For the officers that have separated from your agency, do you know the reasons why they
 have left? If yes, please tell me about these reasons.
- Do you think female officers and officers of color separate at different rates than White males?
- o How do you think your agency is doing on retaining your full-time officers?
- What does your agency do to retain full-time officers in general? For female officers and officers of color? What else could your agency do to retain officers?
- What external or internal factors encourage or hinder retention of your officers? For female officers and officers of color?



 Closing Question: Is there anything else you would like to share about your agency's recruiting, hiring, and retention processes?



REFERENCES

- Bachman, R., & Schutt, R. K. (2014). *The practice of research in criminology and criminal justice* (5th ed.). Los Angeles, CA: Sage Publications.
- Baxter, P., & Jack, S. (2008). Qualitative case study methodology: Study design and implementation for novice researchers. *The Qualitative Report*, *13*(4), 544-559. Retrieved from http://www.nova.edu/ssss/QR/QR13-4/baxter.pdf
- Berg, B. L., & Lune, H. (2012). *Qualitative research methods for the social sciences* (8th ed.).

 Boston, MA: Pearson Education.
- Bloch, P. B., & Anderson, D. (1974). *Policewomen on patrol: Final report*. Washington, DC: The Police Foundation.
- Bradbury, M. D., & Kellough, J. E. (2008). Representative bureaucracy: Exploring the potential for active representation in local government. *Journal of Public Administration Research* and Theory, 18(4), 697-714. http://doi.org/10.1093/jopart/mum033
- Brunson, R. K., & Gau, J. M. (2015). Officer race versus macro-level context: A test of competing hypotheses about Black citizens' experiences with and perceptions of Black police officers. *Crime & Delinquency*, *61*(2), 213-242. http://doi.org/10.1177/0011128711398027



- Cochran, J. C., & Warren, P. Y. (2012). Racial, ethnic, and gender differences in perceptions of the police: The salience of officer race within the context of racial profiling. *Journal of Contemporary Criminal Justice*, 28(2), 206-227.
 http://doi.org/10.1177/1043986211425726
- Cordner, G., & Cordner, A. (2011). Stuck on a plateau? Obstacles to recruitment, selection, and retention of women police. *Police Quarterly*, *14*(3), 207-226. http://doi.org/10.1177/1098611111413990
- Crank, J. P. (2004). *Understanding police culture* (2nd ed.). Maryland Heights, MO: Anderson Publishing, Lexis Nexis Group.
- Della Porta, D., & Keating, M. (Eds.). (2008). *Approaches and methodologies in the social sciences: A pluralist perspective*. New York, NY: Cambridge University Press.
- Dempsey, J. S., & Forst, L. S. (2013). *Police* (2nd ed.). Clifton Park, IL: Delmar, Cengage Learning.
- Dempsey, J. S., & Forst, L. S. (2016). *An introduction to policing* (8th ed.). Boston, MA: Cengage Learning.
- Denzin, N. K. (1970). *The research act: A theoretical introduction to sociological methods*. Chicago, IL: Aldine Publishing.
- Dolan, J. (2000). The Senior Executive Service: Gender, attitudes, and representative bureaucracy. *Journal of Public Administration Research and Theory: J-PART*, 10(3), 513-529. Retrieved from JSTOR database.
- Eisenberg, T., Kent, D. A., & Wall, C. R. (1973). *Police personnel practices in state and local governments*. Washington, DC: Police Foundation.



- Eisinger, P. K. (1982). Black employment in municipal jobs: The impact of Black political power. *The American Political Science Review*, 76(2), 380-392. http://doi.org/10.2307/1961117
- Fyfe, J. J., & Kane, R. (2006, September). *Bad cops: A study of career-ending misconduct*among New York City police officers (NCJRS Publication No. 215795). Retrieved from https://www.ncjrs.gov/pdffiles1/nij/grants/215795.pdf
- Gilliard-Matthews, S., Kowalski, B. R., & Lundman, R. J. (2008). Officer race and citizenreported traffic ticket decisions by police in 1999 and 2002. *Police Quarterly*, 11(2), 202-219. http://doi.10.1177/1098611107309657
- Goode, S. J., & Baldwin, J. N. (2005). Predictors of African American representation in municipal government. *Review of Public Personnel Administration*, 25(1), 29-55. http://doi.org/10.1177/0734371X04270984
- Greene, H. T. (2003). Do African American police make a difference? In M. D. Free, Jr. (Ed.),

 *Racial issues in criminal justice: The case of African Americans (pp. 207-220). Westport,

 CT: Praeger.
- Greene, J. R. (2000). Community policing in America: Changing the nature, structure, and function of the police. In National Institute of Justice (Ed.), *Policies, processes, and decisions of the criminal justice system* (Vol. 3) (pp. 299-370). Department of Justice. Retrieved from https://www.ncjrs.gov/criminal_justice2000/vol_3/03g.pdf
- Grissom, J. A., Nicholson-Crotty, J., & Nicholson-Crotty, S. (2009). Race, region, and representative bureaucracy. *Public Administration Review*, 69(5), 911-919. Retrieved from JSTOR database.



- Groeneveld, S., & Van de Walle, S. (2010). A contingency approach to representative bureaucracy: Power, equal opportunities and diversity. *International Review of Administrative Sciences*, 76(2), 239-258. http://doi.org/10.1177/0020852309365670
- Gustafson, J. (2013). Diversity in municipal police agencies: A national examination of minority hiring and promotion. *Policing: An International Journal of Police Strategies & Management*, 36(4), 719-736. http://doi.org/10.1108/PIJPSM-01-2013-0005
- Hagan, F. E. (2012). Essentials of research methods in criminal justice and criminology (3rd ed.). Upper Saddle River, NJ: Pearson Education.
- Hindera, J. J. (1993). Representative bureaucracy: Imprimis evidence of active representation in the EEOC district offices. *Social Science Quarterly*, 74(1), 95-108. Retrieved from JSTOR database.
- Hindera, J. J., & Young, C. D. (1998). Representative bureaucracy: The theoretical implications of statistical interaction. *Political Research Quarterly*, *51*(3), 655-671. http://doi.org/10.2307/3088043
- Hochstedler, E., Regoli, R. M., & Poole, E. D. (1984). Changing the guard in American cities: A current empirical assessment of integration in twenty municipal police departments.

 *Criminal Justice Review, 9(1), 8-14. http://doi.org/10.1177/073401688400900102
- Hurwitz, J., & Peffley, M. (2005). Explaining the great racial divide: Perceptions of fairness in the U.S. criminal justice system. *Journal of Politics*, 67(3), 762-783. http://doi.org/10.1111/j.1468-2508.2005.00338.x
- Independent Commission on the Los Angeles Police Department. (1991). Report of the

 Independent Commission on the Los Angeles Police Department. Los Angeles, CA:

 Author.



- Ingraham, P. W. (1995). *The foundation of merit: Public service in American democracy*.

 Baltimore, MD: The Johns Hopkins University Press.
- Institute for Intergovernmental Research. (2015). After-action assessment of the police response to the August 2014 demonstrations in Ferguson, Missouri. Washington, DC: Office of Community Oriented Policing Services Critical Response Initiative.
- Jordan, W. T., Fridell, L., Faggiani, D., & Kubu, B. (2009). Attracting females and racial/ethnic minorities to law enforcement. *Journal of Criminal Justice*, *37*(4), 333-341. http://doi.org/10.1016/j.jcrimjus.2009.06.001
- Kellough, J. E. (1990). Integration in the public workplace: Determinants of minority and female employment in federal agencies. *Public Administration Review*, *50*(5), 557-566. http://doi.org/10.2307/976786
- Kelly, R. M. (1998). An inclusive democratic polity, representative bureaucracies, and the New Public Management. *Public Administration Review*, 58(3), 201-208. http://doi.org/10.2307/976560
- Kennedy, B. (2014). Unraveling representative bureaucracy: A systematic analysis of the literature. *Administration & Society*, 46(4), 395-421.
 http://doi.org/10.1177/0095399712459724
- Kim, P. S., & Mengistu, B. (1994). Women and minorities in the work force of law-enforcement agencies. *The American Review of Public Administration*, 24(2), 161-179. http://doi.org/10.1177/027507409402400203
- Koper, C. S. (2004). *Hiring and keeping police officers* (NCJRS Publication No. 202289). Washington DC: National Institute of Justice.



- Koper, C. S., Maguire, E. R., & Moore, G. E. (2002, March). *Hiring and retention issues in police agencies: Readings on the determinants of police strength, hiring and retention of officers, and the federal COPS program* (NCJRS Publication No. 193428). Washington DC: U.S. Department of Justice. Retrieved from https://www.ncjrs.gov/pdffiles/193428.pdf
- Langton, L. (2010, June). *Women in law enforcement, 1987-2008* (NCJRS Publication No. 230521). U.S. Department of Justice, Office of Justice Programs, Bureau of Justice Statistics. Retrieved from http://www.bjs.gov/content/pub/pdf/wle8708.pdf
- Lanier, M. M., & Briggs, L. T. (2014). Research methods in criminal justice and criminology: A mixed methods approach. New York, NY: Oxford University Press.
- Lasley, J. R., Larson, J., Kelso, C., & Brown, G. C. (2011). Assessing the long-term effects of officer race on police attitudes towards the community: A case for representative bureaucracy theory. *Police Practice & Research*, *12*(6), 474-491. http://doi.org/10.1080/15614263.2011.589567
- Legewie, J., & Fagan, J. (2016, May). *Group threat, police officer diversity and the deadly use of police force* (Paper No. 14-512). Columbia Law School Public Law & Legal Theory Working Paper Group. Retrieved from Social Science Research Network website: http://ssrn.com/abstract=2778692
- Levitan, D. M. (1946). The responsibility of administrative officials in a democratic society. *Political Science Quarterly, 61*(4), 562-598. Retrieved from JSTOR database.
- Lewis, W. G. (1989). Toward representative bureaucracy: Blacks in city police organizations, 1975-1985. *Public Administration Review*, 49(3), 257-268. http://doi.org/10.2307/977008



- Lim, H-H. (2006). Representative bureaucracy: Rethinking substantive effects and active representation. *Public Administration Review*, 66(2), 193-204. Retrieved from JSTOR database.
- Lipsky, M. (1980). Street-level bureaucracy: Dilemmas of the individual in public services.

 New York, NY: Russell Sage Foundation.
- Long, N. E. (1952). Bureaucracy and constitutionalism. *The American Political Science Review*, 46(3), 808-818. Retrieved from JSTOR database.
- Mansbridge, J. (1999). Should Blacks represent Blacks and women represent women? A contingent 'yes.'. *The Journal of Politics*, *61*(3), 628-657. Retrieved from Academic Search Premier database.
- Martin, S. E. (1993). Female officers on the move: A status report on women in policing. In R. Dunham and G. Alpert (Eds.), *Critical issues in policing: Contemporary readings* (2nd ed.) (pp. 327-347). Prospect Heights, IL: Waveland Press.
- Martin, S. E. (1994). Outside within the station house: The impact of race and gender on Black women police. *Social Problems*, *41*(3), 383-400. Retrieved from JSTOR database.
- Mason, P. L. (Ed.). (2013). Racial profiling and biased policing. In *Encyclopedia of Race and Racism* (2nd ed.) (pp. 401-406). New York, NY: Macmillan Reference USA.
- Mastrofski, S. D., Worden, R. E., & Snipes, J. B. (1995). Law enforcement in a time of community policing. *Criminology*, *33*(4), 539-563. Retrieved from HeinOnline database.
- Matthies, D. F., Keller, K. M., & Lim, N. (2012). *Identifying barriers to diversity in law*enforcement agencies (Occasional paper). Rand Center on Quality Policing. Retrieved from http://www.rand.org/content/dam/rand/pubs/ occasional_papers/2012/

 RAND_OP370.pdf



- Meier, K. J., & Bohte, J. (2001). Structure and discretion: Missing links in representative bureaucracy. *Journal of Public Administration Research and Theory*, 11(4), 455-470. Retrieved from JSTOR database.
- Meier, K. J., & Nicholson-Crotty, J. (2006). Gender, representative bureaucracy, and law enforcement: The case of sexual assault. *Public Administration Review*, 66(6), 850-860. Retrieved from JSTOR database.
- Meier, K. J., & Smith, K. B. (1994). Representative democracy and representative bureaucracy: Examining the top-down and bottom-up linkages. *Social Science Quarterly*, 75(4), 790-803. Retrieved from JSTOR database.
- Meier, K. J., Wrinkle, R. D., & Polinard, J. L. (1999). Representative bureaucracy and distributional equity: Addressing the hard question. *Journal of Politics*, 61(4), 1025-1039.
- Mladenka, K. R. (1991). Public employee unions, reformism, and Black employment in 1,200

 American cities. *Urban Affairs Review*, 26(4), 532-548.

 http://doi.org/10.1177/004208169102600405
- Morabito, M. M., & Shelley, T. O. (2015). Representative bureaucracy: Understanding the correlates of the lagging progress of diversity in policing. *Race and Justice*, *5*(4), 330-355. http://doi.org/10.1177/2153368715575376
- Mosher, F. C. (1968). *Democracy and the public service*. New York, NY: Oxford University Press.



- National Center for Women and Policing, Feminist Majority Foundation. (2001, June).

 *Recruiting and retaining women: A self-assessment guide for law enforcement (NCJRS Publication No. 188157). Bureau of Justice Assistance, Office of Justice Programs, U.S. Department of Justice. Retrieved from https://www.ncjrs.gov/pdffiles/bja/185235.pdf
- National Center for Women and Policing, Feminist Majority Foundation. (2002, April). *Equality denied. The status of women in policing: 2001*. Los Angeles, CA: National Center for Women and Policing. Retrieved from http://womenandpolicing.org/PDF/2002_Status_Report.pdf
- National Center for Women and Policing, Feminist Majority Foundation. (2003a, Spring). *Hiring*and retaining more women: The advantages to law enforcement agencies. Los Angeles,

 CA: National Center of Women and Policing. Retrieved from

 http://womenandpolicing.com/pdf/NewAdvantagesReport.pdf
- National Center for Women and Policing, Feminist Majority Foundation. (2003b, Spring). *Under scrutiny: The effect of consent decrees on the representation of women in sworn law enforcement*. Los Angeles, CA: National Center for Women and Policing. Retrieved from http://womenandpolicing.com/pdf/Fullconsentdecreestudy.pdf
- Niederhoffer, A. (1969). *Behind the shield: The police in urban society*. Garden City, NY: Anchor Books.
- President's Commission on Law Enforcement and Administration of Justice. (1967, February).

 The challenge of crime in a free society. Washington, DC: U.S. Government Printing

 Office.



- President's Task Force on 21st Century Policing. (2015, May). *Final report of the President's task force on 21st century policing*. Washington, DC: Office of Community Oriented Policing Services.
- Raganella, A. J., & White, M. D. (2004). Race, gender, and motivation for becoming a police officer: Implications for building a representative police department. *Journal of Criminal Justice*, 32(6), 501-513. http://doi.org/10.1016/j.jcrimjus.2004.08.009
- Reaves, B. A. (2010). *Local police departments, 2007.* U.S. Department of Justice, Office of Justice Programs, Bureau of Justice Statistics. Retrieved from http://www.bjs.gov/content/pub/pdf/lpd07.pdf
- Reaves, B. A. (2012, October). *Hiring and retention of state and local law enforcement officers,* $2008 statistical \ tables \ (NCJRS \ Publication \ No. \ 238251). \ U.S. \ Department of Justice,$ Office of Justice Programs, Bureau of Justice Statistics. Retrieved from http://www.bjs.gov/content/pub/pdf/hrslleo08st.pdf
- Reaves, B. A. (2015, May). Local police departments, 2013: Personnel policies, and practices (NCJRS Publication No. 248677). U.S. Department of Justice, Office of Justice Programs, Bureau of Justice Statistics. Retrieved from http://www.bjs.gov/content/pub/pdf/lpd13ppp.pdf
- Reaves, B. A., & Goldberg, A. L. (1999, April). Law Enforcement Management and

 Administrative Statistics, 1997: Data for individual state and local agencies with 100 or

 more officers (NCJRS Publication No. 171681). U.S. Department of Justice, Office of

 Justice Programs, Bureau of Justice Statistics. Retrieved from

 https://www.bjs.gov/content/pub/pdf/lemintro.pdf



- Reaves, B. A., & Hickman, M. J. (2002, May). *Police departments in large cities, 1990-2000*(NCJRS Publication No. 175703). U.S. Department of Justice, Office of Justice Programs, Bureau of Justice Statistics. Retrieved from https://www.bjs.gov/content/pub/pdf/pdlc00.pdf
- Ruccucci, N. (1986). Female and minority employment in city government: The role of unions.

 *Policy Studies Journal, 15(1), 3-15. Retrieved from http://minitex.umn.edu
- Salzstein, G. H. (1979). Representative bureaucracy and bureaucratic responsibility: Problems and prospects. *Administration & Society*, *10*(4), 465-475. http://doi.org/10.1177/009539977901000404
- Sass, T. R., & Troyer, J. L. (1999). Affirmative Action, political representation, unions, and female police employment. *Journal of Labor Research*, *XX*(4), 571-587. Retrieved from ABI/INFORM Global database.
- Schuck, A. M. (2014). Female representation in law enforcement: The influence of screening, unions, incentives, community policing, CALEA, and size. *Police Quarterly*, *17*(1), 54-78. http://doi.org/10.1177/1098611114522467
- Scrivner, E. (2006). *Innovations in police recruitment and hiring: Hiring in the Spirit of*Service. U.S. Department of Justice, Office of Community Oriented Policing Service.

 Retrieved from
 - http://www.cops.usdoj.gov/files/ric/Publications/innovationpolicerecruitmenthiring.pdf
- Selden, S. C. (1997). The promise of representative bureaucracy: Diversity and responsiveness in a government agency. New York, NY: M. E. Sharpe.



- Selden, S. C., Brudney, J. L., & Kellough, J. E. (1998). Bureaucracy as a representative institution: Toward a reconciliation of bureaucratic government and democratic theory.

 **American Journal of Political Science*, 42(3), 717-744. Retrieved from http://mtrproxy.mnpals.net/login?url=http://search.proquest.com/docview/196852042?ac countid=12415
- Shaffir, W. B., & Stebbins, R. A. (2003). Introduction to fieldwork. In M. R. Pogrebin (Ed.), *Qualitative approaches to criminal justice: Perspectives from the field* (pp. 2-16).

 Thousand Oaks, CA: Sage Publications.
- Shusta, R. M., Levine, D. R., Wong, H. Z., Olson, A. T., & Harris, P. R. (2011). *Multicultural law enforcement: Strategies for peacekeeping in a diverse society* (5th ed.). Boston, MA: Prentice Hall.
- Sigelman, L., Welch, S., Bledsoe, T., & Combs, M. (1997). Police brutality and public perceptions of racial discrimination: A tale of two beatings. *Political Research Quarterly*, 50(4), 777-791. http://doi.org/10.1177/106591299705000403
- Sklansky, D. A. (2006). Not your father's police department: Making sense of the new demographics of law enforcement. *The Journal of Criminal Law and Criminology*, 96(3), 1209-1243. Retrieved from JSTOR database.
- Skolnick, J. H. (1994). *Justice without trial: Law enforcement in democratic society* (3rd ed.). New York, NY: Macmillan College.
- Skolnick, J. H., & Bayley, D. H. (1988). Theme and variation in community policing. In M. Tonry, & N. Morris, *Crime and justice: A review of research* (pp. 1-37). Chicago, IL: The University of Chicago Press.



- Slack, J. D. (1987). City managers, police chiefs, and fire chiefs in the South: Testing for determinants and impact of attitudes toward Affirmative Action. *Review of Public Personnel Administration*, 8(1), 11-32. Retrieved from ABI/INFORM Global database.
- Sowa, J. E., & Selden, S. C. (2003). Administrative discretion and active representation: An expansion of the theory of representative bureaucracy. *Public Administration Review*, 63(6), 700-710. Retrieved from JSTOR database.
- Steel, B. S., & Lovrich Jr., N. P. (1987). Equality and efficiency tradeoffs in Affirmative

 Action real or imagined? The case of women in policing. *The Social Science Journal*,

 24(1), 53-70. https://doi.org/10.1016/0362-3319(87)90006-1
- Stein, L. (1986). Representative local government: Minorities in the municipal work force. *The Journal of Politics*, 48(3), 694-713. http://doi.org/10.2307/2131173
- Stevens, D. J. (2018). *An introduction to American policing* (2nd ed.). Burlington, MA: Jones & Bartlett Learning.
- Sullivan, E., & Gillum, J. (2014, September 8). Diversity in the police ranks. *The Associated Press*. Retrieved from https://www.abqjournal.com/458611/report-hispanics-on-forces-lacking.html
- Sun, I. Y., & Payne, B. K. (2004). Racial differences in resolving conflicts: A comparison Between Black and White police officers. *Crime & Delinquency*, *50*(4), 516-541. http://doi.org/10.1177/0011128703259298
- Theobald, N. A., & Haider-Markel, D. P. (2008). Race, bureaucracy, and symbolic representation: Interactions between citizens and police. *Journal of Public Administration**Research & Theory, 19(2), 409-426. http://doi.org/10.1093/jopart/mun006



- U.S. Census Bureau. (2013, April). *U.S. Census Data*. Author. Retrieved from https://www.census.gov/2010census/popmap/ipmtext.php?fl=27
- U.S. Commission on Civil Rights. (2000, November). Recruitment, selection, and training for police work. In *Revisiting "Who is guarding the guardians": A report on police practices and civil rights in America* (Ch. 2). Author. Retrieved from http://www.usccr.gov/pubs/guard/ch2.htm
- U.S. Department of Justice. (2001, January). *Principles for promoting police integrity: Examples of promising police practices and policies*. Author. Retrieved from https://www.ncjrs.gov/pdffiles1/ojp/186189.pdf
- U.S. Department of Labor. (2018, May). Labor force statistics from the current population survey. Bureau of Labor Statistics. Retrieved on May 24, 2018 from https://data.bls.gov/timeseries/LNS14000000
- U.S. National Advisory Commission on Civil Disorders. (1968). *Report of the National Advisory Commission on civil disorders*. Washington, DC: U.S. Government Printing Office.
- U.S. National Advisory Commission on Criminal Justice Standards and Goals. (1973). *Police*.Washington, DC: U.S. Government Printing Office.
- U.S. President's Commission on Law Enforcement and Administration of Justice. (1967). *Task force report: The police*. Washington, DC: U.S. Government Printing Office.
- Van der Lippe, T., Graumans, A., & Sevenhuijsen, S. (2004). Gender policies and the position of women in the police force in European countries. *Journal of European Social Policy*, 14(4), 391-405. http://doi.org/10.1177/0958928704046880



- Walker, S. (1985). Racial minority and female employment in policing: The implications of "glacial" change. *Crime & Delinquency*, *31*(4), 555-572. http://doi.org/10.1177/0011128785031004006
- Walker, S. (2008). The neglect of police unions: Exploring one of the most important areas of American policing. *Police Practice & Research*, 9(2), 95-112. http://doi.org/10.1080/15614260802081253
- Walker, S. (2015). The challenge of policing in a democratic society: A personal journey toward understanding. In R. G. Dunham & G. P. Alpert (Eds.), *Critical issues in policing* (7th ed.): *Contemporary readings* (pp. 596-609). Long Grove, IL: Waveland Press.
- Walker, S. W., & Katz, C. M. (2002). *Police in America: An introduction* (4th ed.). New York, NY: McGraw-Hill Higher Education.
- Wareham, J., Smith, B. W., & Lambert, E. G. (2015). Rates and patterns of law enforcement turnover: A research note. *Criminal Justice Policy Review*, 26(4), 345-370. http://doi.org/10.1177/0887403413514439
- Warner, R. L., Steel, B. S., & Lovrich, N. P. (1989). Conditions associated with the advent of representative bureaucracy: The case of women in policing. *Social Science Quarterly*, 70(3), 562-578. Retrieved from JSTOR database.
- Weitzer, R., & Tuch, S. (2006). *Race and policing in America: Conflict and reform.* New York, NY: Cambridge University Press.
- Weitzer, R., Tuch, S. A., & Skogan, W. G. (2008). Police-community relations in a majority-black city. *Journal of Research in Crime and Delinquency*, 45(4), 398-428. http://doi.org/10.1177/0022427808322617



- Westley, W. A. (1970). Violence and the police: A sociological study of law, custom, and morality. Cambridge, MA: Massachusetts Institute of Technology.
- White, M. D., & Escobar, G. (2008). Making good cops in the twenty-first century: Emerging issues for the effective recruitment, selection and training of police in the United States and abroad. *International Review of Law Computers and Technology*, 22(1-2), 119-134. http://doi.org/10.1080/13600860801925045
- Wilkins, V. M., & Williams, B. N. (2008). Black or blue: Racial profiling and representative bureaucracy. *Public Administration Review*, 68(4), 654-664. http://doi.org/10.1111/j.1540-6210.2008.00905.x
- Wilkins, V. M., & Williams, B. N. (2009). Representing blue: Representative bureaucracy and racial profiling in the Latino community. *Administration & Society*, 40(8), 775-798. http://doi.org/10.1177/0095399708326332
- Wilson, J. Q., & Kelling, G. L. (1982). Broken windows: The police and neighborhood safety. InJ. E. Jacoby, T. A. Severance, & A. S. Bruce (Eds.), *Classics of Criminology* (4th ed.)(pp. 544-551). Long Grove, IL: Waveland Press.
- Yin, R. K. (2009). *Case study research: Design and methods* (4th ed.). Applied Social Research Methods Series (Vol. 5) (Series). Thousand Oaks, CA: Sage Publications.
- Zhao, J., He, N., & Lovrich, N. (2005). Predicting the employment of minority officers in U.S. cities: OLS fixed-effect panel model results for African American and Latino officers for 1993, 1996, and 2000. *Journal of Criminal Justice*, 33, 377-386. Retrieved from Science Direct database.



- Zhao, J., Herbst, L., & Lovrich, N. (2001). Race, ethnicity and the female cop: Differential patterns of representation. *Journal of Urban Affairs*, 23(3/4), 243-257. Retrieved from Academic Search Premier database.
- Zhao, J., & Lovrich, N. (1998). Determinants of minority employment in American municipal police agencies: The representation of African American officers. *Journal of Criminal Justice*, 26(4), 267-277. Retrieved from Science Direct database.

